BUILDING BALANCE TO BUILD RESILIENCE

TITT

AN EMPIRICAL STUDY ON THE NEIGHBOURHOOD BALANCE POLICY OF ROTTERDAM'S RESILIENCE STRATEGY

Guillermo Prieto Viertel

BUILDING BALANCE TO BUILD RESILIENCE

AN EMPIRICAL STUDY ON THE NEIGHBOURHOOD BALANCE POLICY OF ROTTERDAM'S RESILIENCE STRATEGY

Master thesis submitted to Delft University of Technology in partial fulfilment of the requirements for the degree of **MASTER OF SCIENCE** in **Engineering and Policy Analysis** Faculty of Technology, Policy and Management

by

Guillermo Prieto Viertel

To be defended in public on August 25th 2022

Graduation committee:

Chairperson and 2nd Supervisor:

Dr. ir. C. (Els) van Daalen. Faculty of TPM. Section Policy Analysis.

1st Supervisor:

Dr. Samantha Copeland. Faculty of TPM. Section Ethics and Philosophy of Technology. Advisor:

Dr. Camilo Benitez Avila. Faculty of TPM. Section Ethics and Philosophy of Technology. Advisor:

Dr. Claudiu Forgaci. Faculty of Architecture. Section Urban Design.

Advisor:

Mikhail Sirenko. Faculty of TPM. Section Transport & Logistics.

Keywords: Urban resilience, Social resilience, Social cohesion, Balanced neighbourhoods, Social mixing, Rotterdam, PLS-SEM, Spatial econometrics

Cover design: "Todos a una" August 10th 2022, Javier Prieto Fanjul

Copyright ©2022 by G. Prieto Viertel

An electronic version of this master thesis is available at https://repository.tudelft.nl/



EXECUTIVE SUMMARY

The rise in urban stresses has prompted the need for preparedness of urban areas to face precarious circumstances. Consequently, the concept of urban resilience has grown in popularity not only to tackle sudden shocks but also to face long-lasting socio-economic tensions. To implement the arrangements for resilience policy, literature suggests that social factors govern the resilience of urban areas. Communities in which the residents work together and have common goals have a stronger willingness to cooperate. To this end, social cohesion has been proven to be significant for subsistence in the event of a catastrophe. Cohesive communities protect residents against threats, care of others during hardships, and ultimately promote community resilience.

Balanced neighbourhood policies aim to strengthen the cohesion between citizens, communities, and social institutions departing from the assumption that social mix fosters social cohesion. Their goal is to increase the social mix of specific areas to avoid the clustering and segregation of disadvantaged households to, as a result, promote resilient actions. There is, however, literature that suggests that the anticipated effects are rather inconclusive and usually not achieved. Instead, balanced neighbourhood policies would promote segregation by forcing the displacement of groups of residents.

The issue arises whether balanced neighbourhoods trigger resilient actions that are pivotal in resilient communities. In other words, **does neighbourhood balance increase resilient action of neighbourhood residents?** We took a cross-sectional confirmatory approach to understand the social mechanism that triggers resilient action in balanced neighbourhoods based on Partial Least Squares—Structural Equation Modelling (PLS-SEM) and spatial econometrics. To this end, the research tests the assumptions on which the municipality of Rotterdam grounds their Resilience Strategy's balanced neighbourhood policy, the Woonvisie, using a 2019 public survey on social development. Rotterdam's definition of a balanced neighbourhood is defined from a set of conditions for the amount of houses in different house price segments. Therefore, the tested model is based on the grounds that geographically connected people become affected by their neighbourhood's balance to promote resilient action. As such, we use the willingness to help friends and neighbours to characterise informal support as resilient action. Here we show that balanced neighbourhoods are associated with less informal support: the higher the balance, the fewer residents are willing to help their friends and neighbours.

The results indicate that social cohesion is the social mechanism that triggers help between friends and relatives and fully acts as the mechanism for resilient actions triggered by the balance in a neighbourhood. From the multiple combinations of houses in different house price segments that the definition of balanced neighbourhoods allows, we distinguish two associations. On one hand, house price distributions which foment a reduction in polarisation (more middle-priced houses) are negatively associated with social cohesion. On the other hand, balanced neighbourhoods which foment polarisation (more low- and high-priced houses) are positively associated with social cohesion. This indicates that our results are in line with Putnam's homophily principle, i.e. 'birds of a feather flock together'. This outcome is opposite to the policy discourse of governments in favour of balanced neighbourhoods, including the municipality of Rotterdam, that mixed neighbourhoods foster social cohesion and therefore resilient action.

The testing of the theory is complemented in two ways. First, we show that social cohesion and informal support are not constrained by administrative boundaries, so the social perceptions and actions in nearby neighbourhoods affects the level of the other neighbourhoods. Second, we found no moderating effect of factors related to the demographics and the built environment that can promote or deter social interactions, and thus are aspects of what can be considered a resilient neighbourhood.

The analysis also shows that Rotterdam's definition of balance allows multiple and dispersed combinations of the amount of houses in each price segment, which can result in counterintuitive conceptions of balance. In addition, the results show apparently contradicting results of the relationship between balance and social cohesion depending on whether the distribution foments house price polarisation. As a result, we argue that the definition is under-specified and can be misleading.

Finally, only 2.1% of the possible balance distributions yielded an acceptable goodnessof-fit of our model. This could be indicative that the model needs to be reevaluated. We found that neighbourhood ethnic heterogeneity and house type heterogeneity are directly associated to social cohesion and informal support, respectively. Future research should elaborate on the theory on which the model is grounded and create coherence to the empirical relationships identified. In contrast, the few fitting distributions could otherwise indicate that that social cohesion and informal support cannot be explained by the balance in a neighbourhood and that the policy should be reevaluated. Under this second interpretation, the study has uncovered which are the balance distributions for the city that can actually show the alleged effects of balance.

Based on these findings, a policy advice is formulated. If the objective is to increase social cohesion and resilient actions, we discourage the municipality of Rotterdam to approach this by building balanced neighbourhoods. Nevertheless, we have seen that building social cohesion is a way to build social resilience, so recently developed city programs focused on the development of neighbourhood organisations, which not only provide a space for social cohesion but also to collect and share resources directly, are a step forward from the Woonvisie.

PREFACE

Love your neighbour!

Not too long ago, our neighbour alerted my family of a possible robbery. Friday night, they detected suspicious behaviour from a security camera and motion sensor they had set up after a previous robbery in their flat. The camera recorded a person placing a paper on their front door hinge, and then walking towards our house, to do the same. In summer, many families spend the weekends in their second residences, so if the paper is still there on Saturday, the family is probably away for the whole weekend. Coincidentally, our neighbour was away, but we were in town, so we coordinated to face the situation. Saturday night at 3am, the motion sensor activated, and our neighbour notified us. My family called the police and let them in the building for inspection. Nobody was to be found. Still, the police decided to stay in the block, and later that night, they arrested people coming out of the building's main entrance. They cross-checked the arrested with the security camera recordings and found a match. That night, none of the building houses were broken into.

The trust between neighbours galvanised them into resilient action. They were willing to help each other and to protect their neighbourhood. To understand what generated that willingness is essential. This motivated me to contribute to the field of resilience and write this study. Hopefully, this will cease to be an ideal behaviour and will become the norm in neighbourhoods.

> *Guillermo Prieto Viertel Barcelona, August 2022*

CONTENTS

Ex		ive Summary	
Pr	eface		
Li	st of]	Tables	
Li	st of l	Figures	
Ac	knov	vledgements	
1	Inte	aduction	1
1	11	Background	1
	1.1	111 Global challenge	1
		112 Call for local action	2
	1.2	Case study: Rotterdam	3
	1.3	Knowledge gap	4
	1.4	Research questions	5
	1.5	Research approach	5
	1.6	Research structure	6
2	Lite	rature review	9
	2.1	Urban resilience	9
		2.1.1 The resilient city	9
		2.1.2 Informal support as resilient action	0
	2.2	How can social cohesion improve resilience?	1
		2.2.1 From social cohesion to resilient actions.	2
		2.2.2 Origin and definitions of social cohesion.	2
		2.2.3 Social cohesion conceptual frameworks	13
		2.2.4 Urban elements that shape social cohesion and informal support	17
	2.3	Balanced neighbourhoods	8
		2.3.1 Social mixing	9
		2.3.2 From social mix to social cohesion	20
		2.3.3 Balanced housing	22
	2.4	Resilient Rotterdam	23
		2.4.1 Resilience Strategy	24
		2.4.2 A balanced society	26
		2.4.3 Controversy of balance	27
	2.5	Hypothesis development	29
	2.6	Conclusion	31

3	Met	thodology	33
	3.1	Data	33
		3.1.1 Social cohesion and informal support	34
		3.1.2 Balanced neighbourhoods	35
		3.1.3 Moderating factors	36
		3.1.4 Neighbourhood shape	36
	3.2	Measurement model	38
	3.3	Research methods	38
		3.3.1 Kullback-Leibler divergence	40
		3.3.2 K-means clustering	43
		3.3.3 Partial Least Squares Structural Equation Modelling	43
		3.3.4 Spatial econometrics.	46
	3.4	Computational framework and software implementation.	48
	3.5	Conclusion	49
4	Res	ults	51
	4.1	Balance of Rotterdam	51
	4.2	PLS-SEM model results	56
		4.2.1 Model goodness-of-fit	56
		4.2.2 Reflective measurement model	58
		4.2.3 Formative measurement model	58
		4.2.4 Structural model	59
	4.3	Spatial econometrics results	61
		4.3.1 Spatial distribution of Neighbourhood balance	61
		4.3.2 Spatial effects in Social cohesion.	62
		4.3.3 Spatial effects in Informal support	64
	4.4	Results of moderation analysis	66
	4.5	Conclusion	67
5	Dise	cussion	71
	5.1	The dichotomy of balance	71
	5.2	Social indicators for resilience	76
	5.3	Balanced neighbourhoods as an instrument to achieve social resilience by	
		means of social cohesion	79
6	Con	nclusion	87
-	6.1	Research questions	87
	6.2	Policy advice	90
	6.3	Limitations and future research directions	91
Α	App	Dendix A	11
	A.1		111
В	Арр	endix B J	15
	B.1	Wijkprofiel data	115
	B.2	Exploratory factor analysis	116

С	C Appendix C	121
	C.1 Informal support as the common factor between willingness to help the order of the balance	riends
		121
	C.2 Informal support as willingness to help others	122
	C.3 Informal support as the common factor between willingness to help	rela-
	tives, friends and neighbours, and others	122
	C.3.1 PLS-SEM model results	122
D	D Appendix D	129
	D.1 PLS-SEM structural model	129
	D.2 Outliers	131
	D.3 LISA results	131
	D.3.1 LISA of Social cohesion.	131
	D.3.2 LISA of Informal support	131
Е	E Appendix E	135
	E.1 Spatial effects in Social cohesion taking into account river effects.	135
	E.2 Spatial effects in Informal support taking into account river effects.	135
F	Appendix F	139
	F.1 Image deconstruction of urban design elements	139

LIST OF TABLES

2.1	WOZ-value housing segments in Rotterdam in 2018	27
3.1	Characteristics of the indicators for <i>Social cohesion</i> and <i>Informal support</i> in the Wijkprofiel in 2019.	34
3.2	WOZ-value segments from 2018 and 2019	35
3.3	Characteristics of the WOZ-value segments used to measure the level of balance for the year 2019.	35
3.4	Characteristics of the exogenous factors.	37
3.5	Differences in the Wijkprofiel survey structure compared to the official CBS neighbourhood division	37
3.6	Software implementation of the research methods.	50
4.1	Goodness-of-fit of the estimated models.	58
4.2	Formative construct assessment.	59
4.3	Structural model total, direct, and indirect path coefficients.	61
4.4	Results of the Durbin spatial model with <i>Social cohesion</i> as dependent vari- able	64
4.5	Results of the Durbin spatial model with <i>Informal support</i> as dependent	
	variable	66
4.6	Results of moderation analysis.	67
4.7	Status of the formulated hypotheses for every Model	69
A.1	Definitions of social cohesion in the academic literature.	111
A.2	Definitions of social cohesion in the policy discourse.	112
B.1	Questions in the Wijkprofiel survey associated to the indicators used in the	110
DO	study for each construct.	116
B.2	Indicators and their corresponding dimensions for EFA	117
D.3	EFA results using varimax rotation	119
D.4		119
C .1	Characteristics of the indicators for care giving in the Wijkprofiel in 2019.	121
C.2	Goodness-of-fit of the estimated models	123
C.3	Correlation matrix of Informal support indicators.	125
C.4	Reflective construct reliability assessment.	125
C.5	Reflective construct item loading assessment.	125
C.6	Formative construct assessment.	126
C.7	Assessment of the structural model reliability.	126
C.8	Assessment of the predictive power of the structural model	127

C.9	Structural model total, direct, and indirect path coefficients.	127
D.1 D.2 D 3	Assessment of the structural model reliability	130 131
D.0	ure 4.8 for visibility purposes.	131
E.1	Results of the Durbin spatial model with <i>Social cohesion</i> as dependent variable taking into account river effects.	137
E.2	Results of the Durbin spatial model with <i>Informal support</i> as dependent variable taking into account river effects.	137

LIST OF FIGURES

1.1	Balance as a remedy: Rotterdam strategy to increase the municipal re- silience.	4
1.2	Research flow diagram.	7
2.1	Phases and characteristics of urban resilience	10
2.2	Diagrams of the two theories for the development of constructs discussed in this study	15
2.3	Social cohesion indicators in Schnabel et al. (2008) for the Dutch Office of	15
	Social and Cultural Planning.	15
2.4 2.5	Dimensions and indicators used in CBS (2015) to determine social cohesion Conceptual model of the association between urban elements and social	16
	cohesion	18
2.6	Alleged benefits of social mix	20
2.7	Conceptual model of balanced neighbourhood policies as the alleged means	22
		23
2.8	Rotterdams Resilience Strategy goals	25
2.9	Balanced neighbourhood objective of the municipality of Rotterdam	27
2.10	Boxplot of the number of houses that belong to the Social, Middle, Higher,	
	and Top segment in each neighbourhood in Rotterdam in 2019.	28
2.11	Illustration of the concepts and the relationships that are the analysis ob-	
	jective of this research	29
3.1	Structural and measurement model	39
3.2	Example of the Kullback-Leibler divergence	42
3.3	Exogenous factors as moderators of the effect of Neighbourhood balance	
	on Social cohesion and on Informal support.	45
3.4	Example of polygon neighbourhoods	47
3.5	Computational framework.	49
4.1	Solution space for the definition of balance according to the municipality	
	of Rotterdam	52
4.2	Density estimation plot of the distribution of <i>Neighbourhood balance</i> for	
	each neighbourhood	53
4.3	House stock distributions within the balance definition of Rotterdam that	
	have an acceptable goodness-of-fit.	54
4.4	Elbow method for the optimal number of clusters.	55
4.5	Clusters of the house stock distributions that meet the definition of balance	
	from the municipality and that have an acceptable goodness-of-fit.	55

4.6	Boxplot of the path coefficients for each cluster.	56
4.7	Housing stock distributions of the models with best goodness-of-ht for each identified cluster.	57
4.8	Choropleth of the standardised measure of <i>Neighbourhood balance</i> in 2019 for every model	62
4.9	Choropleth of the standardised measure of <i>Social cohesion</i> in 2019 for ev-	02
4 10	ery model	63
4.10	ance in 2019	63
4.11	Choropleth of the standardised measure of <i>Informal support</i> in 2019 for all Models	65
4.12	Choropleth of the distribution of <i>Informal support</i> and <i>Social cohesion</i> in	05
	2019	65
5.1	Example of the implication of the definition of balanced neighbourhood of	
5.2	Analogy of model fit.	73 75
5.3	Concave and convex distributions of balance follow the homophily princi-	01
	pie	81
C.1	House stock distributions within the balance definition of Rotterdam that have an accentable goodness-of-fit	123
C.2	Clusters of the house stock distributions that meet the definition of balance	125
C 3	from the municipality and that have an acceptable goodness-of-fit.	123
0.0	each identified cluster	124
D.1	LISA choropleth of <i>Social cohesion</i>	132
D.2	LISA choropleth of <i>Informal support</i>	133
D.3	LISA choropieth of <i>informal support</i> and <i>Social conesion</i>	133
E.1 F 2	LISA choropleth of <i>Social cohesion</i> taking into account river effects.	136 138
E.3	LISA choropleth of <i>Informal support</i> and <i>Social cohesion</i> taking into ac-	150
	count river effects.	138
F.1	Special features of public-private interplay in selected neighbourhoods.	141
F.2 F.3	Special features of public spaces in selected neighbourhoods	141
	hoods.	141
F.4	Approximate location of the images in figures A.1a, A.1b, A.2a, A.2b, A.3a, and A.3b in Rotterdam.	142

ABBREVIATIONS

Abbreviation	Concept
100RC	100 Resilient Cities
AIC	Akaike information criterion
AVE	Average variance extracted
BAG	Addresses and Buildings Key Registry
BFPI	Best fitting proper indices
BRP	Personal Records Database
CBS	Statistics Netherlands
CB-SEM	Covariance based Structural equation modelling
CI	Confidence interval
ESDA	Exploratory Spatial Data Analysis
f^2	Cohen's effect size
GDP	Gross Domestic Product
Н	Hypothesis
HH	High-High
HHI	Herfindahl-Hirschman Index
HL	High-Low
IPCC	Intergovernmental Panel on Climate Change
IS	Informal support
KL	Kullback-Leibler
КМО	Kaiser-Meyer-Olkin
LISA	Local Indicators of Spatial Association
LH	Low-High
LL	Low-Low
MAE	Mean absolute error
NA	Not Available
NB	Neighbourhood balance
NHG	National Mortgage Guarantee
NPRZ	National Programme Rotterdam South
ns	non-significant
OBI	Research and Business Intelligence
OSM	OpenStreetMap
PLS-SEM	Partial least squares structural equation modelling
Q^2	Predictive relevance
R ²	Coefficient of determination
r^2	Pearson's correlation coefficient
RMSE	Root mean squared error
s.d.	standard deviation
SDG	Sustainable Development Goal
SC	Social cohesion
SQ	Sub-question

Abbreviation	Concept
SRMR	Standardized root mean squared residual
SSE	Sum-squared error
VIF	Variance inflation factor
WHO	World Health Organization
WOZ	Valuation of Immovable Property Act

ACKNOWLEDGEMENTS

This thesis represents the end of my two-year period as a master's degree student in Engineering and Policy Analysis at the TU Delft. Having experienced the lack of vitamin D due to the long hours in the library and to the Dutch weather, I want to take a moment to thank all the people that have been next to my side.

First, thank you, yaya, for being like a second mom to Sandra and me. We all know you would have loved to come to the Netherlands and keep travelling like you always did. I am glad I was able to show you around in your last moments, and that you could see the place that has been my home for the last two years. We all take you in our hearts like the charming person you always were.

Thank you mama and papa for always investing in my education without hesitation. Thank you for bringing me up with ambition to learn and explore, but always being appreciative and not forgetting where we come from. Especial thanks to my dad for designing the amazing front page of the report.

Thank you, Sandra, for being an annoying sister, but being there when necessary. The last two years were tough for the family, but you took care of everyone better than anyone could ask for. I will always insist to you that you are more than capable to pursue your master's at the TU Delft.

Thank you, Carla, for coping with the long distance for such a long time. It was not easy, but we succeeded and grew stronger from it. Your unconditional love was crucial for my survival on a daily basis and I cannot wait to be next to you again.

Second, I want to thank my graduation committee who made this thesis possible. Thank you, Els, for your personal support and your reality checks. Samantha, for accepting to be involved in this project and your insights on the philosophical side. Claudiu, for reminding me that there is more to life than numbers. Camilo and Mikhail for your weekly guidance and support, rescuing me when I felt like drowning. It was a pleasure working hand in hand and I hope our paths meet again in the future.

Finally, I want to thank all the friends that were around during these two years and made the whole experience unforgettable. Thanks to my EPA crew for the long evenings at the beach that made me forget about the COVID-19 limitations. Thanks to Sundayyy for existing and creating the most cherished moments. Thanks to all my housemates who ended up becoming part of my family. Finally, thanks to the TU library crew for holding each other up during the thesis period, without you this report would have probably never seen the light of day.

1

INTRODUCTION

1.1. BACKGROUND

1.1.1. GLOBAL CHALLENGE

At present, more than half of the world's population lives in urban areas, and it is expected that this amount will increase to 68% by 2050 (United Nations et al., 2018). The high concentration of people living in urban spaces gives room for innovation and learning but also results in an increase in vulnerability to threats like climate change, resource scarcity, and social exclusion (Meerow et al., 2016; Wolf et al., 2015). This conspicuously complicates the objective of attaining the Sustainable Development Goal (SDG) 11 'sustainable cities and communities' and has cross-cutting impact on other SDGs, especially SDG 9 'industry, innovation and infrastructure', and SDG 13 'climate action'. As a result, policy-makers have prioritised the interest of urban areas in their agenda (UN-Habitat, 2016).

Cities are of special relevance due to their large number of vulnerable people. An accelerating trend in adverse effects of living in major urban areas has been increasingly observed around the globe. Income inequality has exacerbated in cities across the United States (Heinrich Mora et al., 2021), female citizens have seen an increase in inactivity and unemployment across East and South-East Asian urban regions (ILO, 2020), and a greater shortage of housing threatens major European municipalities (OECD, 2021). What is more, the COVID-19 pandemic has clearly shown how globally connected cities are more exposed to global emergencies and require appropriate urban policy planning (UN-Habitat, 2020).

Even though urbanisation comes with several challenges, sustainable and inclusive growth can also be achieved if properly managed (European Commission, 2010). The potential of urban areas is clear, given that they contribute to 80% of the global GDP (UN-Habitat, 2016). Consequently, the concept of urban resilience has grown in popularity to tackle these challenges that abrupt urbanisation and a lack of proper planning have put in place (Saja et al., 2019).

1.1.2. CALL FOR LOCAL ACTION

While the definition of urban resilience subsumes multiple dimensions and is still debated among scholars and policy-makers, it is widely shared that social elements shape the level of resilience of urban areas. This can be seen, for instance, in Barcelona's neighbourhood L'Eixample, that ranked first place in 2020 TimeOut's Coolest neighbourhoods in the world index. With COVID-19 wreaking lock-downs over the world, the community spirit and solidarity kept the neighbourhood as a hub of mutual aid during the pandemic (Walker-Arnott, 2020).

Social cohesion has been proven to be significant for subsistence in the event of a catastrophe (Elliott et al., 2010; Larimian et al., 2020). Cohesive communities are more willing to take resilient actions like protecting residents against threats, caring of others during hardships, and ultimately promote community resilience (Center for American Progress, n.d.; Forrest & Kearns, 2001; Larimian et al., 2020). Therefore, governments are looking to strengthen the relationships between citizens, communities, and social institutions to boost social cohesion.

The fundamental reason for the growing interest in neighbourhoods in today's policy debate is their context effects. Negative neighbourhood effects have often been associated with the segregation of poor communities or ethnic minorities. Large concentrations of these groups in specific areas are believed to reinforce and perpetuate poverty and exclusion, and consequently reduce social cohesion (Bolt et al., 2010; Colomb, 2011). Major world cities have observed a gradual decrease in social cohesion since the late 20th century (Schiefer & van der Noll, 2017), so this relation was embraced by policy-makers, as it implies that eliminating such concentrations would address negative neighbourhood effects (Méreiné-Berki et al., 2021). As a consequence, neighbourhoods have been targeted by policy-makers and urban planners to design solutions to create more resilient cities.

A policy that aims to halt the deterioration of social cohesion is the establishment of balanced neighbourhoods. Balanced neighbourhood policies aim to increase the social mix of specific areas to avoid the clustering and segregation of disadvantaged house-holds (Hananel et al., 2022). They aim to increase the social cohesion of cities as they allegedly counter antisocial behaviours as well as criminal activity to, as a result, increase the resilience to distress and tackle long-lasting systemic issues (Jordan, 2018; Ministry for Housing and the Civil Service of the Netherlands, 2017). Creating neighbourhoods with a balanced social composition is a commonly accepted technique by policy-makers and scholars for addressing perceived negative neighbourhood effects (Bolt et al., 2010; Chaskin & Joseph, 2011; Kearns & Mason, 2007; Manley et al., 2011; van Ham & Manley, 2012). There is, however, literature that challenges this believed abstraction and suggests that the anticipated effects are rather inconclusive and usually not achieved (Custers, 2021; van Ham & Manley, 2012). In addition, balanced neighbourhoods are also widely criticised for promoting exclusion and having counterproductive effects on social cohesion and resilience (van Eijk, 2010).

Given the dearth of consensus on the impacts of balanced neighbourhoods, it is remarkable that governments have embraced these policies with such zeal. To elucidate the extent to which governments should implement balanced neighbourhoods, this study aims to contribute to the field of urban resilience studies by analysing the effect of balanced neighbourhoods as a policy instrument to improve the resilience of cities by increasing social cohesion. The research will use the city of Rotterdam, Netherlands, as a case study.

1.2. CASE STUDY: ROTTERDAM

As the second-largest city in the Netherlands, Rotterdam is critical in the development of the country. Today, Rotterdam is considered an attractive place to live, as can be seen, reflected in the inflation of housing prices brought by its increasing popularity (Custers, 2021). Therefore, the government has a dire need to 'fight for a sustainable, safe, united and healthy future' for the city and its residents (Municipality of Rotterdam, 2016a, p. 13).

The complex geographical nature of the city, which is entirely situated below sea level, has historically called for policies that fight against the adverse effects of climate (Spaans & Waterhout, 2017). To address these growing challenges, the municipality has been actively involved in working toward a climate-proof city by developing and participating in local, regional, national, and international initiatives. In 2014, the city joined Rockefeller Foundation's 100 Resilient Cities program (100RC) and the efforts of fighting against the force of nature culminated in 2016, when Rotterdam presented its Resilience Strategy as a member of the 100RC (Municipality of Rotterdam, 2016a).

Rotterdam already has a reputation for designing and engineering robust systems, but the municipality acknowledges that future risks might call for a different response (Municipality of Rotterdam, 2016a). As such, the Resilience Strategy has not only been motivated by the response to sudden shocks (e.g. earthquakes, fires, floods), but also to face systemic socio-economic tensions. To address the different vulnerabilities, the Resilience Strategy is divided into seven Resilience Goals and six Challenges. Of particular interest to this study is Goal 1 'Rotterdam: A balanced society' which aims to address Challenge 1 'Social cohesion and education'. The Strategy advocates for a more socially cohesive Rotterdam as a way to increase social resilience.

As part of the efforts to achieve Goal 1, in 2012 the municipality developed the National Programme Rotterdam South (NPRZ) (Municipality of Rotterdam, 2016a, p. 109), and in 2016 the Woonvisie (Housing Vision of Rotterdam) (Municipality of Rotterdam, 2016a, p. 65), which aim to create a balance in the groups of residents by demolishing and renewing old cheap housing to make way for higher-income groups. A physical change in the housing structure mutates the composition of a neighbourhood and is presumed to foster social cohesion to achieve the resilience Goal 1 and increase social resilience (see Figure 1.1).

The resilience view of Rotterdam is endorsed by a balanced society that in practice is linked to the housing policy. The housing policy of Rotterdam has, however, been criticised for its apparent motivation to disperse and exclude social groups. Scholars and activists claim that low and middle-income families, the homeless, young people, and ethnic minorities are excluded due to their limited purchasing power (Right to the City, 2021a; Versluis, 2017). Although the municipality claims that advocating for more balanced neighbourhoods will have a positive effect in Rotterdam, some scholars attribute these claims to a misleading use of indicators and statistics (Schinkel & van den Berg, 2011; Uitermark et al., 2017). They argue that political decision-makers frame data to fit their storyline of a balanced society while obscuring controversial motives and negative

1



Figure 1.1: Balance as a remedy: Rotterdam strategy to increase the municipal resilience. Rotterdam's resilience office divides resilience into six dimensions: Social, Energy, Cyber, Climate, Critical infra, and Governance (Molenaar, 2016). Beating the Challenge 1 'Social cohesion and education' contributes to the Social resilience dimension. Source: Author.

outcomes (de Bruijn, 2019). Other studies suggest that the storyline does not even hold, given the lack of demonstrable promised improvements (Hochstenbach et al., 2015). In that sense, some scholars and organisations regard these measures as a transgression of freedom and justice (Uitermark et al., 2017; van Eijk, 2010; van Gent et al., 2018). This sentiment has also reached the public. Detractors of the Woonvisie have organised protests under the movement Recht op de Stad (Right to the City) that rejects the city in balance and champion a city for everyone (Right to the City, 2022). Remarkably, in 2021 the United Nations filed a report that denounced these policies as racist and classist (UN-OHCHR, 2021). As a consequence, it requested an investigation to assess how the policy relates to human rights and called for the application of measures if these are not guaranteed (Right to the City, 2021b).

Taking into account the heavy load of dispute that Rotterdam's balanced neighbourhood policies carry, and in order to elucidate their impact on urban resilience, it is vital to study whether the assumptions on which the municipality grounds their policies are valid.

1.3. KNOWLEDGE GAP

This study aims to address the existing literature gap from the urban resilience angle. In general, academic and governmental publications agree that in cohesive societies, local communities protect residents against threats, care of others during hardships, and ultimately promote community resilience (Arup, 2015; Pelling, 2003; Quigley et al., 2018; Saja et al., 2019). Nonetheless, there are few ex-post empirical studies that confirm this relation. Therefore, we aim to understand if balanced neighbourhoods trigger resilient actions that are pivotal in resilient communities (Doff, 2017; Fay-Ramirez et al., 2015). To this end, this study contributes to the existing literature in three ways: (i) determining if social cohesion acts as the mechanism for resilient actions triggered by the balance in a neighbourhood, (ii) measuring the level of neighbourhood balance in terms of the housing stock value, and (iii) use a multidimensional conceptualisation of social cohesion.

The relationship between balanced neighbourhoods and urban resilience has been scarcely explored. Custers (2021) evaluated the association between balance and willingness to help — which, as we will argue, can be used as a proxy to measure resilient actions —, and the relationship between balance and social capital. Our research elaborates on these results in the three above-mentioned ways: (i), Custers (2021) analysed the relationships between balance, social cohesion, and willingness to help independently, lacking the capability to understand the mechanism between balance and resilient actions; (ii), it considered balance in terms of ethnicity, socio-economic status, or tenure, whereas we go down to the physical level considering house prices; and (iii), by measuring social cohesion as a unidimensional variable. From the existing studies, to our knowledge, only Wang and Kemeny (2022) have empirically studied the relationship between balanced neighbourhoods in terms of the housing stock value and social cohesion. Nevertheless, their study does not address points (i) and (iii). Therefore, little is known about the empirical relationship between balanced neighbourhoods, social cohesion, and resilient action.

1.4. RESEARCH QUESTIONS

This thesis aims to contribute to the literature by measuring and studying the implications of the value of houses as indicators of balance, and performing an empirical analysis of the relationship between balanced neighbourhoods, social cohesion, and resilient action. Taking into account the gaps detected in the literature, the research aims to answer the following main question:

Does neighbourhood balance increase resilient action of neighbourhood residents?

The following sub-questions are considered in order to address this main question:

- SQ1: What are the underlying mechanisms that drive balanced neighbourhoods to act more resiliently?
- SQ2: How can the indicators of the mechanisms that drive balanced neighbourhoods to act more resiliently be operationalised?
- SQ3: What is the empirical relationship between the indicators that drive balanced neighbourhoods to act more resiliently?

1.5. Research Approach

A **deductive (confirmatory) approach** is suitable as the objective is to empirically test the mechanism that triggers resilient action in balanced neighbourhoods (Hair et al., 2016; Henseler, 2020). To this end, the research aims to test the assumptions on which the municipality of Rotterdam grounds their balanced neighbourhood policies. Following the deductive approach (Dudovskiy, 2016), first a theory is deduced from the literature; second, hypotheses on the relationship between the variables are formulated; third, the hypotheses are tested using quantitative methods; fourth, the theory is accepted or rejected, and recommendations to policymakers to adapt their resilience governance if necessary are provided. Furthermore, by virtue of the vastly detailed data available in the Netherlands, a comprehensive geo-spatial investigation can be carried out. We combine resident and neighbourhood census tract data to unravel the effect of contextual and individual factors. More specifically, the deductive approach is based on Partial Least Squares—Structural Equation Modelling (PLS-SEM) and spatial econometrics, two statistical modelling techniques which provide empirical evidence for the research hypotheses.

A deductive approach is advantageous as it starts with a clear hypothesis (Schwab & Held, 2020; Shih & Chai, 2016). This avoids stating a hypothesis after using a data-driven approach to fit the data. In addition, it also prevents selectively reporting what works and ignoring what does not. For instance, previous neighbourhood effects studies use a data-driven approach (e.g. Abada et al. (2007), Custers (2021), and Wang and Kemeny (2022)) which can test various covariates until the model fit is good enough to consider satisfactory results. On the other hand, the limitation of a deductive approach is its reliance on the theory on which the hypotheses are grounded (Hair et al., 2016; Henseler, 2020). A strong theoretical foundation based on the literature review can help mitigate this limitation (Manley et al., 2011; van Ham & Manley, 2010). In addition, expert interviews were performed to increase the confirmatory power of the results (Barth et al., 2011).

1.6. RESEARCH STRUCTURE

The report consists of six chapters. Figure 1.2 graphically depicts the structure of the thesis. Chapter 2 is dedicated to a literature review that provides the theoretical foundations for the research. The chapter concludes with the formulation of the hypotheses to test, grounded on the literature review, and answers SQ1.

Chapter 3 describes the data and methods used in the development of the study. The chapter concludes with the computational framework to be applied to the case study, i.e. the way of how the data and methods are used to test the research hypotheses and respond to the research questions.

Chapter 4 presents the outcomes of applying the research methods to the Rotterdam case. This chapter provides empirical evidence of the acceptance or rejection of the research hypotheses, and answers SQ2 and SQ3.

Chapter 5 provides an interpretation of the obtained results based on the theoretical concepts grounded in literature and the input of interviews to experts.

Finally, Chapter 6 gives an answer to the posed research questions and formulates policy advice. It also discusses the limitations of the study and suggests directions of future research.



Figure 1.2: Research flow diagram.

2

LITERATURE REVIEW

A literature review was necessary to break down, conceptualise, and operationalise balanced neighbourhoods, social cohesion, and resilient actions. For this, we relied on academic literature as well as government documents. On one hand, the concepts are very debated in the academic world, with contrasting views and ideas. Desk research is useful in creating an understanding of concepts. On the other hand, government documentation that defines or discusses the concepts is critical in order to "speak" the language of the politicians.

The first section of this chapter is concerned with the reconciliation between social cohesion and resilience. The chapter follows with the current understanding of housing policies as a way to increase social cohesion by achieving a social mix of the population. Next, the state of the policies in Rotterdam is discussed and their understanding of a balanced neighbourhood explained. Finally, the chapter concludes with the formulation of the hypotheses to test, grounded on the literature review.

2.1. URBAN RESILIENCE

This section first introduces the concept of urban resilience and explains the characteristics of a resilient city. Next, it exhibits how social factors govern the resilience of a community to argue that informal support is a voluntary action characteristic of resilient cities.

2.1.1. THE RESILIENT CITY

The rise in urban stresses throughout the world, coupled with the consequences of climate change, such as extreme weather and natural catastrophes, has prompted the need for preparedness of urban areas to face precarious circumstances (UNISDR, 2015). Consequently, the concept of urban resilience has grown in popularity to tackle these challenges that abrupt urbanisation and a lack of proper planning have put in place.

Urban resilience is ill-defined, and the variety of conceptualisations make it hard to put it into practice (Larimian et al., 2020; Meerow et al., 2016). Here, we under-

stand urban resilience as 'the ability of an urban system and all its constituent socioecological and socio-technical networks across temporal and spatial scales to maintain or rapidly return to desired functions in the face of a disturbance, to adapt to change, and to quickly transform systems that limit current or future adaptive capacity' (Meerow et al., 2016, p. 39). This definition subsumes a continuum of actions that consider different timescales of the system. First, a resilient urban system should be able "to maintain ...desired functions" during a shock. This means they should have the coping capability to absorb and withstand an adverse event. Next, the city should be able "to adapt" to the new conditions by learning from their experience and later on "to transform" if required (see Figure 2.1). Shocks are regarded as opportunities to address long-lasting inequity problems such as poverty concentration, however, persistence may still be desirable for certain components such as infrastructure. Therefore, a resilient city is a city that can sustain, adapt, and develop under a variety of unexpected shocks and stresses.

	Phase of resilience	Characteristic	
	Before an event	Preparedness	
ne	During an event	Absorption	
ţ	(Recovery	
	After an event {	Adaptability	
	l	Transformability	

Figure 2.1: Phases and characteristics of urban resilience. The downward time arrow indicates the order of phases. Source: Tong (2021)

2.1.2. INFORMAL SUPPORT AS RESILIENT ACTION

It has been generally agreed upon that **social factors govern the resilience of urban areas** (Copeland et al., 2020). Aldrich (2017) goes as far as to state that resilience comes from responses built on communities' social networks rather than on complicated engineering. Within this idea, social resilience 'concerns the extent to which a local community is able to respond to or anticipate changes' (Doff, 2017, p. 4). At this level, resilience is shaped by the structure of social networks, the access to resources, and the social institutions (Adger et al., 2002). Resilience can only be present if the members in a community are capable of collective action. Only in this way, they can collect enough resources and have the capacity to mobilise them in the event of a disaster or to address a chronic stress (Veld Academie, 2021).

Communities' everyday resilience to unanticipated adversities are therefore fostered by pro-community behaviours. Communities in which the residents work together and have common goals find it easier to understand the needs of others and share necessary resources. In the event of a disaster, the existence of shared norms and trust result in a stronger willingness to cooperate and in a better cooperation (Aldrich, 2017; Baldwin & King, 2017).

How do we then know how resilient a community is? Unfortunately, **resilience cannot be measured directly**. Therefore, studies employ proxy indicators to construct resilience variables or, otherwise, measure phenomena that are indirectly related to resilient capabilities (Copeland et al., 2020). In line with Aldrich (2017), we argue that the end mechanism through which social resilience is achieved is mutual aid. A resilient community is one that can collaborate and is caring. Therefore, **informal support** (or informal help) is a suitable proxy to measure the resilient action of a neighbourhood.

Informal support is essential before and after disasters, as local care systems cannot work without the support of the local community. It contributes to desaturate local care systems, as well as to be a *locum tenens* when these normal providers of assistance are shut after a disaster (Aldrich, 2017; Naganuma et al., 2018). Informal support is characterised by emotional and instrumental support (Noguchi, 1991). Forms of emotional support are for example listening to the worries of someone else, giving a hug, or saying encouraging words. On the other hand, forms of instrumental support are for example lending money, watching someone else's child, or helping with moving.

Residents with strong connections before a disaster are more likely to ask and receive informal support than those who are not in a cohesive community. After a disaster, it is too hard to form connections and trust in such a compressed and tense time (Aldrich, 2017). Therefore, Doff (2017) distinguishes between **effective and potential resilient ac-tions**. Effective actions are those which happen only after a stressor has occurred. Potential action refer to the 'potential' to act resiliently. If only effective action is considered, potential resilience is missed. If only potential action is considered, obstacles to actual action are not taken into account. As such, it is important to consider both effective and potential informal support.

Informal support can come from the hand of **strong ties** like friends and family, from **weak ties** like neighbours and acquaintances, or **invisible or non-existent ties** like strangers (Felder, 2020). The CBS argues that the accessibility to informal support is greater with strong ties, but that most Dutch citizens still have access to informal support from weak ties (CBS, 2012, 2015). In this sense, **social cohesion is posited to be the mechanism underlying informal support from weak, invisible, or non-existent ties**. However, the National Academies of Sciences, Engineering, and Medicine (2020) alerts of the decrease in weak ties relationships and of the possible consequences in informal support. As such, urban planning policies are concerned about strategies to promote weak ties in the community.

Previous empirical studies have related informal support to the resilience after a disaster. For example, Aldrich (2017) shows how neighbourhoods in New Orleans had to rely on the support from other locals to recover, especially because shops and grocery stores lacked resources for more than a year after hurricane Katrina. Another study by L. Cheshire (2015) analysed the extent to which weak ties helped each other in the 2011 floods in Queensland, Australia. Finally, Klinenberg (2015) studied how the capacity of response and recovery of those more vulnerable to heat waves in Chicago was raised by the care provided by those around them.

2.2. How can social cohesion improve resilience?

The starting point of this section is the justification in resilience thinking of social cohesion as the mechanism that triggers informal support. The section follows with a clarification on what we understand by social cohesion. This is especially necessary because of the cross-disciplinary use of the concept between social sciences and policymaking. To this end, we identified the key characteristics of social cohesion that have been explored in the academic literature and policy discourse. The section finalises with the review of urban elements that have been recognised to have an impact on social cohesion and informal support.

2.2.1. FROM SOCIAL COHESION TO RESILIENT ACTIONS

Social cohesion has been acknowledged to contribute to the general well-being of society. Cohesive societies are characterised by their political stability and their economic and business growth (OECD, 2011). Exclusion and marginalisation are reduced by minimising the disparities among social groups, and members of society are offered the opportunity of upward mobility.

Resilience thinking defines as **resilient city, the city with social cohesion** (Center for American Progress, n.d.; Forrest & Kearns, 2001; Larimian et al., 2020). Literature suggests that in cohesive societies, local communities protect residents against threats, care of others during hardships, and ultimately promote community resilience. Social bonding not only provides for daily activities, but also facilitates access to resources and information through social connections (Forrest & Kearns, 2001). Strong social cohesion is associated with the ability of members to be able to cooperate and take actions to support each other in the presence of adverse events (Baldwin & King, 2017; Elliott et al., 2010; Larimian et al., 2020). As such, voluntary action in terms of **informal support** is deemed an important factor of a so-called resilient city (G. Huang, 2020). Therefore, governments are looking to strengthen the relationships between citizens, communities, and social institutions to boost social cohesion.

Past studies have found a positive relationship between social cohesion and community resilience. For example, Berkman et al. (2000) developed a conceptual framework that explains how social networks enhance informal support like having someone to talk to or to ask for a ride. An empirical study by Kawachi et al. (1997), found out that higher social cohesion was related to lower mortality, partially explained by the increase in income inequality with the reduction of social cohesion. Research by Sampson et al. (1997) also showed empirical evidence that socially cohesive communities influence behaviour through social norms and as a result have better access to medical care, healthy food, and exercise facilities. This effect of social contagion is especially relevant in this study due to the **geographically bounded conceptualisation** within neighbourhoods. Finally, in a study of heat waves in Chicago, Klinenberg (2015) concluded that during natural disasters like heat waves, elder citizens in poorly cohesive neighbourhoods lack social support from caring neighbours who will check on them, as well as they know of fewer safe common locations where they may seek help and shelter.

2.2.2. ORIGIN AND DEFINITIONS OF SOCIAL COHESION

Resilience thinking associates a cohesive society with a resilient society. But, what is social cohesion? Several authors attribute the beginning of the discussion on social cohesion to Durkheim's (1897) *Étude de sociologie* (Fonseca et al., 2019; Hulse & Stone, 2007). Since that starting point, sociologists and social psychologists have been working on the definition and development of social cohesion as a construct. A century after the term was coined, since the 1990s, policymakers in developed countries have utilised social cohesion to justify public spending on social policies and economic growth (Hulse & Stone, 2007; Klein, 2013). Despite its long-lasting presence in the literature and government policies, social cohesion remains an ill-defined concept whose definition varies between authors and institutions (Chan et al., 2006). Bernard (1999) already emphasises this difficulty when he considers social cohesion as 'a quasi-concept, that is, one of those hybrid mental constructions that politics proposes to us more often in order to simultaneously detect possible consensuses on a reading of reality, and to forge them.' (as cited in Chan et al., 2006; Klein, 2013).

In order to have an overview of the different conceptualisations of social cohesion, Table A.1 and Table A.2 show definitions coined by scholars and political institutions, respectively. These definitions have been collected from previous literature reviews on social cohesion (i.e. Fonseca et al. (2019), Hulse and Stone (2007), Klein (2013), Langer et al. (2017), and Schiefer and van der Noll (2017)) and from the core research papers of the study.

From the evaluation of the literature, we can see in Table A.1 and Table A.2 that there is no single definition of social cohesion. Nevertheless, some similarities can be identified from the multiple definitions. For instance, most authors believe that social cohesion is a positive feature of a social entity, and therefore it is not an individual quality (Schiefer & van der Noll, 2017). These definitions are mostly related to the economic components of society such as well-being, trust, and equitable opportunities (Fonseca et al., 2019). Furthermore, Table A.1 exhibits how social cohesion is a characteristic of a social entity that emerges from the individuals. Finally, Table A.2 also shows that social cohesion does not encompass a single policy concept. This strengthens the idea that social cohesion is viewed as a multidimensional construct.

2.2.3. SOCIAL COHESION CONCEPTUAL FRAMEWORKS

Beyond the definitions of social cohesion, policymakers have been especially interested in monitoring the social cohesion of citizens. Several conceptualisations can be found in the literature that indicate the lack of agreement on how to approach this issue.

In general, the authors have embraced the above-mentioned consideration of a multidimensional construct. For example, Forrest and Kearns (2001) understands social cohesion as being composed of Common values, Social order, Social solidarity, Social capital, and Identity. Later, Chan et al. (2006) developed a two-by-two framework that considers a Horizontal dimension (cohesion with society) and a Vertical dimension (statecitizen cohesion) divided between a Subjective component (state of mind) and an Objective component (behavioural manifestations). Finally, Langer et al. (2017) goes on to simplify that social cohesion can be characterised by the dimensions of Inequalities, Trust, and Identity.

The discrepancy within the construct of social cohesion is clear. Nevertheless, Schiefer and van der Noll (2017) indicate that there is far more overlap than is assumed. In an effort to reduce the concept to its essential dimensions by analysing previous conceptualizations, they argue that many elements identified in the literature are outcomes of social cohesion rather than constituting elements. In this way, shared values, inequality, and quality of life are not inherent in the concept, but are consequences of social cohesion. For instance, they argue that a society is not cohesive because individuals feel well, but individuals feel well because there is a strong social cohesion. The direction of causality frames social cohesion as an antecedent to well-being. As a result, they characterise social cohesion as a construct of Social relations, Attachment/Belonging, and Orientation towards the common good.

Taking into account the multiple ways to interpret and put together the multiple dimensions of social cohesion, three important considerations were identified in the literature to define an operationalisable conceptualisation for social cohesion: (i) type of construct, (ii) variables, and (iii) geographic level of analysis.

Beginning with (i), the literature agrees that social cohesion is composed of several distinct characteristics, thus it should be treated as a construct (Langer et al., 2017; van Beuningen & Schmeets, 2013). To operationalise such a multidimensional concept, we can distinguish between two main approaches: measurement theory and synthesis theory (see Figure 2.2). Measurement theory aims at revealing the structure of empirical data by analysing the common factors among the data that conform to the construct (Henseler, 2020). In this sense, the theoretical concept is first developed and then connected to the empirical data, usually for confirmation purposes. In measurement theory, the construct is regarded as a **latent variable** from the underlying common factors, where the unobserved latent concept causes the observed data and their relationships (Henseler, 2020). This direction of causality is known as reflective (Skrondal & Rabe-Hesketh, 2004, p. 77). On the other hand, synthesis theory is a pragmatic strategy that conceives these unobservable variables in terms of what is already familiar and well known (Henseler, 2020). Instead of finding the common factors within the data, synthesis theory creates emergent variables by forging the different components as a whole, instead of being an assembly of parts. While measurement theory is usually used for confirmatory purposes, synthesis theory is an instrumental approach to solve and analvse problems. For this reason, emergent variables rely as much as possible on the results from previous measurement theory studies (Henseler, 2020). This direction of causality is known as formative (Skrondal & Rabe-Hesketh, 2004, p. 77).

The distinction between these perspectives demands a comprehensive examination of the grounds for determining the approach. Although synthesis theory is mostly used for social constructs (widely seen in the conceptualisation of socioeconomic status) (Diamantopoulos & Winklhofer, 2001), some examples of measurement theory can be found, such as Dragolov et al. (2013).

In the Dutch context, we can find the social cohesion construct of Netherlands Statistics (CBS, 2015). Based on the work of van Beuningen and Schmeets (2013), CBS (2015) constitutes one of the two official Dutch social cohesion frameworks. By means of measurement theory, it includes the following three dimensions: Participation, Trust; and Integration. In their framework, Participation and Trust are subdivided within social, organisational, and political. Integration is not subdivided (see Figure 2.4). Studies like Schiefer and van der Noll (2017) and Dragolov et al. (2013) take a similar approach.

Within the Dutch context, we can also identify the study of Schnabel et al. (2008), that uses synthesis theory to conceptualise social cohesion for the Dutch Office of Social and Cultural Planning. Instead of defining all the different factors that give place to the latent subdimensions, the study makes use of the previously validated results of measurement



Figure 2.2: Diagrams of the two theories for the development of constructs discussed in this study. Source: Henseler (2020).

theory to forge an emergent variable for social cohesion. They devised social cohesion as an emergent variable composed of the elements in Figure 2.3. These elements have previously been validated as the pieces that forge social cohesion from measurement theory. Other studies that take the same approach from similar components are Baldwin and King (2017), Miao et al. (2019), and Stuij (2020).

Social cohesion in Schnabel et al. (2008). Each item ranked on a Likert scale as:			
(1) Totally disagree, (2) Disagree, (3) Nor agree or disagree (4) Agree and (5) Totally agree.			
i. I have a lot of contact with my immediate neighbors			
ii. In this neighborhood, people treat each other in a pleasant way			
iii. We live in a cozy neighborhood with a lot of togetherness			
iv. People hardly know each other in this neighborhood			
v. I am satisfied with the population composition in this neighbourhood			

Figure 2.3: Social cohesion indicators in Schnabel et al. (2008) for the Dutch Office of Social and Cultural Planning.

Finally, it is also important to mention that many other studies do not use a construct to determine social cohesion (see Chan et al. (2006), Sluiter et al. (2015), or Tolsma et al. (2009)). These investigations analyse several individual variables, which can be used as proxies to define social cohesion. Generally, these studies are concerned with correlational analyses that do not just want to measure social cohesion, but want to use the measurement in further analyses. These studies acknowledge the limitation of such an approach.

Once the dimensions and the type of construct have been conceptualised, it is also

Indicators	Variables	Subdimensions	Dimensions
Frequency of contacts			
Social life satisfaction	Family and sta		
Superficiality of contacts	Family contacts		
Feeling of being understood			
Frequency of contacts		-	
Social life satisfaction	Friend contacts		
Superficiality of contacts	Filend contacts	Social participation	
Feeling of being understood			
Frequency of contacts		-	
Social life satisfaction	Neighbor contacts		
Superficiality of contacts	Neighbor contacts		
Feeling of being understood			
Given unpaid help	Informal help	-	Derticipation
Type of association	Participation associations		Participation
Frequency of participation	rancipation associations	Organizational participation	
Type of volunteering	Volunteering	-	
Voting turnout	Voting		
Radio, TV, or newspaper follower		-	
Engaged in a political party or organization	Political activities	Political participation	
Attended consultation meeting or hearing			
Contacted a politician or official			
Participated in activism			
Participated in a demonstration			
Participated in a signature campaign			
Participated in a political action via Internet or email			
Has done something to raise political issue			
Generalized trust on people	General trust	Social trust	
How much trust on army	Army trust		
How much trust on police	Police trust	-	
How much trust on judges	Judges trust	Organizational trust	Truct
How much trust on civil servants	Civil servants trust	- Organizational trust	IIust
How much trust on press	Press trust		
How much trust on large companies	Large companies trust	-	
How much trust on the Parliament	Parliament trust	Political trust	-
N/A	N/A	N/A	Integration

Figure 2.4: Dimensions and indicators used in CBS (2015) to determine social cohesion. Indicators for Integration are not available because in the study, Integration is inferred by indirectly comparing the scores of Participation and Trust of the various population groups. Adapted from CBS (2015).

relevant to determine (ii), the **variables** which define social cohesion. On the premise that indicators need to be based in both qualitative data and a theoretical model (Townshend et al., 2015), a wide range of indicators for social cohesion can be found in literature. To give two relevant examples, we show the indicators used in the two social cohesion frameworks used by the Dutch government. First, Figure 2.4 compiles the indicators used in CBS (2015). Given that the study bases their definition of social cohesion on measurement theory, the framework consists of both behavioural and attitudinal indicators for each dimension. It can be seen, however, that there are no indicators for the Integration dimension. This is because their framework infers Integration by indirectly comparing the scores of Participation and Trust of the various population groups. On the other hand, Figure 2.3 shows the indicators used in Schnabel et al. (2008) based on synthesis theory. The various dimensions of social cohesion as conceptualised in measurement theory are reduced to single indicators for each of the dimensions. They express the behaviour, perception, and participation of individuals as members of society.

Finally, consideration (iii) the geographic level of the analysis, is also of relevance
when conceptualising social cohesion. Sociologists do not agree at which geographical level social cohesion is properly defined and measured (Harell & Stolle, 2010). While some believe social cohesion is a small community phenomenon, others argue that it can be best understood at a national level. This entails that sources and outcomes of social cohesion will vary accordingly. Consequently, this had led to the study of a variety of geographical levels in social cohesion research spanning from countries (CBS, 2015; Langer et al., 2017), going through municipalities (Islam et al., 2008) and neighbourhoods (Custers, 2021; Wang & Kemeny, 2022), all the way down to postcodes (Tolsma et al., 2009). What is more, several authors indicate that social cohesion crosses the physical boundaries of country, municipality, or neighbourhood, and more important are the form and content of social networks (Forrest & Kearns, 2001; Takagi & Shimada, 2019). Given that in Rotterdam's policy context policies are targeted at a neighbourhood level, this study will consider social cohesion a characteristic at a neighbourhood level.

In this research, social cohesion is treated as a multi-dimensional construct. Both measurement theory and synthesis theory are explored to find out which conceptualisation fits better the available data. In addition, neighbourhoods are considered the focal points for shaping and strengthening the social cohesion of cities insofar we are focusing on a spatial-bounded problem (Forrest & Kearns, 2001).

2.2.4. URBAN ELEMENTS THAT SHAPE SOCIAL COHESION AND INFORMAL SUPPORT

As a final consideration, both social cohesion and informal support have been associated to **characteristics of the built environment and the demographics**. This is based on the idea that urban elements can promote or deter social interactions, and thus are aspects of what can be considered a resilient neighbourhood (Mouratidis & Poortinga, 2020; Wood et al., 2010) (see Figure 2.5). However, results of these studies show mixed results.

To begin with, social demographic factors have been previously accounted for variance in neighbourhood social cohesion. **Ethnic heterogeneity** has been identified as positively associated with social cohesion (Tunstall & Lupton, 2010), negatively associated (Coffé & Geys, 2006), and with non-significant effect (Abada et al., 2007; Musterd & Andersson, 2005). The effects of ethnic heterogeneity on social cohesion and support can be explained by the contact hypothesis and the homophily principle developed later in Section 2.3.2. A second relevant type of mix is **house type heterogeneity**. House type heterogeneity has been less explored empirically, but qualitative studies and policy documents argue that a more heterogeneous range of house types allows residents to stay in the neighbourhood as they move in life stages, thus increasing the cohesion to the community (Bolt et al., 2010; Galster et al., 2015). Finally, most literature agrees that long-term residence is strongly associated to social aspects like integration, capital, and cohesion (Rogers & Sukolratanametee, 2009; Wood et al., 2010), but there is also evidence of studies like Small (2007) which contradict the status-quo.

The land use has been regarded as another relevant factor. First, higher **population density** increases the chances of social interaction, and as Rogers and Sukolratanametee (2009) concluded, density had a positive effect on local support. However, in the study by Koohsari et al. (2021) these characteristics were all associated with a negative score. Next,



Figure 2.5: Conceptual model of the association between urban elements and social cohesion. Adapted from: Wood et al. (2010)

the amount of **amenities** has also been argued to affect social cohesion as they provide a space for encounter and interaction (Mouratidis & Poortinga, 2020; Veld Academie, 2021).

Finally, the built environment has also been considered in previous studies as a relevant factor (Mazumdar et al., 2018; Mouratidis & Poortinga, 2020). According to previous research, **greenery** may also promote social interaction and hence build social cohesion (Hartig et al., 2014). Second, Cabrera (2013) argued that street encounters can increase local social interaction and strengthen ties within the community. Therefore, the **walk-ability** of a neighbourhood has also been considered as a relevant built environment factor.

Urbanism literature, notices that demographic, built environment and land use elements are not sufficient to understand their effect on the social cohesion of neighbourhood residents (Kim & Kaplan, 2004; Rogers & Sukolratanametee, 2009; Wood et al., 2010). Individual characteristics such as intentions and preferences regulate the perception and experience of urban elements. In addition, urban design elements that encourage interaction (e.g. street width, the interconnectedness of streets, the presence of porches and balconies, or physical aesthetics) have been associated with a higher sense of community.

2.3. BALANCED NEIGHBOURHOODS

In the last section, we saw how socially cohesive areas have a stronger willingness to cooperate in the event of a disaster. But, how do we achieve a socially cohesive community? In this section, we develop the idea of balanced neighbourhoods as a policy instrument to achieve a more resilient city by improving its social cohesion. Balanced neighbourhoods are ill-defined and context dependent, but generally aim for a mix of local demographics with the objective of promoting inclusion, liveability, and the economy (Tunstall & Lupton, 2010). In this section, we start by explaining the roots of balanced neighbourhoods from the concept of social mix. Then, we relate social mix to social cohesion by showing relevant studies that connect both concepts. We finally follow up by taking social mix to the housing policy realm.

2.3.1. SOCIAL MIXING

Governments and scholars generally agree that residing in a deprived neighbourhood has a detrimental impact on people's life opportunities. Previous research has identified a relation between deprived neighbourhoods and a variety of negative outcomes such as higher school dropout rates, lower childhood achievement, lack of social mobility, deviant behaviour, and social exclusion (Manley et al., 2011). These so-called **neighbourhood effects** have been explained by deprived neighbourhood characteristics such as social and physical disconnection, poor culture, discrimination, limited public accessibility, and high exposure to criminal activity (van Ham & Manley, 2010).

Negative neighbourhood effects have often been associated with the segregation of poor communities or ethnic minorities. Large concentrations of these groups in specific areas are believed to reinforce and perpetuate poverty and exclusion (Bolt et al., 2010; Colomb, 2011). This relation was embraced by policy-makers, as it implies that eliminating such concentrations would address negative neighbourhood effects. This accessible and actionable solution has led to the development of area-based policies to create socially mixed neighbourhoods (Méreiné-Berki et al., 2021).

Social mixing aims to prevent stigmatised groups from congregating in one location and subsequently provide better conditions for generating social capital to restore or create conditions of integration, cohesion, workability, and sustainability (Musterd & Andersson, 2005; Tunstall & Lupton, 2010). Figure 2.6 summarises the alleged positive effects of social mix. First, the local economy and employment would be boosted as a result of improved local public and private services. Second, the behaviour and aspirations of segregated communities should improve. Third, an improvement in community bonding would increase neighbourhood stability. Fourth and final, we would see an improvement in bridging with other communities as a result of the loss of stigma. This objective can be achieved in different ways: through the mixing of household types and tenures (Bolt et al., 2010; van Gent et al., 2018); occupation and income levels (Galster et al., 2015; van Ham & Manley, 2012); ethnicities and birthplaces (Jordan, 2018); and languages and cultural backgrounds (Wang & Kemeny, 2022).

A key problem for those who have consistently advocated for social mixing is that this point of view has relied on a **theoretical rather than empirical evidence** base. Despite its theoretical foundation and broad adoption in urban policy, several scholars have questioned the goal of socially mixed neighbourhoods on conceptual and practical grounds (Galster et al., 2015; Graham et al., 2009). Opposing scholars suggest that there is a paucity of evidence that social mix is the answer to neighbourhood effects, and most empirical studies point in the opposite direction. Findings by Kearns and Mason (2007) suggest that mixed-tenure neighbourhoods do not guarantee that neighbourhood problems such as vandalism, graffiti, crime or littering will be reduced. Also, Graham et

Economic and service impacts	Community-level effects
Better quality public services	Increased social interaction
Improved quality & quantity of private services	Enhanced sense of community and place attachment
Enhanced local economy	Reduction in mobility
Increased rates of employment	Greater residential stability
Social & behavioural effects	Overcoming social exclusion
Social & behavioural effects Reduction in anti-social behaviour	Overcoming social exclusion
Social & behavioural effects Reduction in anti-social behaviour Better upkeep of properties and gardens	Overcoming social exclusion Reduction in area stigma
Social & behavioural effects Reduction in anti-social behaviour Better upkeep of properties and gardens Raised aspirations	Overcoming social exclusion Reduction in area stigma Increased connectivity with other places Enhanced again patterning

Figure 2.6: Alleged benefits of social mix. Source: Kearns and Mason (2007)

al. (2009) find little support for mixed-tenure neighbourhoods as good for social wellbeing. A study by Bricocoli and Cucca (2016) argues that strategies to achieve mixedincome neighbourhoods foster exclusion from social housing. Similarly, Custers (2021) concludes that balanced neighbourhoods in Rotterdam do not positively relate to employment for the less well-educated. Finally, Musterd and Andersson (2005) indicate that the housing mix did not lead to a considerable social and ethnic mix, even after almost forty years of policy implementation in Sweden.

The success of social mix in the political discourse has also been attributed to the neoliberal strategy that aims to perpetuate capitalism in cities (van Eijk, 2010). By means of these policies, governments try to ensure that stigmatised groups do not congregate in one location with the objective of luring middle-class households and keeping lowerclass households away (Uitermark et al., 2017). A mixed neighbourhood is therefore an "upgraded" neighbourhood where affluent people live (Versluis, 2017). The end objective is not a just city, but real estate development. This has led to the belief that the pursuit of social mix may be a waste of public funds and resources that diverts attention away from or even worsens structural inequalities (Colomb, 2011; van Gent et al., 2018).

2.3.2. FROM SOCIAL MIX TO SOCIAL COHESION

Departing from the **assumption that social mix fosters social cohesion**, urban policymakers strive to create socially heterogeneous neighbourhoods (Méreiné-Berki et al., 2021). Social mix policies have been marketed as solutions to encourage class, racial, ethnic or religious cohesion to prevent increasing segregation (Tunstall & Lupton, 2010). However, several studies that focus on the relationship between social mix and social cohesion also gainsay the claims of the status quo.

Research with a qualitative approach, such as Curley (2009), shows that the social capital of women did not benefit from relocation to mixed-income Boston neighbourhoods. Similarly, Chaskin and Joseph (2011) indicate that mixed-income communities resulted in more division and isolation in Chicago. Furthermore, Bolt et al. (2010) conclude that mixed-tenure neighbourhoods do not meet the expectations of the policy on social cohesion.

There is a paucity of studies on the relationship between socially mixed neighbourhoods and social cohesion at a quantitative level. Notably, most of these studies challenge the positive relationship advocated by the status quo. For example, Abada et al. (2007) did not find significant effects on the social cohesion of youth living in ethnically balanced neighbourhoods in Canada. Furthermore, Coffé and Geys (2006) did not find a correlation between income-mix and social capital in Flemish neighbourhoods and even indicates that ethnic balance had a negative effect on social capital. Finally, Wang and Kemeny (2022) explored the impact of different balance indicators in China to obtain contrasting results, although they found that mixed tenure or educational backgrounds are associated with higher social cohesion, a mix in income or birthplace indicates the opposite.

The concept of **social capital** can help understand both contradicting stances. The first efforts to define social capital can be traced back to the work of Bourdieu (1986). However, it was Putnam (2000) who laid the ground for the contemporary policy approach of social capital (Forrest & Kearns, 2001). According to Putnam, 'social capital refers to connections among individuals—social networks and the norms of reciprocity and trustworthiness that arise from them.' (Putnam, 2000, p. 19). In his work, he conceptualised social capital as being composed of **bonding** social capital (inward looking, promoting homogeneity) and bridging social capital (outward looking, promoting heterogeneity). Later, Woolcock (2001) included a third component, linking social capital, which refers to the ties across the individual and an authority in society to gain access to valuable resources (Claridge, 2018).¹ The bonding, bridging, and linking properties of social capital can help understand the contradicting stances of the effects of social mix. First, the **contact hypothesis** argues that mixing communities can increase the bridging and linking of marginalised groups, such that ties between heterogeneous groups lead to overcoming social differences (Allport et al., 1954). In socially mixed areas, different groups meet and interact with each other, and eventually this reduces hostility toward out-groups. On the other hand, according to the **homophily principle** (Putnam, 2000), even though segregation of stigmatised communities can decrease the bridging and linking of marginalised groups with other communities, homogeneity can create bonding within the community (Méreiné-Berki et al., 2021). In this second view, people prefer to be surrounded by those who resemble them, thus social cohesion in mixed neighbour-

¹Even if social capital and social cohesion appear similar at first glance, the examination of their definitions reveals a contrast between these concepts. One fundamental difference present in most definitions of social capital is that social capital is created by collective and individual actors in order to provide future returns that, like any other form of capital, are individually appropriated (Klein, 2013). This capital is generated by investing in social networks, norms, and trust. In this way, the accumulation of social capital refers to the accumulation of resources available to individuals which can yield, for instance, health returns from the help and support from others when being sick (CBS, 2015; Klein, 2013). While social capital is regarded as a voluntary bottom-up approach, social cohesion is on the other hand seen as an involuntary top-down product of the form and quality of shared values, trust, and the relationships among the individuals of a society (Forrest & Kearns, 2001; Klein, 2013). Social cohesion does not aim to accumulate; instead, it refers to the reduction of gaps between individuals within a community and between communities. Therefore, social cohesion is more than the sum of its parts, i.e. more than the sum of social capital and gap reduction (Hulse & Stone, 2007). The ambiguous interpretation and arbitrary application of the terms have consequently led to their overuse and vagueness in the policy arena (Forrest & Kearns, 2001). This can be seen, for example, in the interchangeable use by The World Bank of the terms social capital and social cohesion (Schiefer & van der Noll, 2017). The CBS (2015) Dutch social cohesion framework takes another approach and considers the dimensions of Participation and Trust as the building blocks of social capital which, together with Integration, compose social cohesion (see Figure 2.4).

hoods is affected by one's aversion to the alien (Wang & Kemeny, 2022). In other words, birds of a feather flock together.

Finally, an even smaller body of literature has studied the relationship between socially mixed neighbourhoods and informal support. A relevant example comes from the Rotterdam context in which Custers (2021) found that socio-economically mixed neighbourhoods do not foment bridging between social groups, and that even if there exist mixed social networks in a neighbourhood, these are not strong enough to understand the needs of others and share necessary resources in the event of need. As a result, more evidence is necessary to unravel the relationship between the policy and the observed effects.

2.3.3. BALANCED HOUSING

Governments have developed a variety of policies to increase the social mix of specific areas. Interventions through the housing mix are among the most widely employed to avoid the clustering and segregation of disadvantaged households because of their ability to diversify households that reside next to each other (Hananel et al., 2022). These mix policies have been labelled with multiple terms that have been used interchangeably like social mix, housing mix, balanced community, social balance or balanced neighbourhood (Manley et al., 2011; Ruiz-Tagle, 2019). We will refer to these mixing strategies as **balanced neighbourhood** policies, because that is the term in the political discourse of the municipality of Rotterdam (Municipality of Rotterdam, 2016b).

Western governments have explicitly adopted balanced housing policies as part of their agenda, including those in the Netherlands, the United Kingdom, the United States, Germany, France, Finland, and Sweden (Bolt et al., 2010; Chaskin & Joseph, 2011; Kearns & Mason, 2007; Manley et al., 2011; van Ham & Manley, 2012). Figure 2.7 shows the conceptual framework that current urban policy-makers explain for the link between balanced neighbourhood policies as the alleged means to achieve a complete community, social equity, diversity and social mix. A physical change in the housing structure mutates the composition of a neighbourhood and is presumed to foster social contact and interaction to achieve the alleged beneficial goals.

A wide range of balanced neighbourhood policies have been developed. A popular strategy is to promote heterogeneity in housing tenure, i.e. a mix between different types of housing such as private rental, public rental, or private ownership. Examples of such policies are social housing management and tenant-based housing allowances (Galster et al., 2015). The theory is that combining homeowners and social tenants in a neighbourhood will generate a more diversified social mix that reduces the negative neighbourhood effects (Musterd & Andersson, 2005).

In line with the previous strategy, some governments have also opted to apply tenant allocation reforms. By means of new legislation, households can be banned the choice to move to certain housing types or neighbourhoods (Galster et al., 2015). A good example of such policy is the Act on Extraordinary Measures for Urban Problems implemented in problematic neighbourhoods in the city of Rotterdam.

Another popular, but very contested, balanced neighbourhood policy is urban renewal. These strategies aim to deconcentrate stigmatised groups while attracting inhabitants with greater status and better incomes by selling, refurbishing, or demolishing



Figure 2.7: Conceptual model of balanced neighbourhood policies as the alleged means to the ends in the orange boxes. Solid lines represent direct evidence, whereas dashed lines circumstantial evidence. Adapted from: Perrin and Grant (2014)

public housing and installing more affluent houses (Doff, 2017; van Gent et al., 2018). In addition to the benefits mentioned above that balanced housing can provide through social mixing, proponents of these policies also argue that increasing the number of high-quality homes in a neighbourhood gives housing career prospects for the socially mobile, thus promoting intergenerational cohesion (Custers, 2021).

Studies that analyse the link between housing and social cohesion have come to both positive and negative relationships (Bolt et al., 2010; Kearns & Mason, 2007). As a consequence, most authors have concluded that balanced housing does not result in tangible benefits (P. Cheshire, 2009; Colomb, 2011; Galster et al., 2015; Klein, 2013). Moreover, some authors point out how these regulations might contribute to discriminatory housing market dynamics and emphasise the prevalence of discrimination in limiting minority groups' housing options. For example, Bolt et al. (2010) explains that in Sweden and Belgium minorities are denied to move in ethnically dense areas or that urban redevelopment plans in the Netherlands and the United States result in a loss of cheap housing, further limiting housing options.

2.4. RESILIENT ROTTERDAM

Following the Western trend, Rotterdam also jumped on the balanced neighbourhood bandwagon. Although the municipality today has a more balanced class structure than previously, recent evidence suggests that the city may become a victim of its own success. Rotterdam is on the verge of becoming inaccessible to both the middle and lower classes (Custers, 2021).

To understand the assumptions underlying the housing policies in Rotterdam as well as their development, this chapter starts by introducing the strategy of the municipality to improve the resilience of the city. This leads to the explanation of the programs that the municipality has implemented to transform the city to a balanced city, and their definition of a balanced neighbourhood is introduced. Finally, the section closes with a reflection of the controversy of their policies that motivate this study.

2.4.1. RESILIENCE STRATEGY

The complex geographical nature of the city, which is entirely situated below sea level, has historically called for policies that fight against the adverse effects of climate (Spaans & Waterhout, 2017). To address these growing challenges, the municipality has been actively involved in working toward a climate-proof city by developing and participating in local, regional, national, and international initiatives such as the Rotterdam Climate Initiative (2007), the Rotterdam Climate Change Adaptation Strategy (2013), or the C40 Cities Climate Leadership Group (2013) (Spaans & Waterhout, 2017). In 2014, the city joined Rockefeller Foundation's 100 Resilient Cities program (100RC) and the efforts of fighting against the force of nature culminated in 2016, when Rotterdam presented its Resilience Strategy as a member of the 100RC (Municipality of Rotterdam, 2016a).

Rotterdam already has a reputation for designing and engineering robust systems, but the municipality acknowledges that future risks might call for a different response (Municipality of Rotterdam, 2016a). As such, the Resilience Strategy has not only been motivated by the response to sudden shocks (e.g. earthquakes, fires, floods, or heat waves), but also to face systemic socio-economic tensions. To address the different vulnerabilities, the Resilience Strategy defined six focus areas that represent the main resilience challenges:

- 1. Social cohesion and education,
- 2. Energy transition,
- 3. Climate adaptation,
- 4. Cyber use and security,
- 5. Critical infrastructure,
- 6. Changing urban governance (Municipality of Rotterdam, 2016a, p. 23).

To address them, the Resilience Strategy has been divided into seven goals (Figure 2.8). With the purpose of monitoring the tensions in social cohesion and improving social resilience, the Goal 1 'Rotterdam: a balanced society' was set in scope. Municipality of Rotterdam (2019b) indicates that a balanced population demographic in Rotterdam strengthens resilience at the individual and the societal level. The expansion of the resilience view from climate change adaptation to include social urban issues intends to develop community involvement and awareness to build urban resilience (G. Huang, 2020; Spaans & Waterhout, 2017).

As part of the efforts to achieve Goal 1, in 2012 the municipality developed the National Programme Rotterdam South (NPRZ) (Municipality of Rotterdam, 2016a, p. 109), and in 2016 the Woonvisie (Housing Vision) of Rotterdam (Municipality of Rotterdam, 2016a, p. 65), which aim to create a balance in the groups of residents by demolishing and renewing cheap old housing to make way for higher-income groups. Later, in 2022, the municipality updated their Resilience Strategy and adopted a new policy, the



Figure 2.8: Rotterdam's Resilience Strategy goals. Source: Molenaar (2016)

Resilient BoTu programme, which is in line with the previous policies (Municipality of Rotterdam, 2022a, p. 25).

The policy mix of Rotterdam has not only been motivated the resilience view. The social balance of Rotterdam can be traced back to 2005 when the Rotterdam Act was approved. By means of a tenant allocation reform, the Rotterdam Act can exclude newcomers who have not lived in Rotterdam for the last six years and who rely on unemployment benefits or social assistance from moving to specific city neighbourhoods (Uitermark et al., 2017; van Eijk, 2010). The objective of the Rotterdam Act is to prevent the influx of disadvantaged households to improve the liveability of the targeted neighbourhoods. Still, in 2016, an amendment to the Rotterdam Act was approved that allows towns to exclude households accused of extremism, crime, or disturbance (Uitermark et al., 2017). The Rotterdam Act has been in force in the neighbourhoods of Carnisse, Hillesluis, Oud-Charlois, Tarwewijk, Bloemhof, and a number of streets in the Delfshaven area since 2006 and no other city in the Netherlands has adopted the policy.

During the development and implementation of the housing policies, some politicians referred to ethnic mixing, while others spoke of socially mixed areas. Whatever the label, the policies grounded on the assumption that a concentration of residents relying on unemployment benefits leads to a lower quality of life in the neighbourhood, so dispersing the concentration will increase the overall well-being (Uitermark et al., 2017). This discussion was tied to a larger, contentious debate on integration (Musterd & Andersson, 2005). However, the result was the clear preference of the municipality 2

for higher socioeconomic groups over lower socioeconomic ones, usually associated to immigrants. Thus, the spark for controversy was ignited.

2.4.2. A BALANCED SOCIETY

This section elaborates on the housing policies within the Resilience Strategy that the municipality of Rotterdam has developed to achieve balanced neighbourhoods. Furthermore, it explains the definition that the municipality has adopted for the idea of balanced neighbourhood.

Beginning with the NPRZ, this initiative had started before the Resilience Strategy was put in place. Still, it was included as part of it. The aim of the NPRZ is to reach average national levels in education, participation in labour, and quality of life in Rotterdam South by 2030 (Municipality of Rotterdam, 2019b). Furthermore, the municipality explicitly states that the NPRZ is key for the strengthening of social resilience and social cohesion in Rotterdam South.

The programme focuses on the discourse on urban renewal in Rotterdam, and therefore has mainly looked at the housing and spatial aspects of this plan. They aim for a decrease of thousands of low-cost housing in exchange for more costly in the neighbourhoods of Feijenoord, Afrikaanderwijk, Hillesluis, Bloemhof, Tarwewijk, Carnisse, and Oud-Charlois in the South of Rotterdam (Municipality of Rotterdam, 2019a). This has attracted the attention of researchers, activists and international organisations such as the United Nations given the exclusion of low-income households from these neighbourhoods (UN-OHCHR, 2021).

Following with the Woonvisie, the end goal of the programme is to achieve a balance in the housing stock of Rotterdam neighbourhoods for 2030. A balance in the housing stock will result in a balanced society. Nonetheless, the initial document did not explicitly state what the municipality understands by a balanced neighbourhood. It was not until three years later that balance was defined in the Gebiedsatlas, an atlas which monitors the evolution of the housing programme in the city (Municipality of Rotterdam, 2020). Figure 2.9 shows the conditions that the housing stock needs to meet for the municipality of Rotterdam to consider the neighbourhood balanced (Municipality of Rotterdam, 2020, p. 14). Social, Middle, Higher, and Top are the names the municipality gives to the different housing price segments. This is measured by means of the WOZ-value (Valuation of Immovable Property Act value). The WOZ-value is the value of a property for tax purposes in the Netherlands. They are based on market values of the characteristics of the building, official valuations, and the selling price of nearby properties (Ministry of General Affairs, 2016). Table 2.1 shows the WOZ-values that define the housing segments in 2018. The addendum notes that the prices need to be adjusted per year according to the NHG (National Mortgage Guarantee) limit.

The definition in Figure 2.9 suggests that there is not a unique distribution of balance, but that it can be achieved in multiple ways. This is because the Woonvisie targets the whole city of Rotterdam, but has tailored strategies for the different areas of the city. The municipality states that the aim is not to have a homogenous structure, but that each neighbourhood can retain its own character and, within certain bandwidths, each neighbourhood's quality of life comes into its own (Municipality of Rotterdam, 2020). In turn, this complicates the understanding of balance and can result in ambiguous inter-

>=45% Social and Middle, =<60% Social, and >=40% Middle, Higher, and Top.

Figure 2.9: Balanced neighbourhood objective of the municipality of Rotterdam. The conditions indicate the percentage of houses that they aim to have in that price segment in each neighbourhood.

Table 2.1: WOZ-value housing segments in Rotterdam in 2018.

	Social	Middle	Higher	Тор
WOZ-value 2018 (in thousand €)	<220	220-265	265-400	>400

pretations of the concept.

To achieve this goal, the municipality of Rotterdam plans to reduce the housing stock by 13,500 homes, of which 10,900 homes in the cheapest (Social) segment are planned to be demolished. At the same time, the municipality wants to increase the number of houses in the Middle, Higher, and Top price segments by 46,600 new housing units (Municipality of Rotterdam, 2019c). The programme aims to intervene on a city-wide and neighbourhood level where necessary and in a targeted manner. The guideline for new housing throughout the city — in the short and long term — is to build the following composition: 20% Social segment, 30% Middle segment, 30% Higher segment and 20% Top segment (Municipality of Rotterdam, 2019c).²

2.4.3. CONTROVERSY OF BALANCE

As a result of the housing policy-mix, the municipality of Rotterdam has seen a decrease in the social housing stock. Consequently, low-income individuals, mostly persons of immigrant heritage, are being pushed to leave the city (UN-OHCHR, 2021). The decision to reduce cheap housing stock is based on the assumption that the amount of cheap housing is greater than the amount of households that belong to the cheap (Social) price segment. To have a reference, Figure 2.10 shows how, in 2019, the amount of Social housing predominated in the city. Nevertheless, this assumption is debated and some even estimate that by the year 2030 there will be a shortage of nine thousand social homes in Rotterdam (Staalduine, 2019). In addition, the UN alerts that the housing policies do not take into account the impact that reallocating households has in social networks, as breaking social ties increases the vulnerability of low-income families (UN-OHCHR,

² In 2019, the Resilient BoTu 2028 programme was also launched. This initiative was not included in the first Resilience Strategy, but in the updated version of 2022. Resilient BoTu 2028 targets the neighbourhoods of Bespolder and Tussendijken to raise their Wijkprofiel Social Index score to that of Rotterdam's average within 10 years (Veld Academie, 2021). This programme encompasses a wider range of proposals than the Woonvisie, and is similar to the NPRZ in scope and magnitude. In the Resilient BoTu 2028 programme, the municipality is in line with the previously implemented housing policies. They aim to upgrade the current houses and to build new constructions to 'rebalance' the neighbourhood (Municipality of Rotterdam, 2019b, p. 20). So far, they have relied on the Woonvisie, nonetheless, the official document suggests that additional efforts should be put in place for the BoTu neighbourhoods. Even so, they do not provide specific details of these additional housing policies.

2021).



Figure 2.10: Boxplot of the number of houses that belong to the Social, Middle, Higher, and Top segment in each neighbourhood in Rotterdam in 2019.

Even though the municipality claims that advocating for more balanced neighbourhoods will have a positive effect in Rotterdam, some studies suggest that the storyline does not hold given the lack of demonstrable promised improvements (Hochstenbach et al., 2015). Supporting this claim, academics and social activists attribute these assumptions to a misleading use of indicators and statistics (Schinkel & van den Berg, 2011; Uitermark et al., 2017). For example, Lubberink et al. (2018) alerts that the WOZ-value on which the level of balance is based is not a valid and reliable market value indicator because it does not include enough housing characteristics. Furthermore, the idea of balance has been posited as a necessity on the basis of common sense, without regard for the problems that the co-living of different social groups may bring (Bricocoli & Cucca, 2016).

Finally, this sentiment of forced displacement has also reached the public. Detractors of the Woonvisie have mobilised under the movement Recht op de Stad (Right to the City) that rejects the city in balance and champion a city for everyone. In their efforts, they debate with politicians and councillors about a new better plan for housing policy and also created an electoral guide to inform voters in the elections of Rotterdam in March 2022 about the housing plans of each candidate. They also defend participatory policymaking by organising events at which they expose and ask about the current and future of Rotterdam's housing. Finally, in their efforts to come forward, they organised a public demonstration in October 17th, 2021 (Right to the City, 2022).

In sum, the urban policy mix of Rotterdam has become institutionalised over time, and become part of the policy tool kit. Therefore, it is vital to study whether the assumptions on which the municipality grounds the policies are valid.

2.5. Hypothesis development

Following the discourse of the municipality of Rotterdam coupled with the results of the literature review, the chapter concludes that the theoretical conceptualisation foundation of this research can be illustrated in Figure 2.11. With this, the following hypotheses are formulated:



Indirect hypothesis: H4: NB→SC→IS

Figure 2.11: Illustration of the concepts and the relationships that are the analysis objective of this research. Social cohesion is supposed to mediate the relationship between having balanced neighbourhoods and more informal support in the city.

H1: Social cohesion positively affects Informal support. In cohesive societies, local communities protect residents against threats, care of others during hardships, and ultimately promote community resilience (Larimian et al., 2020; Municipality of Rotterdam, 2019b).

H2: Neighbourhood balance positively affects Social cohesion. In socially mixed areas, different groups meet and interact with each other, and eventually this reduces hostility toward out-groups (Allport et al., 1954). Balanced neighbourhoods mix the population composition in a neighbourhood to improve the social cohesion of its residents (Municipality of Rotterdam, 2016b, 2019b).

H3: Neighbourhood balance positively affects Informal support. Balanced neighbourhoods are considered a policy instrument that promote resilient actions by means of social mix (Allport et al., 1954; Municipality of Rotterdam, 2019b).

H4: The relationship between Neighbourhood balance and Informal support is mediated by Social cohesion. The previous hypotheses build the grounds for the mediation hypothesis between the concepts. Social cohesion acts as the mechanism for resilient actions triggered by the balance in a neighbourhood (Méreiné-Berki et al., 2021; Municipality of Rotterdam, 2016b, 2019b). The study of these hypotheses is complemented in two ways. First, by taking into account the implications of spatial distribution. Social cohesion and informal support are not constrained by administrative boundaries, so the social perceptions and actions in nearby neighbourhoods affects the level of the other neighbourhoods (Forrest & Kearns, 2001; Sampson et al., 1997). This formulates the following hypotheses:

H5a: Spatial spillover effects are significant on the relationship between Neighbourhood balance and Social cohesion.

H5b: Spatial spillover effects are significant on the relationship between Neighbourhood balance and Informal support.

Second, the study of the relationships is complemented by considering the moderating effect of demographic and built environment elements identified in the literature review. Moderating effects overcome the supposition that the relationships between neighbourhood balance, social cohesion, and informal support are exactly the same for all neighbourhoods and identify instances in which the moderating variables change the strength or direction of the relationship between the concepts (Henseler, 2020). This formulates the following hypotheses:

H6a: Ethnic heterogeneity moderates the relationship between Neighbourhood balance and Social cohesion.

H6b: Ethnic heterogeneity moderates the relationship between Neighbourhood balance and Informal support.

H7a: Residence length moderates the relationship between Neighbourhood balance and Social cohesion.

H7b: Residence length moderates the relationship between Neighbourhood balance and Informal support.

H8a: Population density moderates the relationship between Neighbourhood balance and Social cohesion.

H8b: Population density moderates the relationship between Neighbourhood balance and Informal support.

H9a: House type heterogeneity moderates the relationship between Neighbourhood balance and Social cohesion.

H9b: House type heterogeneity moderates the relationship between Neighbourhood balance and Informal support.

H10a: The amount of greenery moderates the relationship between Neighbourhood balance and Social cohesion.

H10b: The amount of greenery moderates the relationship between Neighbourhood balance and Informal support.

H11a: Walkability moderates the relationship between Neighbourhood balance and Social cohesion.

H11b: Walkability moderates the relationship between Neighbourhood balance and Informal support.

H12a: The number of amenities moderates the relationship between Neighbourhood balance and Social cohesion.

H12b: The number of amenities moderates the relationship between Neighbourhood balance and Informal support.

2.6. CONCLUSION

This chapter reviewed previous literature to get a deeper understanding and extract the necessary concepts, to operationalise balanced neighbourhoods, social cohesion, resilient action, and the relationship between them. Concluding with the research hypotheses, this chapter answers SQ1: *What are the underlying mechanisms that drive balanced neighbourhoods to act more resiliently*?.

Literature suggests that social factors govern the resilience of urban areas. Communities in which the residents work together and have common goals have a stronger willingness to cooperate. Given that resilience is not directly measurable, we argue that informal support is a suitable proxy to measure the potential resilience of a neighbourhood, as the end mechanism through which social resilience is achieved is mutual aid.

Social cohesion is an involuntary product of the form and quality of shared values, trust, and the relationships among the individuals of a society. As a result, in cohesive societies, local communities protect residents against threats, care of others during hard-ships, and ultimately promote community resilience. The municipality of Rotterdam also maintains this claim, and regards social cohesion as a challenge to address to reach the goal of a resilient city. We then consider social cohesion as a multidimensional concept that is a geographically bounded property that characterises neighbourhoods.

Finally, balanced neighbourhoods are considered a policy instrument to achieve a more resilient city by means of social mix. Balanced neighbourhoods mix the population composition in a neighbourhood to allegedly improve the social cohesion of its residents. The effectiveness of this strategy has been very contested in literature, as empirical studies have mostly conflicting results. Furthermore, the exclusionary nature of such policies increase the controversy of their implementation. We have seen the strategies that the municipality of Rotterdam has taken so far to promote balanced neighbourhoods, and conclude that it is imperative to study whether the assumptions on which the municipality grounds the policies are legitimate.

3

METHODOLOGY

This chapter has three main objectives. First, to identify the data that can transfer the conceptual model to a set of measurable indicators. Second, the chapter then introduces the measurement model which will be used for the study. Finally, it identifies and explains the research methods that can satisfy the main requirements for the analysis of the proposed model. Putting all together, a computational framework that will be applied to the case study is presented.

3.1. DATA

This thesis strongly depends on the availability of open data due to the public nature of the research. The operationalisation of individually measured characteristics, such as social cohesion or informal support, is best captured by the use of microdata. Microdata are administrative data on the individual level that are derived from various registers such as income, age, sex, or, very important for this study, spatial data. Such an infrastructure with detailed and extensive microdata is present in few countries, but the Netherlands is superb in the collection and accessibility to microdata (Custers, 2021). Due to the nature of the research at a *buurt* level (the neighbourhood level from here on), the aggregate data at this level will be used. Microdata provides the capabilities of identifying the neighbourhood to which the respondents belong. In addition, due to data privacy concerns, other data such as the WOZ-value is not available at a micro level, so we are forced to use aggregated data sets.

In this section, the data used in the research is presented together with its characteristics and sources. The **data from the year 2019 was selected** for the cross-sectional analysis, as it has the most recent survey data before the impact of COVID-19. We argue that the pandemic will have an impact on the social cohesion and willingness to help of residents, and therefore analyse the situation before it to avoid possible biases. First, the indicators for social cohesion and informal support are shown as collected by a survey conducted in Rotterdam. Second, the collection and aggregation of WOZ-values for the houses in each neighbourhood is explained. Then, the data collection and pre-

Cons	Constructs and indicators			Mean	s.d.
Socia	l cohesion				
SC1	% of residents who say that local residents know each other	70	2/72	37	16
SC2	% of residents who say that local residents spend a lot of time with each other	70	2/60	29	12
SC3	% of residents who say that local residents share each other's views	70	16/64	32	11
SC4	% of residents who say that local residents help each other	70	25/56	41	6
SC5	% of residents who say they feel at home with local residents	70	28/81	55	11
Infor	mal support				
IS1	% of residents who say they are willing to care for neighbors or friends who need help	70	45/71	57	6

Table 3.1: Characteristics of the indicators for *Social cohesion* and *Informal support* in the Wijkprofiel in 2019. N indicates the number of neighbourhoods included in the Wijkprofiel. Source: OBI.

processing for the moderating factors are presented. Finally, the section explains how the data had to be adapted to the survey collection areas.

3.1.1. SOCIAL COHESION AND INFORMAL SUPPORT

The **Wijkprofiel** (Neighbourhood profile) provides the data to assess the operationalisation of social cohesion and informal support. The profile was developed by the Research and Business Intelligence (OBI) department in Rotterdam over two decades ago and serves as an instrument to monitor social development and inform local policies (Municipality of Rotterdam, 2022b).

The Wijkprofiel is based on registrations and survey data collected biannually to determine the scores for three domains: physical, safety, and social. A stratified sample approach is utilised since the Wijkprofiel seeks to be representative at a neighbourhood level. As a result, random sampling is done at the neighbourhood scale. The sample system is based on a municipal address list from which possible respondents are selected at random. The response rates vary between 20% and 25% per wave (Custers, 2021). Due to the nature of the research at a neighbourhood level, the **aggregate data** at this level will be used. Appendix B.1 provides information of the aggregation procedure used by the municipality of Rotterdam, as well as information of the survey questions.

To identify the set of indicators which fit the conceptual model, this study follows the Wijkprofiel survey design. The survey is composed of sets of indicators that correspond to established psychometric scales. To begin with, the Wijkprofiel uses the same indicators as the Dutch Office of Social and Cultural Planning for the conceptualisation of *Social cohesion* (Schnabel et al., 2008). This conceptualisation is based on synthesis theory and its indicators can be seen in the first panel of Table 3.1. These elements have previously been validated as the pieces that forge social cohesion.

Second, the survey also contains an independent section for care giving. The second panel of Table 3.1 shows the indicator from the Wijkprofiel to quantify the willingness of citizens to provide care to friends and neighbours who need help. We argue that this indicator is a good representation of informal support in Rotterdam and can act as a proxy for potential resilient actions. As we have seen in Chapter 2, the end mechanism through which social resilience is achieved is mutual aid, therefore, the willingness of citizens to help can capture the potential resilience of a neighbourhood (Doff, 2017). The indicator captures the willingness to help others in the strong (friends) and weak (neighbours) relationship ties.

35

Table 3.2: WOZ-value segments from 2018 and 2019. "Real" shows the value used by the municipality of Rotterdam. "Approximated" shows the segments that can be used from the OBI data. The values in bold for 2018 "Real" are the official values, the 2019 values have been adjusted following the % increase NHG limit, rounded to multiples of five thousand. Source: Municipality of Rotterdam (2019c)

		WOZ-value Social	WOZ-value Middle	WOZ-value Higher	WOZ-value Top
		(in thousand €)	(in thousand €)	(in thousand €)	(in thousand €)
2010	Real	<220	220-265	265-400	>400
2018	Approximated	<225	225-275	275-400	>400
2010	Real	<240	240-290	290-440	>440
2019	Approximated	<250	250-300	300-450	>450

Table 3.3: Characteristics of the WOZ-value segments used to measure the level of balance for the year 2019. N indicates the number of neighbourhoods included in the Wijkprofiel. Source: BAG.

WOZ-value segments	Ν	Min/Max	Mean	s.d.
<250k (Social)	70	80/13,113	3,498	2,639
250-300k (Middle)	70	0/1,545	265	289
300-450k (Higher)	70	0/2,066	306	395
>450k (Top)	70	0/1,757	175	336

3.1.2. BALANCED NEIGHBOURHOODS

To measure the level of balance of neighbourhoods, the **distribution of WOZ-values** per neighbourhood was used. The Addresses and Buildings Key Registry (BAG) is an automated system that stores information on local addresses and buildings. They collect the WOZ-values of every house in The Netherlands and make them available to governments, companies, institutions, and private citizens (Netherlands Enterprise Agency, 2022). In the case of Rotterdam, this data is stored by the OBI and made publicly accessible in their data portal (Research and Business Intelligence, 2022).

The Woonvisie addendum points out that the Social, Middle, Higher, and Top segments are **adjusted** following the percentage increase National Mortgage Guarantee (NHG) limit, rounded to multiples of five thousand (Municipality of Rotterdam, 2019c, p. 2). In the Netherlands, the mortgage is specified by the NHG to guarantee that the risk and the interest rate are lower for buyers. The limit for a mortgage backed by the NHG is adjusted every year based on the average price of a home in the Netherlands (ABN-AMRO, 2022). Adjusting the brackets in Table 2.1 to this limit allows defining the balance to the yearly market situation. In Municipality of Rotterdam (2019c, p. 10), the municipality only includes the value for the 2018 balance, so segments for the other years had to be manually adjusted. In addition, the WOZ-values data is presented in the data portal as the count of houses that fall within a set of predefined price brackets per neighbourhood. Therefore, to fit the definition of the brackets from the municipality of Rotterdam, the WOZ-values were aggregated to the **best approximation** as can be seen in Table 3.2. Finally, Table 3.3 shows the characteristics of the data used to measure the level of balance for the year 2019.

3.1.3. MODERATING FACTORS

To determine the moderating effect of exogenous factors, Table 3.4 shows the selection and the characteristics of items that the literature review has identified as relevant.

To begin with, *Ethnic heterogeneity* has been measured as one minus the Herfindahl-Hirschman Index (HHI), a very commonly used method to calculate the degree of ethnic diversity per neighbourhood (Chakraborty & McMillan, 2018).¹ This index was calculated using data collected by the Dutch Personal Records Database (BRP), which includes each share of nine ethnic groups per neighbourhood: Dutch, Turks, Moroccans, Surinamese, Cape Verdeans, Antilleans, other EU, other Western, and others. A value of 0 means all residents are from the same ethnicity. Values that approach 1 show that residents at that neighbourhood are from multiple ethnicities. Another factor modelled using one minus the HHI is *House type heterogeneity*. This calculates the degree of house type diversity per neighbourhood. This index was also calculated using data collected by the BAG, which includes each share of four house type groups per neighbourhood: single-family house, multi-family house with elevator, multi-family house without elevator, and other multi-family house.

The other demographic factor considered was *Residence length*. The data was provided by the Wijkprofiel survey conducted by the OBI and includes the percentage of citizens in a neighbourhood that have resided in that same neighbourhood for ten years or more.

The land use has been regarded as another relevant factor. First, *Population density* was obtained from the data collected by the BRP and is given in citizens per neighbourhood hectare. In the same context of land use, *Amenities* represents the total number of cafés, restaurants, community centres, bars, and pubs in each neighbourhood. Data was obtained from OpenStreetMap (OSM) using Overpass (OpenStreetMap contributors, 2017).

Finally, the built environment has also been considered in previous studies as a relevant factor (Mazumdar et al., 2018; Mouratidis & Poortinga, 2020). First, *Greenery* represents the percentage of green area in a neighbourhood which serves that specific function. For example, a green lane split, or a green riverbank do not count as green area, but as road infrastructure or water, respectively. This value was obtained from the 'Functional urban land use at ground level' data set from the OBI. Second, *Walkability* represents the percentage of walkable area in a neighbourhood which serves that specific function. This value was obtained from the same data set as *Greenery*.

3.1.4. NEIGHBOURHOOD SHAPE

Finally, the Wijkprofiel does not contain information on all neighbourhoods, plus it does not strictly follow the official neighbourhood division of Rotterdam. Consequently, the analysis, especially in the spatial context, has to be adapted accordingly.

Heterogeneity = 1 – HHI = 1 –
$$\left(\sum_{i=1}^{k} \pi_i^2\right)$$
.

This value varies between 0 and 1. A value of 0 means no diversity at all in the neighbourhood. Values that approach 1 show a high degree of diversity.

¹Consider k groups with each having a share of the total composition, π , then the heterogeneity was calculated as:

Exogenous factors	Description	Ν	Min/Max	Mean	s.d.
Ethnic heterogeneity	One minus the Herfindahl-Hirschman Index of the ethnic groups	70	0.2/0.86	0.67	0.16
Residence length	% of residents who have lived for a long time in the neighbourhood	70	0.06/0.62	0.42	0.10
Population density	Number of residents per hectare	70	2.86/209.47	78.73	57.10
House type heterogeneity	One minus the Herfindahl-Hirschman Index of the house type groups	70	0/0.74	0.56	0.16
Greenery	% of green space	70	2.8/76.5	20.33	13.03
Walkability	% of walkable infrastructure	70	0.6/31.6	13.98	6.43
Amenities	Number of amenities	70	0/76	6.73	11.86

Table 3.4: Characteristics of the exogenous factors.

Table 3.5: Differences in the Wijkprofiel survey structure compared to the official CBS neighbourhood division. "Not included" column shows all those neighbourhoods that are not in the Wijkprofiel survey. "Combined" column shows the neighbourhoods that were combined into one same collection area for the Wijkprofiel survey. "Divided" shows the neighbourhoods that were split into two different survey data collection areas.

Not included	Combined	Divided
Landzicht	Oud Mathenesse + Witte Dorp	Groot IJsselmonde
Spaanse Polder	Blijdorp + Blijdorpsepolder	
Nieuw Mathenesse	Kralingen Oost + Kralingse Bos	
Waalhaven	Zuiderpark + Zuidrand	
Eemhaven	Dorp + Rijnpoort	
Waalhaven Zuid	Noord Kethel + Schieveen + Zestienhoven	
Vondelingenplaat		
Botlek		
Europoort		
Maasvlakte		
Bedrijvenpark Noord West		
Rivium		
Bedrijventerrein Schieveen		
Noordzeeweg		

Table 3.5 summarises the differences between the Wijkprofiel survey structure and the official CBS neighbourhood division. To begin with, a list of neighbourhoods was not included in the survey. Most of these neighbourhoods belong to the port area or to neighbourhoods with few residents. Second, the survey groups certain neighbourhoods into larger areas to reduce the reliability margin. Finally, in the survey, the neighbourhood of Groot Ijsselmonde was divided into North and South Groot Ijsselmonde due to its large size.

For this study, the Rotterdam shapefiles and the data associated to them had to be adapted to take into consideration these changes in the administrative boundaries for the survey. First, the neighbourhoods **not included** in the survey were also not included in the subsequent research. Second, the **combined** neighbourhoods in the survey had their shapefiles combined as well as the data values associated to them. In this aggregation process, values that do not belong to the Wijkprofiel survey were aggregated by means of a weighted average. For example, *Residence length* used as weights the number of citizens in each neighbourhood. Finally, the neighbourhood of Groot Ijsselmonde was **divided** into North and South. Given that the housing data is collected at the official administrative boundaries' division, the Wijkprofiel values for Groot Ijsselmonde North and South had to be aggregated by means of a weighted average using as weights the number of Wijkprofiel respondents in each of them.

3.2. MEASUREMENT MODEL

The combination of the available data and the theoretical foundations conceptualise the measurement model in Figure 3.1, which shows the indicators that make up the constructs and the relationships that this study aims to model and analyse. The constructs of the model are defined as follows²:

- *Neighbourhood balance*: How far a neighbourhood is from achieving balance as defined by a target house WOZ-value distribution. It is modelled as a single-indicator emergent construct, i.e. using synthesis theory, as there is no systematic error associated to the measurement.
- *Social cohesion*³: 'The extent to which people express in behaviour and perception their involvement in social connections in their personal lives, as citizens in society and as members of society.' (Schnabel et al., 2008, p. 13). It is modelled as an emergent construct, i.e. using synthesis theory, as it is not reducible to the several dimensions expressed in behaviour, perception, and involvement of individuals as members of a social entity.
- *Informal support*: 'Help you give to your neighbours or friends, if that person is sick, dependent or disabled for a long time. It is not paid, a volunteer from a volunteer centre is not an informal caregiver, it is not about professional care, and it is not about the normal care of parents to their children.' (Municipality of Rotterdam, 2022b, p. 12)⁴. It is modelled as a latent construct, i.e. using measurement theory, as it is a unique behavioural aspect of individuals.

In Figure 3.1 the indicators that make up the measurement model of each construct are shown. Unlike the dichotomous definition of the municipality of Rotterdam (see Figure 2.9), we are interested in quantifying the possible outcomes that can result from the feasible ranges of the definition of balanced neighbourhood. Several techniques have been developed to measure diversity and segregation, but in the following section we argue why the **Kullback-Leibler (KL) divergence** is the most suitable measurement for *Neighbourhood balance* in this study (Kullback, 1987).

3.3. RESEARCH METHODS

Having the measurement model, this section provides motivation for the chosen research methods and, finally, explains the necessary fundamentals for each of them.

Five major aspects were taken into consideration: (i) the constructs are composed of multiple indicators, so the method had to be able to aggregate them into one construct.

²From this point all constructs in the model will be in italics for disambiguation.

³This definition of *Social cohesion* is adopted given that it is an emergent construct, given the indicators with which it is modelled. This research also tried to model *Social cohesion* as a latent variable following the other definition of the Dutch government (CBS, 2015) by means of Exploratory Factor Analysis (EFA). The results indicate that no clear factors could be identified, so we conclude that from the data available, *Social cohesion* is best modelled formatively. Appendix B.2 shows the methodology and the obtained results.

⁴The presented model conceptualises *Informal support* as willingness to help friends or neighbours. Appendix C shows the results of adding indicators of the willingness to help relatives, and of the willingness to help others in the surroundings. The discussion section elaborates on the implications of the findings using the different indicators.



Negative of the Kullback-Leibler divergence of the WOZ-value distribution in a neighbourhood
with respect to the objective balance distribution
% of residents who say that local residents know each other
% of residents who say that local residents spend a lot of time with each other
% of residents who say that local residents share each other's views
% of residents who say that local residents help each other
% of residents who say they feel at home with local residents
% of residents who say they are willing to care for neighbours or friends who need help

Figure 3.1: Structural and measurement model. Incoming arrows to a construct mean that is modelled formatively, so it is an emergent construct. When arrows point away from the construct, it means that it is modelled reflectively, so it is a latent construct. Furthermore, *Social cohesion* is an emergent variable, while *Informal support* is a latent variable, which added another layer of complexity in the identification of a methodology that could estimate both constructs. Next, it is necessary to study three types of associations: (ii) direct effects between the constructs, (iii) the mediating effect of *Social cohesion* as a way to increase *Informal support* by means of *Neighbourhood balance*, and (iv) the moderating effect of exogenous factors on the strength of the relationships between the constructs. Consideration (v) and finally, the dependence of the results on the spatial distribution of the neighbourhoods in the city had to be taken into account.

To fulfil these requisites, **Partial Least Squares Structural Equation Modelling** is a modelling technique that enables us to simultaneously examine considerations (i), (ii), (iii), and (iv). On one hand, it allows estimating the measurement of emergent and latent constructs that arise from other observable indicators. On the other hand, it can estimate the direct, indirect and total effects between *Neighbourhood balance, Social cohesion*, and *Informal support* as well as the moderating effect of exogenous variables (Hair et al., 2016; Henseler, 2020; Lauro et al., 2018).

Finally, given that the theory of the model is based on the grounds that geographically connected people become affected by the balance in their neighbourhood to promote informal support, **spatial econometric** techniques will be used to address the spatial interaction and effects among observations in the model in consideration (v). Within the many available techniques, we will make use of **Local Indicators of Spatial Association** and the **Durbin spatial model** (Anselin, 1988, 1995). Using spatial econometrics techniques to consider spatial autocorrelation is a prime concern, as previous research by the municipality of Rotterdam has obviated it in their analyses (Uitermark et al., 2017).

Together, the combination of these methods provide a strong analysis framework. As a result, better assumptions regarding neighbourhood effects may be developed and subsequently evaluated in different neighbourhood contexts.

3.3.1. KULLBACK-LEIBLER DIVERGENCE

The measurement of inequality, segregation, and diversity is critical for the understanding of complex social phenomena. The measurements are relevant to policy-makers as they provide insights into how to plan urban areas. As a result, several measures have been developed. This section will start by exploring different indicators and explaining why the Kullback-Leibler (KL) divergence is the most suitable measurement for the balance of neighbourhoods in this study.

To begin with, the most popular measure of residential segregation is the Dissimilarity index (Duncan & Duncan, 1955). The index score indicates the proportion of one of two groups used in the calculation that would have to relocate to another subarea to match the distribution of the wider area. The success of this index lies in its simplicity; however, it is limited to the segregation of two groups. Another main drawback of the index is that it is not additively decomposable, so total segregation cannot be decomposed into segregation occurring within and between areas.

Another popular segregation measure is the Simpson Diversity index. It overcomes the limitation of the Dissimilarity index, as it is used to measure the degree of concentration when elements are classified into several types. The measure equals the probability that two entities taken at random from the dataset of interest represent the same type (Simpson, 1949). The Simpson Diversity index has been widely used in several fields and has been coined with different names in the different fields: Herfindahl–Hirschman index in economics, Hunter–Gaston index in microbiology, or Housing Diversity Index in real estate (Chakraborty & McMillan, 2018). Nevertheless, as with the Dissimilarity index, it is not additively decomposable. Furthermore, in the research context, while the Simpson Diversity index (or the Housing Diversity index, to be more precise) serves to capture a mix of the existing housing stock, it does not allow capturing the mix relative to a target housing stock.

To address the issue of additive decomposition, a popular measure of inequality and segregation is the Theil index. It measures the probability of an outcome occurring weighted by its probability of occurrence. It is based on the concept of entropy, where entropy is a measure of randomness in a given set of information (Roberto, 2015). The World Bank has defined Theil's entropy as the 'best-known entropy measure ...to decompose inequality into the part that is due to inequality within areas (e.g. urban, rural) and the part that is due to differences between areas (e.g., the rural-urban income gap).' (Haughton & Khandker, 2009, p. 101). However, some inconveniences are present in the use of Theil's index. First, the entropy can only be calculated for discrete distributions, so the inequality of continuous distributions like income cannot be measured. Second, and of importance to this research, is the limitation that it measures the difference between a set of distributions and a summary statistic of those distributions, e.g., the mean. Therefore, it is limited to measure whether a neighbourhood is balanced according to the definition by the municipality of Rotterdam.

Theil also developed the Information Theory Index, which is another entropy-based measure. The index measures the extent to which the entropy of a subarea is below or above the entropy of the entire area. Even if the measurement has become the standard for segregation studies, the Information Theory Index measures relative homogeneity, and it is misleading to interpret it as a measure of diversity (Roberto, 2015). In addition, it has the same limitations as the Theil index.

The **Kullback-Leibler divergence** offers the opportunity to overcome the mentioned issue of the previous indices to develop a measurement of the level of balance of a neighbourhood. Unlike Theil's indices, which are entropy-based, the KL divergence is based on relative entropy, a measure of the difference between two probability distributions (Kullback, 1987). Formally, given an empirical probability distribution, *P*, and a theoretical reference probability distribution, *Q*, defined in the same probability space, \mathcal{X} , the KL divergence from *Q* to *P* is defined as

$$D_{\mathrm{KL}}(P \parallel Q) = \sum_{x \in \mathscr{X}} P(x) \log\left(\frac{P(x)}{Q(x)}\right)$$
(3.1)

Similar to the entropy measures, the KL divergence measures the amount of information lost when Q is used to approximate P. Consequently, the minimum value is 0, indicating that there is no difference between P and Q, and the maximum value can be greater than 1. The logarithms in Equation (3.1) are taken to base e and information is measured in nats (Roberto, 2015). One nat is the information content of an event when the probability of that event occurring is 1/e. This means that large values of the KL divergence suggest a larger deviation from the balance.



x	Distribution <i>P</i> (<i>x</i>)	Distribution Q(x)
0	9/25	1/3
1	12/25	1/3
2	4/25	1/3

Figure 3.2: Example of the Kullback-Leibler divergence. Top: the graphical representation of the real data distribution, *P*, and the objective distribution, *Q*. Bottom: Values of the distributions. Source: Kullback (1959) as cited on Wikipedia.

Kullback gives an example in Kullback (1959). Let P be the real data distribution and Q the objective uniform distribution (see Figure 3.2). Then, the KL divergence from Q to P is calculated as:

$$D_{\text{KL}}(P \parallel Q) = \sum_{x \in \mathscr{X}} P(x) \ln\left(\frac{P(x)}{Q(x)}\right)$$

= $\frac{9}{25} \ln\left(\frac{9/25}{1/3}\right) + \frac{12}{25} \ln\left(\frac{12/25}{1/3}\right) + \frac{4}{25} \ln\left(\frac{4/25}{1/3}\right)$
 $\approx 0.0852996 \text{ nats}$

For interpretability, we define the measurement of *Neighbourhood balance*, *NB*, as the negative KL divergence between the WOZ-value distribution of a neighbourhood, *P*, and an assumed distribution of balance for that year (from Figure 2.9), *Q*,

$$NB = -D_{\mathrm{KL}}(P \parallel Q). \tag{3.2}$$

The more negative the *NB*, the less likely it is that the WOZ-value distribution of a neighbourhood is the assumed distribution, thus the further way to achieve balance.

As the conditions in Figure 2.9 show, there are multiple combinations of Social, Middle, Higher, and Top house brackets that are considered as balanced. By applying the KL divergence to the set of combinations, the technique allows us to explore the implications of the definition by the municipality of Rotterdam. If the different possibilities of balance are very dispersed, the KL divergence will capture the increase in divergence from one another.

3.3.2. K-MEANS CLUSTERING

The KL divergence allows us to find the level of *Neighbourhood balance* for any objective housing stock value distribution. To find relevant insights of the results that different distributions may yield, it is relevant that we select instances which are representative of the possible situations. To this end, clustering allows us to group the possible distributions that define balance into a number of representative sets.

Clustering is an unsupervised machine learning technique and there are several existing clustering algorithms. In this study, we will use the k-means clustering algorithm because it is simpler, faster and has fewer parameters to set than other algorithms like DBSCAN or Expectation–Maximization (Pedregosa et al., 2011). In k-means, an initial number, k, of clusters is specified, and then, the algorithm places k centroids at random. Then, it calculates the euclidean distance from each point in the dataset to the centroids. With this, it assigns each data point to the closest centroid using the distance in the previous step. The new centroids are calculated by taking the averages of the distances in each cluster and the algorithm is rerun, until the centroids do not change or for a specified number of iterations (Pedregosa et al., 2011).

3.3.3. PARTIAL LEAST SQUARES STRUCTURAL EQUATION MODELLING

Structural equation modelling (SEM) has been labelled as one of the most useful advanced statistical analysis techniques for the social sciences in the recent decades (Hair et al., 2016). SEM is a multivariate technique enables the simultaneous examination of observed variables and constructs, as well as between constructs (Hair et al., 2016). Observed variables are those for which we have concrete values for the individual observations (we will refer to them as indicators). In contrast, the constructs do not have a concrete measure and have to be estimated from the indicators. In addition, SEM is capable of performing **mediation analysis**. As we concluded from the literature review, the mechanism studied in the present thesis is of *Social cohesion* mediating the relationship between *Neighbourhood balance* and *Informal support*. Mediation analysis explains how the change between an independent variable and a dependent variable occurs by means of the mediating variable (Henseler, 2020). Whereas the study of direct effects gives an answer on whether a change in an independent variable evokes a change in a dependent variable, mediation analysis tells how this change occurs.

The partial least squares approach to SEM (PLS-SEM) consists of an iterative algorithm that estimates the constructs measured by a set of indicators and the relationships between them, by means of an interdependent system of equations based on multiple and simple regression (Lauro et al., 2018). We can differentiate between the inner and the outer models. The **outer model** consists of the equations that make up the different constructs from their corresponding indicators, while the **inner model** captures the effects between the constructs themselves. Within the inner model, we can also distinguish between exogenous and endogenous variables. **Exogenous variables** are those that do not depend on any other variable within the inner model, whereas **endogenous variables** have at least one other construct causing them (Henseler, 2020).

The derivations and equations below are cited and summarized from Lauro et al. (2018). In the estimation of the constructs, two relevant types can be identified: latent variables and emergent variables. Latent variables are estimated by what is known as

Mode A (reflective). Consider *k* constructs, ξ_j with *j* from 1 to *k*, which are latent variables of *q* indicators, x_{ij} with *i* from 1 to *q*. Then, Mode A related the indicators to the latent variables by

$$x_{ij} = \lambda_{ij}\xi_j + \epsilon_{ij} \tag{3.3}$$

where λ is the **loading** of the indicator on the latent variable (i.e. the bivariate correlation between the indicator and its construct) and ϵ is the error term associated to each indicator. On the other hand, emergent variables are estimated using what is known as Mode B (formative), which relates the indicators to the emergent constructs by

$$\xi_j = \sum_j \pi_{ij} x_{ij} + \delta_j \tag{3.4}$$

where π is the **weight** of the indicator on the emergent variable (i.e. the regression between the indicators and the construct) and δ is the error term associated to each emergent variable. The distinction between Mode A and Mode B is the different mathematical specification between loadings and weights. While the latter takes into account the collinearity between indicators, the former ignores collinearity (Rigdon, 2012). Conceptually, calculating the correlation in Mode A is equivalent to identifying the common factor between the indicators while in Mode B, the regression corresponds to forging the construct by linearly combining the indicators.

Finally, the constructs are related to each other in the inner model. Here, η plays the role of a construct when it is an endogenous variable and ξ when it is an exogenous variable. This relationship is estimated by

$$\eta_j = \sum_{j'} \beta_{jj'} \eta_{j'} + \sum_h \gamma_{jh} \xi_h + \zeta_j.$$
(3.5)

where η_j is an endogenous dependent construct (*Informal support* in our case), and η'_j is an endogenous construct that is both a dependent and an independent variable (*Social cohesion* in our case). β and γ are known as the path coefficients. Path coefficients are the partial correlation coefficients, adjusted for other independent variables, that express the strength of the dependence relationships between constructs. ζ represents the error term (Lauro et al., 2018).

Besides Mode B, emergent variables can also be constructed by what is known as Mode BNNLS which uses an algorithm known as best fitting proper indices (BFPI) to estimate the constructs. In short, BFPI restricts the signs of the weights of each observable variable to guarantee that it contributes to its own construct in a predefined way. For more information on the underlying mechanism, see Dijkstra and Henseler (2011).

Covariance-based structural equation modelling (CB-SEM) is an alternative approach to SEM estimation. The difference between PLS and CB approaches will not be discussed here. For a more detailed overview of the mechanisms and the differences between PLS-SEM and CB-SEM readers are referred to Hair et al. (2016), Henseler (2020), and Lauro et al. (2018). The underlying mechanisms of both approaches lead to different motivations to use them. PLS-SEM is considered the best approach for this research, given that:

• the research objective is to understand implications of established theories;



Figure 3.3: Exogenous factors as moderators of the effect of *Neighbourhood balance* on *Social cohesion* and on *Informal support*.

- the path model includes formatively measured constructs;
- the path model includes single-indicator constructs;
- the research is based on secondary data;
- the research requires construct scores for follow-up analyses (Hair et al., 2019).

In addition, PLS-SEM has several benefits compared to its counterpart CB-SEM:

- obtains the best prediction of the constructs;
- can estimate the model in the presence of multicollinearity;
- can estimate the model with fewer observations;
- does not require assumptions in the data distribution such as univariate or multivariate normality (Lauro et al., 2018).

The research objective combined with the benefits of PLS-SEM makes it the chosen research approach.

Along with the estimation of the constructs and the relationships between them, PLS-SEM can also estimate more complex relationships such as interaction effects. Interaction effects are brought by the so-called **moderators**. The study of interaction effects, **moderation analysis**, shows the dependency of the relationship between the constructs on the strength of the moderators, indicating instances in which the moderating variables change the strength or direction of the relationship between the constructs in the model (Henseler, 2020) (see Figure 3.3). Put simply and in words of Molin (2018), eating nuts is good; drinking beer is great; but drinking beer while eating nuts is even greater!

The interaction effect of exogenous moderators is of special interest, as they help improve the capability of the model to capture underlying mechanisms between the associations. Including exogenous moderators transforms Equation (3.5) into

$$\eta_j = \sum_{j'} \beta_{jj'} \eta_{j'} + \sum_h \gamma_{jh} \xi_h + \sum_k \gamma_{jk} \mu_k + \sum_h \sum_k \gamma_{jhk} \xi_h \mu_k + \zeta_j.$$
(3.6)

where γ_{jk} is the simple effect of the exogenous moderator μ_k , and γ_{jhk} is the interaction path coefficient of the moderation of μ_k on the relationship between ξ_h and η_j . The interpretation of the path coefficient of the model construct, γ_{jh} , also changes to a simple effect. In the interaction model, these effects now quantify the increase in the dependent variable if the independent variable is increased by one unit and the moderator is kept at zero (Henseler, 2020).

3.3.4. SPATIAL ECONOMETRICS

The interest in spatial models has been growing in recent years. The spatial dependence between phenomena in different areas is key to analyse models that consider multiple regions. Furthermore, the concern of understanding the role of heterogeneity across spatial locations motivates the need for spatially explicit techniques. To tackle this, spatial econometrics is a statistical approach developed to consider interaction and effects among observations in models with a known spatial structure (Ewing & Park, 2020).

Overman (2009) points out the main reasons to consider spatial effects. First, the assumption of non-spatial dependence invalidates the results of other statistical techniques. To unravel the true nature of relationships, spatial effects need to be considered. Modelling spatial effects improves the prediction of models, even when the underlying effect is not well understood. Second, space is a source of information. The interpretation of results is improved by exploring the spatial distribution and dependence.

When it comes to neighbourhood level policies, spatial effects may show up in a variety of ways. Effects are not restricted to act within the borders of administrative boundaries. Rather, the behaviour and characteristics of contiguous areas may be expected to spill over between neighbourhood boundaries. To analyse these effects, in this subsection, we first introduce Local Indicators of Spatial Association as a technique to identify sources of spatial autocorrelation by studying the clustering of neighbourhoods. Then, we describe spatial regression as a way to account for the effects that the spatial structure can have in the relationship between neighbourhood characteristics.

LOCAL INDICATORS OF SPATIAL ASSOCIATION

Exploratory spatial data analysis (ESDA) is the process by which data and research results are analysed taking into account their spatial structure to identify spatial outliers, and discover spatial patterns, clusters or hot spots (Dall'erba, 2009). Within this context, local measures of spatial autocorrelation focus on the relationships between each observation and the observations in their surroundings to obtain insight into the spatial structure of our data.

Local Indicators of Spatial Association (LISA) is part of the set of techniques that belong to the field of ESDA. LISA **classifies the observations** into four groups: high values surrounded by high values (HH), low values surrounded by low values (LL), high values surrounded by low values (HL), and vice-versa (LH). The key principle is to find events in which the value of an observation and the average of its neighbours are either more



Figure 3.4: Example of polygon neighbourhoods. (a) Moore neighbourhood. Polygons are neighbours if they share at least an edge or a vertex. (b) Von Neumann neighbourhood. Polygons are neighbours only if they share at least an edge. Source: Baetens and De Baets (2012).

similar (HH, LL) or dissimilar (HL, LH) than we would expect from pure chance (Arribas-Bel, 2015). Based on the paper from Anselin (1995), consider a neighbourhood *i* with an observation value *x*, then the Local Moran statistic, *I*, of that neighbourhood can be calculated as:

$$I_{i} = \frac{x_{i}}{m_{2}} \sum_{j} w_{ij} x_{j} \quad | \quad m_{2} = \frac{\sum_{i} x_{i}^{2}}{N}$$
(3.7)

where w_{ij} is the spatial weights matrix that represents the spatial structure of the data, and *N* the total number of observations. Once the Local Moran, *I*, values have been determined, their p-values are estimated to test for local spatial clusters in the absence of global spatial autocorrelation. For that, we assume that the underlying distribution for generating the observations is normal. In other words, we test the hypothesis that no local spatial association is present. The significance is then combined with the location of each observation to classify the significant locations as spatial clusters of HH and LL, and spatial outliers of HL and LH. This classification is relative to the mean of the variable (Leung et al., 2003).

For simplicity, in this research w_{ij} is equivalent to a binary matrix with ones in position ij whenever observation i is in the Moore neighbourhood of observation j, and zero otherwise (see Figure 3.4):

$$w_{ij} = \begin{cases} 1 & \text{if } j \text{ is a Moore neighbour of } i \\ 0 & \text{otherwise} \end{cases}$$
(3.8)

LISA is a useful technique that can easily show areas in which values are clustered and provide suggestive evidence about the processes that might be at work (Arribas-Bel, 2015).

SPATIAL REGRESSION—DURBIN SPATIAL MODEL

Spatial regression models allow modelling variables in the presence of spatial autocorrelation. Spatial autoregressive models extend classical regression models by including spatial autocorrelation as a coefficient. Spatial regression models range from the simpler SLX model, which only includes the spatial lag of the covariates, all the way up to more complex models which account for temporal and spatially lagged dependent and independent variables and error terms (Ewing & Park, 2020).

In the present study, we used the **Durbin spatial model**. Considering a neighbourhood *i* with a dependent variable *y*, then the Durbin model is defined as

$$y_i = \rho_i \sum_j w_{ij} y_j + \beta_i x_i + \gamma_i \sum_j w_{ij} x_j + \epsilon_i$$
(3.9)

where w is the spatial weight matrix and ϵ is the normally distributed error term (Anselin, 1988). In the Durbin model, ρ is the spatial autoregressive coefficient, β is the general linear model coefficient, and γ is the coefficient of spatially lagged covariates. The **spatial lag** of a variable is the weighted average of neighbouring values, defined by w.

Models that include spatial lag, like Durbin's, consider that autocorrelation arises from a diffusive process between locations. This process could be in the form of **spillovers**, where effects at one location influence neighbouring locations; or could indicate **contagion**, where effects at one location causally influence other locations (Ewing & Park, 2020; Rey et al., 2020). For instance, if social cohesion is high in a certain neighbourhood, this may spill over into near neighbourhoods, as social cohesion is not constrained by imaginary administrative boundaries. The Durbin model includes lagged dependent and independent variables to include the diffusive effect of both. For example, social cohesion in a neighbourhood might be affected by the level of balance of the surrounding neighbourhoods.

3.4. Computational framework and software implementation

The gathered data and research methods come together in the computational framework (see Figure 3.5). It provides a clear, graphic way of how the data and methods are used to test the research hypotheses and respond to the research questions in this study. The framework is complemented with Table 3.6, which shows the used packages in the research flow to apply the selected methodologies.

First, data is collected and then pre-processed. For every possible distribution that the definition of balance from the municipality of Rotterdam can yield (see Figure 2.9), the *Neighbourhood balance* of each neighbourhood is calculated. Having this, all the possible measurements of *Neighbourhood balance* are fed into the PLS-SEM model. The instances where the model does not show appropriate fit are filtered out. In the next step, the models that fit were clustered using the k-means algorithm and the model that had the best fit in each cluster is selected to study the hypotheses. This ensures that we have a representative samples of the possible combinations of balance. We begin by analysing the mediation mechanism between *Neighbourhood balance, Social cohesion,* and *Informal support* to respond to the hypotheses **H1**, **H2**, **H3**, and **H4**. This is followed by a spatial exploratory analysis using LISA to then respond to **H5** and **H6** using the Durbin model. Finally, **H7** to **H12** are assessed in the moderation analysis using again



Figure 3.5: Computational framework. This diagram is meant to be read from top to bottom, each number indicates the step of analysis, and each arrow is the flow between an output and input. The "Providers" row shows the data owners of the data used in this research. "Data collection" shows the output data from each provider. "Data preprocessing" describes the specific data wrangling steps for each type of data. The whole row is highlighted because all the raw data had to be aggregated to the survey areas of the Wijkprofiel as explained in Section 3.1.4. The "KL divergence" row shows the use of that technique to measure *Neighbourhood balance*. The "K-means clustering" row shows the application of clustering to the models that fit. The "PLS-SEM" row presents the analysis done using the methodology of the same name. Finally, "Spatial econometrics" shows the techniques used for spatial analysis.

3.5. CONCLUSION

The aim of this section was to identify the data that could transfer the conceptual model to a set of measurable indicators, show the proposed research model, and to identify the research methods that can satisfy all the requirements for the analysis.

The measurement of the level of balance of a neighbourhood comes from the hand of a single-indicator, being *Neighbourhood balance*. This is obtained by calculating the Kullback-Leibler divergence between the WOZ-value distribution of a neighbourhood and the possible distributions that imply balance. Given the ambiguity due to the range of possible house stock distributions that can achieve balance, we are interested in quantifying the possible outcomes that can result from the feasible ranges of the definition of balanced neighbourhood. The KL divergence permits the identification and comparison of how far each neighbourhood is to achieve the defined balance for each combination. Furthermore, it is a versatile technique which is transferable between contexts as the target distribution can be adapted accordingly.

49

Research method	Language	Package name	Reference
KL Divergence	Python	SciPy	Virtanen et al. (2020)
K-means	Python	SciPy	Virtanen et al. (2020)
PLS-SEM	R	cSEM	Rademaker and Schuberth (2020)
LISA	Python	PySAL	Rey and Anselin (2007)
Durbin model	Python	PySAL	Rey and Anselin (2007)

Table 3.6: Software implementation of the research methods.

Social cohesion and *Informal support* can be operationalised using the data collected by the Wijkprofiel survey in Rotterdam. The survey is already structured into validated psychometric scales, which easily allow us to identify the relevant indicators to measure the constructs. Furthermore, the Wijkprofiel is a tool used by the municipality for the analysis and development of local policies, which strengthens the translation of the model to real application.

The chapter also describes the sources and the pre-processing of the following exogenous factors identified in the literature: *Ethnic heterogeneity, Residence length, Population density, House type heterogeneity, Greenery, Walkability,* and *Amenities.* These variables will be used in moderation analysis to overcome the supposition that the relationships are exactly the same for all neighbourhoods and identify instances in which the moderating variables change the strength or direction of the relationship between the concepts

To analyse the relationships, two main research methods are proposed. PLS-SEM allows estimating constructs made up from multiple indicators modelled either formatively or reflectively. In addition, it can estimate the direct effects, and mediating effects of the constructs as well as the moderating effects of the exogenous variables. As a second method, spatial econometrics provides insights on the dependence of the results on the spatial distribution of the neighbourhoods, and of the possible spillover or contagion effects by means of LISA and the Durbin model. Putting all together creates a computational framework which is the foundations of the analysis of this research.

4

RESULTS

This chapter presents the results of the case study analysis. The developed computational framework is applied to the city of Rotterdam to understand the relationship between the level of balance of neighbourhoods, their social cohesion, and their informal support.^{1,2}

The chapter starts with an analysis of the definition of balance from the municipality of Rotterdam. Then, the house WOZ-value balance distributions that best fit the model are selected for further analysis. The following section presents the results of the PLS-SEM mediation analysis and shows the relationships between the model constructs in the year 2019. The results of spatial econometrics are then showed to quantify the relevance of spatial effects in the role between the model constructs. Finally, the outcomes of the external factors moderation analysis are exhibited.

4.1. BALANCE OF ROTTERDAM

The first step of the research process was to analyse the level of balance of Rotterdam's neighbourhoods. The definition provided by the municipality of Rotterdam (see Figure 2.9), is formulated in such a way that balance is a dichotomous variable. If the neighbourhood complies with the conditions it is in balance, otherwise it is not. These conditions can be translated to the following set of inequalities:

¹There was an attempt to operationalise *Social cohesion* by means of measurement theory using Exploratory Factor Analysis. The results indicate that no clear factors could be identified, so we conclude that from the data available, *Social cohesion* is best modelled formatively. Appendix B.2 shows the methodology and the obtained results.

²The presented model conceptualised *Informal support* as willingness to help friends or neighbours. Appendix C shows the results of adding indicators of the willingness to help relatives, and of the willingness to help others in the surroundings. We saw that the willingness to help others does not correlate with the willingness to help relatives, friends, and neighbours, so a common factor could not be found. In addition, informal support as willingness to help others does not show an acceptable model fit. Finally, a common factor was found between willingness to help relatives and willingness to help friends and neighbours, which yielded an acceptable model fit. Appendix C and the discussion chapter provide further details on these results.



Figure 4.1: Full solution space for the definition of balance according to the municipality of Rotterdam. Only integer solutions are shown.

 $0\% \le \text{Social} \le 60\%$ $45\% \le \text{Social} + \text{Middle} \le 100\%$ $40\% \le \text{Middle} + \text{Higher} + \text{Top} \le 100\%$ $0\% \le \text{Middle} \le 100\%$ $0\% \le \text{Higher} \le 100\%$ $0\% \le \text{Top} \le 100\%$ Social + Middle + Higher + Top = 100\%

Figure 4.1 shows the full solution space that satisfies the inequalities established by the conditions of the municipality to have balance. Any combination within that space that adds up to 100% is an acceptable solution, and thus the municipality will argue that the neighbourhood is in balance. We can see that even though the municipality distinguishes Higher and Top as two different brackets, they do not specify a difference in their definition of balance, thus they can be combined.

Given that the definition of balance from the municipality is not a unique distribution, but a whole solution space, it is of interest to analyse (i) the dispersion of the definition, and (ii) the effect that this might have in the analysis of the relationship between a balanced neighbourhood and its social cohesion and informal support.

As explained in Section 3.3, the KL divergence is a suitable technique to operationalise the measurement of the level of balance of a neighbourhood, i.e. the *Neighbourhood balance*. For that, we need an objective distribution from which to measure the distance of the actual WOZ-value distribution in the neighbourhood. Therefore, as a next step in the exploration of the definition from the municipality, *Neighbourhood balance* was calculated for every neighbourhood for all the possible integer configurations that add up to 100%.


Figure 4.2: Each line shows the density estimation plot for each of the neighbourhoods in the Wijkprofiel of the measurement of *Neighbourhood balance* for every possible combination of Social, Middle, and Higher+Top that satisfy balance according to the municipality of Rotterdam. Density plots are a smoothed version of the histogram and are used to study the distribution of data (the smoothening shows possible values above zero, even though we have seen that the measurement of *Neighbourhood balance* does not allow that, but density estimation is better for visual comparison of the multiple neighbourhoods). The vertical dotted line shows the average median value of -0.79 nats.

Starting with (i) the dispersion, each line in Figure 4.2 shows the density plot of the measurement of Neighbourhood balance for each neighbourhood for every possible integer distribution that satisfies the conditions for balance in Figure 2.9. We can see how the Neighbourhood balance calculated from the KL divergence is distributed for each neighbourhood for all the integer configurations in the solution space exhibited in Figure 4.1. Given that the data is skewed, we use the range, interquartile range and outliers to interpret the dispersion, as the standard deviation and mean are sensitive to outliers (Manikandan, 2011).³ The figure shows the large range of the measurement of *Neigh*bourhood balance, with an average range of -3.63 nats. The neighbourhood with the largest range is Beverwaard, for which the minimum measurement of *Neighbourhood balance* is -0.51 nats, and the largest -4.61 nats. The average interquartile range is -0.86 nats, which indicates that 50% of the data is distributed within 23% of the average total range. Having the interquartile range, we found that the average percentage of outliers per neighbourhood is 5%. Making an analogy with a perfectly normal distribution which has 0.3% of outliers (Ruan, 2005), our results suggest that the definition of balance from the municipality has a considerable variability and is susceptible to outliers.

 $^{^{3}}$ The range is the difference between the highest and the lowest values in a distribution (Manikandan, 2011). The interquartile range measures the spread 50% of the data. Outliers are defined as values which are 1.5 interquartile ranges above and below the 25% and 75% quartiles.



Figure 4.3: House stock distributions within the balance definition of Rotterdam that have an acceptable goodness-of-fit ($\approx 2.1\%$) in the PLS-SEM model.

Continuing with (ii), to obtain further insights about the implications of the possible distributions of balance, the PLS-SEM model presented in Figure 3.1 was evaluated for all the *Neighbourhood balance* measures for each of the possible integer configurations that add up to 100% in Figure 4.1. Only the models which reported an acceptable standardised root mean squared residual (SRMR) measure were considered. The SRMR is a goodness-of-fit measure that quantifies how strongly the empirical correlation matrix differs from the model-implied correlation matrix (Henseler, 2020). A value of 0 for the SRMR would indicate a perfect fit and the generally accepted threshold is of 0.08 established by Hu and Bentler (1999). Therefore, models with an SRMR below this threshold and which meet the 95% confidence interval (CI) quantile criteria for the estimated models were considered. Out of 3,162 models — the total number of possible integer configurations for the distribution of balance—, only 66 ($\approx 2.1\%$) met these criteria (see Figure 4.3)

From Figure 4.3, we can infer that the models that fitted are separated into different groups with similar attributes. To unravel the structure of these groups, the results were clustered using the k-means algorithm. To begin, k-means requires the specification of the number of clusters. The elbow method is a common way of choosing the optimal number of clusters. The method assumes that the optimal number of clusters is that for which adding another cluster doesn't give much better representation of the data (Pedregosa et al., 2011). First, we calculate the sum-squared error (SSE) of the k-means algorithm for an arbitrary range of 1 to 10 clusters. Then, the number of clusters that minimises the SEE without overfitting is the recommended number of clusters. This corresponds to the elbow of the plot, as can be seen in Figure 4.4. In this case, the elbow method recommended 3 different clusters (Figure 4.5).

Figure 4.6 shows the values of the significant path coefficients (95% CI) for each model within each cluster. While the relationship between *Social cohesion* and *Infor*-



Figure 4.4: Elbow method for the optimal number of clusters. The vertical dotted line indicates the position of the elbow at 3 clusters.



Figure 4.5: Clusters of the house stock distributions that meet the definition of balance from the municipality and that have an acceptable goodness-of-fit. Clusters 1, and 2 exhibit concave shapes, while Cluster 3 has a convex shape.



Figure 4.6: Boxplot of the path coefficients for each cluster. SC~NB: path coefficient of the relationship between *Neighbourhood balance* and *Social cohesion*. IS~NB: Path coefficient of the relationship between *Neighbourhood balance* and *Informal support*. IS~SC: Path coefficient of the relationship between *Social cohesion* and *Neighbourhood balance*. Only the path coefficients significant to the 95% CI are shown.

mal support is similar throughout all clusters, we can see that the relationship between *Neighbourhood balance* and *Social cohesion*, and the relationship between *Neighbourhood balance* and *Informal support* change direction in Cluster 3 compared to the others. In addition, Cluster 3 does not have any fit model with a significant relationship between *Neighbourhood balance* and *Informal support*. To provide further insights about the relationships and the effect of the different distributions in the model, we selected the model from each cluster that had the best goodness-of-fit, i.e. the lowest SRMR value (Figure 4.7).

4.2. PLS-SEM MODEL RESULTS

Before presenting the outcomes of the selected models, we first show the values for the model fit tests. After goodness-of-fit is guaranteed, the measurement models are evaluated to determine which items better form the model constructs. We begin with the assessment of the reflective measurement models, i.e. *Informal support*, and we follow with the assessment of the formative measurement models, i.e. *Social cohesion* and *Neighbourhood balance*. Finally, the reliability and validity of the inner structural model is assessed and the relationships between the constructs are presented.

4.2.1. MODEL GOODNESS-OF-FIT

We start by reporting the estimated model misfit using non-parametric model fit tests (i.e. dG, dM, and dML) and a model fit index (i.e. SRMR) (Table 4.1). The different nonparametric tests calculate the difference between the empirical and the model-implied indicator variance-covariance matrix using different distance measures (Rademaker & Schuberth, 2020). The lower the value of the distance, the better the model fit. On the other hand, the standardised root mean squared residual (SRMR) quantifies how strongly the empirical correlation matrix differs from the model-implied correlation matrix (Henseler, 2020). A value of 0 for the SRMR would indicate a perfect fit. As seen



Figure 4.7: Housing stock distributions of the models with best goodness-of-fit for each identified cluster.

5	0
э	0

	Model 1		Мо	del 2	Model 3		
	Value	95% CI	Value	95% CI	Value	95% CI	
dG	0.044	0.057	0.040	0.056	0.050	0.055	
SRMR	0.042	0.067	0.043	0.069	0.050	0.060	
dL	0.049	0.125	0.053	0.132	0.069	0.102	
dML	0.218	0.278	0.202	0.272	0.247	0.267	

Table 4.1: Goodness-of-fit of the estimated models.

earlier, the selected models scored below the generally accepted threshold of 0.08 established by Hu and Bentler (1999), which indicates an acceptable fit. Furthermore, they meet the 95% confidence interval (CI) quantile criteria for all the goodness-of-fit measures.

4.2.2. REFLECTIVE MEASUREMENT MODEL

Examining the internal consistency reliability is the first step in evaluating a reflective measurement model. For that, Dijkstra-Henseler's rho, ρ_A , should lie between 0.70 and 0.95 to guarantee a good level of internal consistency that avoids indicator redundancy (Dijkstra & Henseler, 2015). Given that only one indicator was used, the value of ρ_A equals one, which suggests that it is not possible to determine the amount of random measurement error in the indicator based on the available data (Henseler, 2020). Literature recommends that, in these cases, an additional error is imposed from the modeller (Brown, 2015), however, the package cSEM does not have that feature yet. Therefore, we acknowledge this limitation in the measurement of *Informal support*.

The following step of the reflective measurement model assessment addresses the convergent validity of each construct. Convergent validity is the extent to which the construct converges to explain the variance of its items (Henseler, 2020). To this end, the average variance extracted (AVE) for all items on each construct is used. The AVE, is the mean-squared loading of each indicator. The AVE of *Informal support* equals 1, which indicates that the construct explains 100% of the variance of its item (Henseler, 2020).

The third step is the assessment of indicator loadings. Although loading values of 0.7 or higher are regarded as highly satisfactory, value loadings of 0.5 or higher are considered acceptable (Chin, 1998; Hair, 2009). As *IS1* explains the whole variance of the extracted latent variable *Informal support*, it has a loading equal to 1.

A final step in the assessment of the reflective constructs is discriminant validity. This determines to what extent a construct is empirically distinct from other constructs in the model. Since we only have one concept modelled as a latent variable from common factors, discriminant validity does not need to be assessed.

4.2.3. FORMATIVE MEASUREMENT MODEL

Assessing the convergent validity is the first step in the evaluation of a formative measurement model. The model goodness-of-fit with an SRMR below 0.08 within the 95% CI (see Table 4.1) indicates the confirmatory power of our emergent construct and is

Tabl	e 4.2:	Formative	construct	assessment.
------	--------	-----------	-----------	-------------

		Models 1, and 2		Model 3	
	Item	Weight	95% CI	Weight	95% CI
	SC1: % of residents who say that local residents know each other	NA ^a	NA ^a	NA ^a	NA ^a
Social Cohesion	SC2: % of residents who say that local residents spend a lot of time with each other	NA^{b}	NA^{b}	NA^{b}	NA^{b}
	SC3: % of residents who say that local residents share each other's views	1.00	[1.00, 1.00]	NA^{b}	NA^{b}
	SC4: % of residents who say that local residents help each other	NA^{b}	NA ^b	NA^{b}	NA^{b}
	SC5: % of residents who say they feel at home with local residents	NA^{b}	NA^{b}	1.00	[1.00, 1.00]
Neighbourhood	NB: Negative of the Kullback-Leibler divergence of the WOZ-value distribution in	1.00	[1.00.1.00]	1.00	[1.00.1.00]
Balance	a neighbourhood with respect to the objective balance distribution	1.00	[1.00, 1.00]	1.00	[1.00, 1.00]

a: Item dropped due to high multicollinearity *b*: Item dropped due to insignificant weight.

suggestive of its reliability and validity (Henseler, 2020).

Next, the statistical significance and size of the indicators weights were assessed. *Neighbourhood balance* is a single-indicator construct, which is to say that its value is inferred from only one observed variable. This means that the item is a perfect ingredient of the emergent variable. On the other hand, *Social cohesion* is a multiple indicator formative construct. When estimating the weights using Mode B as described in Section 3.3.3, the results showed negative signs for the weights corresponding to *SC1* and *SC3*. Negative weights can be indicative of a negative contribution of the items to the construct. However, according to the qualitative theory expectations, all the items should contribute positively to *Social cohesion*. This suggests that negative weights are caused by multicollinearity between the items (Diamantopoulos & Winklhofer, 2001). The variance inflation factor (VIF) is often used to evaluate multicollinearity. VIF values of 5 or above indicate critical multicollinearity issues among the indicators of formatively measured constructs. Ideally, the VIF values should be close to 3 and lower. The VIF showed that the items of *Social cohesion* were above 5. Mode BNNLS is recommended in these situations, as explained in Section 3.3.3.

When applying Mode BNNLS to Models 1, and 2, PLS-SEM reduced *Social cohesion* to the % of residents who say that local residents share each other's views (SC3). On the other hand, when applied to Model 3, *Social cohesion* was reduced to the % of residents who say they feel at home with local residents (SC5) (see Table 4.2).

4.2.4. STRUCTURAL MODEL

After guaranteeing that the measurement model is satisfactory, the structural model results were assessed. First, the Models showed acceptable explanatory power, but low predictive accuracy (details can be found in Appendix D.1). Given that the research approach is deductive and not predictive, this outcome is not considered inadequate as we are mainly concerned about the goodness-of-fit and the values of the path coefficients (Henseler, 2020).

The final step is to assess the statistical significance and relevance of the path coefficients. Table 4.3 presents the total, direct, and indirect effects of the path analysis (β). In mediation analysis, the direct path coefficient represents the effect of exposure on the outcome in the absence of the mediator. On the other hand, the indirect path coefficient shows the effect that acts through the mediator. The total effect is the sum of direct and indirect path coefficients. Another consideration to take into account is that the path coefficients are standardized. This means that, for example, in Model 1 with a path co-

efficient between *Neighbourhood balance* and *Social cohesion* of -0.467, a 10% increase in the *Neighbourhood balance* would report a decrease of 0.0467 standard deviations in the *Social cohesion* of the target neighbourhood.

The results of the effects can be summarised as follows. First, in Models 1, and 2, *Neighbourhood balance* has a significant negative association with *Social cohesion* (H2 Neighbourhood balance positively affects Social cohesion: Not supported). Second, we see a positive association between *Social cohesion* and *Informal support* (H1 Social cohesion positively affects Informal support: Supported). Finally, the direct relationship between *Neighbourhood balance* and *Informal support* is non-significant (H3: Neighbourhood balance positively affects Informal support: Not supported), so we can talk about full mediation (Henseler, 2020), the effects of balance on willingness to help are due to social cohesion. (H4: The relationship between Neighbourhood balance and Informal support is mediated by Social cohesion: Supported).

We can see different results for Model 3. Here, *Neighbourhood balance* is positively associated with *Social cohesion*, but non-significantly (**H2** Neighbourhood balance positively affects Social cohesion: Not supported). Second, the association between *Social cohesion* and *Informal support* is again positive (**H1** Social cohesion positively affects Informal support: Supported). Finally, as in the previous models, the relationship between *Neighbourhood balance* and *Informal support* is non-significant (**H3**: Neighbourhood balance positively affects Informal support: Not supported). Therefore, Model 3 does not show mediation effects (**H4**: The relationship between Neighbourhood balance and Informal support: Not supported).

Even though the relationship between *Neighbourhood balance* and *Social cohesion* in Model 3 is non-significant, the change of sign between Models 1, and 2, and Model 3 is of interest. Especially because, as we saw in Figure 4.6, other models in the same cluster showed significant relationships. This could be explained by two reasons. First, the different specification of *Social cohesion* could change the direction of the relationship if % of residents who say that local residents share each other's views (SC3; Models 1, and 2) and % of residents who say they feel at home with local residents (SC5; Model 3) are negatively related. Second, the distribution of balance in Model 3, yields a *Neighbourhood balance* measurement which is negatively related to the measure from the other distributions. After examination, we saw that *SC3* and *SC5* are positively related and highly collinear (r^2 =0.8, VIF>3), which suggests that the second argument is the reason for the results. This can be explained by the fact that the distribution in Model 3 is convex, whereas the distributions for the other models are concave (see Figure 4.5). Chapter 5 discusses the reasons and implications of these results.

The results show that the PLS-SEM successfully estimated the model constructs and yielded acceptable explanatory power, although low predictive accuracy. Model 1 had the best overall goodness-of-fit, but Model 3 captured the relationship between *Social cohesion* and *Informal support* with better explanatory power. Overall, the latent construct *Informal support* expressed as the % of residents who say they are willing to care for neighbours or friends who need help (IS1) had a low explanatory power. The emergent construct *Social cohesion* was reduced to the % of residents who say that local residents share each other's view (SC3) when the objective balance distribution is concave (Models 1, and 2), or to the % of residents who say they feel at home with local residents (SC5)

	_		Model 1			Model 2			Model 3	
Dependent	Independent	β	β	β	β	β	β	β	β	β
variable	variable	Total	Direct	Indirect	Total	Direct	Indirect	Total	Direct	Indirect
Informal										
Support										
	Social	0 352**	0 352**		0 351**	0.351**		0 325**	0 325**	
	Cohesion	0.332	0.332		0.551 0.55	0.331	0.551	0.525	0.525	
	Neighbourhood	-0.067	0.097	-0.164	-0.037	0.108	-0.145	0.141	0.037	0.104
	Balance	-0.007	0.057	-0.104	-0.037	0.100	-0.145	0.141	0.037	0.104
Social										
Cohesion										
	Neighbourhood	0 467***	0 467***		0.412***	0.4128**		0 221	0 221	
	Balance	-0.407	-0.407		-0.412	-0.412		0.321	0.321	

Table 4.3: Structural model total, direct, and indirect path coefficients.

Significance levels: *p<0.001 **p<0.01 ***p<0.05

Note: No superscript indicates statistically non-significant values.

when the balance distribution is convex (Model 3). This yields two different possible results. On one hand, concave balance distributions to measure *Neighbourhood balance* are negatively associated to *Social cohesion* and only related to *Informal support* by full mediation. On the other hand, convex balance distributions to measure *Neighbourhood balance* are positively associated to *Social cohesion* and again related to *Informal support* by full mediation. Finally, *Social cohesion* and *Informal support* are positively related in all instances.

4.3. Spatial econometrics results

This section elevates the results from the PLS-SEM model to the geographical level. First, the impact of the different distributions of balance in the measurement of *Neighbour*-*hood balance* are analysed. Then the spatial effects in *Social cohesion* are assessed, followed by the effects in *Informal support.*⁴

4.3.1. Spatial distribution of Neighbourhood balance

To introduce this section, Figure 4.8 shows the different outcomes in *Neighbourhood balance* that result from the chosen distribution in each model. Overall, we can see that *Neighbourhood balance* increases from the South of the city towards the North.

When comparing the different models, we can however see a considerable difference in the measurement of *Neighbourhood balance*. Models 1, and 2 present very similar outcomes, showing this increase in balance towards the North, but with less extreme values. On the other hand, Model 3 assigns stronger levels of imbalance to the South of Rotterdam and stronger levels of balance towards the North. Not only that, but we can also see how some neighbourhoods which in the first three models were considered with a very low *Neighbourhood balance*, Model 3 assigns them a high level of *Neighbourhood balance*. The same happens in the opposite way. Neighbourhoods like Blijdorp which presented high levels of *Neighbourhood balance* in Models 1, and 2, in Model 3 decrease their level. This can be explained by the earlier mentioned change in balance distribution shape. While Models 1, and 2 measure *Neighbourhood* balance from con-

⁴The main results do not take into account the river that divides Rotterdam. Appendix E shows the results considering that the river cancels any type of spillover effect from both sides. The results are talk about in the discussion chapter.



Figure 4.8: Choropleth of the standardised measure of *Neighbourhood balance* in 2019 for every model. The neighbourhoods of Noord Kethel, Strand en Duin, and Nesselande are outliers and are removed from the figure for clarity (Appendix D.2 shows the values for each neighbourhood in each Model).

cave distributions, Model 3 does from a convex distribution (see Figure 4.5). As a result, some measurements are flipped.

4.3.2. SPATIAL EFFECTS IN SOCIAL COHESION

The PLS-SEM model reduced *Social cohesion* to the % of residents who say that local residents share each other's view (SC3) for Models 1, and 2; and to the % of residents who say they feel at home with local residents (SC5) for Model 3. This section begins with a spatial exploration of these results and continues with the use of spatial econometrics to quantify the implications of the spatial arrangement.

Figure 4.9 shows a vertical gradient in *Social cohesion* from North to South in Rotterdam. The neighbourhoods in the South suffer from a lower level of *Social cohesion*, as well as those in the centre of the city. We can also see how neighbourhoods to the West, which are not in the periphery of Rotterdam's centre, also express higher levels of *Social cohesion*. While *Social cohesion* expressed as the % of residents who say that local residents share each other's view (Models 1, and 2; *SC3*) yields higher values towards the positive end, % of residents who say they feel at home with local residents (Model 3; *SC5*) does the opposite. Even so, we can see that the spatial structure has a similar pattern for all models.

To visually explore the relationship between *Social cohesion* and *Neighbourhood balance*, Figure 4.10 shows the distribution of both measurements in Rotterdam. We can observe two different associations. Foremost, we can observe the similarity between the choropleths for Models 1, and 2 given that they measure *Social cohesion* the same way. They show how the balance tends to increase towards the North of Rotterdam, but that the increase in *Social cohesion* towards the North is less progressive, given the low *Social cohesion* in the city centre. While a few neighbourhoods with higher *Neighbourhood balance* and *Social cohesion* can be spotted towards the North, in these models they



Figure 4.9: Choropleth of the standardised measure of Social cohesion in 2019 for every model.

are negatively associated, given the large spread of lower *Social cohesion* throughout the city. The choropleth for Model 3 shows a different story. We can see a progression from neighbourhoods with low *Social cohesion* and low *Neighbourhood balance* in the South, to high in the North, due to the change of the distribution for balance in Model 3 from concave to convex. Nevertheless, the reader is reminded that in the case of Model 3 the relationship between these variables was not significant to the 95% CI, but to the 90%. To further explore the spatial distribution of *Social cohesion*, Appendix D.3 shows its LISA classification.



Figure 4.10: Choropleth of the distribution of *Social cohesion* and *Neighbourhood balance* in 2019. Grey shades represent neighbourhoods in which *Social cohesion* and *Neighbourhood balance* are low. Shades of blue represent neighbourhoods in which the level of *Neighbourhood balance* is high. Shades of red represent counties in which *Social cohesion* is high. The values in the shades were determined by Jenks natural breaks optimization.

Variable	Model 1	Model 2	Model 3
Intercept	-0.025	-0.027	-0.063
	(0.099)	(0.102)	(0.103)
Neighbourhood Balance	-0.458*	-0.403*	0.227***
	(0.102)	(0.104)	(0.104)
Lagged Neighbourhood Balance	-0.242	-0.174	0.219
	(0.173)	(0.168)	(0.163)
Lagged Social Cohesion	0.329***	0.375**	0.404**
	(0.138)	(0.133)	(0.130)
Log-likelihood	-81.818	-84.046	-84.648
Akaike information criterion	171.635	176.092	177.297
R ² (pseudo)	0.365	0.330	0.293

Table 4.4: Results of the Durbin spatial model with *Social cohesion* as dependent variable. Lagged variables are the weighted averages of neighbouring values. Significant values of the lagged variables indicate spillover or contagion effects. In brackets, the standard deviations.

Significance levels: *p<0.001 **p<0.01 ***p<0.05

Note: No superscript indicates statistically non-significant values.

To quantify the strength of the possible spillover or contagion effects, Table 4.4 shows the results of Durbin's spatial autoregressive model with *Social cohesion* as dependent variable and *Neighbourhood balance* as the independent variable. From the results, we can see that *Social cohesion* demonstrates such effects in all models (**H5a** Spatial spillover effects are significant on the relationship between Neighbourhood balance and Social cohesion: Supported). For example, in Model 1, a 10% increase in the average surrounding *Social cohesion* would report an increase of 0.0329 standard deviations in the *Social cohesion* of the target neighbourhood. This effect is of the same order as the direct effect of *Balance neighbourhood* on *Social cohesion*, however, *Balance neighbourhood* did not report significant spillover or contagion effects. This suggests that the PLS-SEM model results are limited, as they do not account for these spatial effects.

4.3.3. Spatial effects in Informal support

This section studies the relationship between *Social cohesion* and *Informal support*. The relationship between *Neighbourhood balance* and *Informal support* was not further analysed because the PLS-SEM results pointed out no significant relationship between the latter in any of the Models.

Figure 4.11 shows how, in comparison to *Social cohesion*, the division in *Informal support* between North and South is less accentuated, although a gradient can still be observed. The choropleths for each model are the same, as they all specified *Informal support* using the same single-indicator.

To visually explore the association of *Informal support* with *Social cohesion*, Figure 4.12 shows the distribution of both constructs in the city. We can see that neighbourhoods in the South of the city are mostly associated with low *Social cohesion* and



Figure 4.11: Choropleth of the standardised measure of Informal support in 2019 for all Models.



Figure 4.12: Choropleth of the distribution of *Informal support* and *Social cohesion* in 2019. Grey shades represent neighbourhoods in which *Informal support* and *Social cohesion* are low. Shades of blue represent neighbourhoods in which the level of *Social cohesion* is high. Shades of red represent counties in which *Informal support* is high. The values in the shades were determined by Jenks natural breaks optimization.

low *Informal support*. As we move towards the North, these two variables tend to increase. Most differences between Model 3 and the rest can be identified in the central and to the East and West of it. This shows that the polarisation is consistent and only fuzzier in the border zone between North and South. A more nuanced exploration of the relevance of the geographical distribution can be found in Appendix D.3, which shows the LISA clusters for *Informal support*.

The presence of clusters is suggestive of possible spillover or contagion effects. Table 4.5 shows the results of Durbin's spatial autoregressive model with *Informal support* as dependent variable and *Neighbourhood balance* and *Social cohesion* as independent variables. The results from Models 1 and 2 suggest that *Informal support* of other neighbourhoods spill over (**H5b** Spatial spillover effects are significant on the relationship between Neighbourhood balance and Informal support: Supported). The spillover or con-

Variable	Model 1	Model 2	Model 3
Intercept	-0.010	-0.010	0.002
	(0.110)	(0.110)	(0.108)
Neighbourhood Balance	0.094	0.096	-0.027
	(0.128)	(0.123)	(0.114)
Lagged Neighbourhood Balance	0.156	0.162	0.097
	(0.210)	(0.191)	(0.166)
Social Cohesion	0.371**	0.366**	0.385**
	(0.136)	(0.133)	(0.130)
Lagged Social Cohesion	-0.105	-0.106	-0.125
	(0.228)	(0.220)	(0.187)
Lagged Informal Support	0.301***	0.293***	0.261
	(0.143)	(0.144)	(0.147)
Log-likelihood	-88.072	-87.971	-86.945
Akaike information criterion	188.145	187.941	185.889
R ² (pseudo)	0.185	0.186	0.205

Table 4.5: Results of the Durbin spatial model with *Informal support* as dependent variable. Lagged variables are the weighted averages of neighbouring values. Significant values of the lagged variables indicate spillover or contagion effects. In brackets, the standard deviations.

Significance levels: *p<0.001 **p<0.01 ***p<0.05

Note: No superscript indicates statistically non-significant values.

tagion of *Social cohesion* is not significant enough to have an effect in this process. In Model 3, no significant spillover or contagion effects can be seen (**H5b** Spatial spillover effects are significant on the relationship between Neighbourhood balance and Informal support: Not supported).

4.4. RESULTS OF MODERATION ANALYSIS

In this section, moderation analysis is conducted to overcome the supposition that the relationships between neighbourhood balance, social cohesion, and informal support are exactly the same for all neighbourhoods and identify instances in which the moderating variables change the strength or direction of the relationship between the concepts (Henseler, 2020).

The results of moderation analysis can be summarised as follows (see Table 4.6). Social cohesion is negatively associated with the *Ethnic heterogeneity* of the neighbourhoods in all models. No other exogenous factors were directly related to Social cohesion or moderated the relationship between *Neighbourhood balance* and Social cohesion (**H6a**, **H7a**, **H8a**, **H9a**, **H10a**, and **H11a**: Not supported). Next, Table 4.6 shows that *Informal support* is only directly related to *House type heterogeneity* in Model 3, with no significant moderation effects. No other exogenous factors were directly related to *Informal support* or moderated the relationship between *Neighbourhood balance* and *Informal* Table 4.6: Results of moderation analysis. In bold the effects which are significant up to 95% CI.

		Model 1			Model 2	Model 3		
Dependent	Indonondont		Model I		Model 2		would 5	
variable	variable	γ	95% CI	γ	95% CI	γ	95% CI	
Social	Vurlable							
Cohesion								
	Neighbourhood balance (NB)	-0.326	[-0.578, 0.122]	-0.234	[-0.479, 0.189]	0.471	[0.238, 0.684]	
	Ethnic heterogeneity	-0.705*	[-0.910, -0.307]	-0.711*	[-0.926, -0.344]	-0.585*	[-0.861, -0.344]	
	Residence length	0.118	[-0.147, 0.336]	0.121	[-0.141, 0.330]	0.093	[-0.097, 0.288]	
	Population density	-0.225	[-0.523, 0.067]	-0.205	[-0.498, 0.076]	-0.004	[-0.254, 0.273]	
	House type heterogeneity	0.050	[-0.287, 0.306]	0.046	[-0.314, 0.337]	-0.089	[-0.300, 0.126]	
	Greenery	-0.197	[-0.404, 0.043]	-0.208	[-0.401, 0.059]	0.088	[-0.171, 0.350]	
	Walkability	0.099	[-0.293, 0.417]	0.082	[-0.281, 0.433]	0.082	[-0.227, 0.419]	
	Amenities	-0.088	[-0.447, 0.206]	-0.107	[-0.452, 0.168]	-0.099	[-0.379, 0.167]	
	NB × Ethnic heterogeneity	-0.139	[-0.677, 0.528]	-0.172	[-0.784, 0.441]	0.215	[-0.238, 0.604]	
	NB × Residence length	-0.102	[-0.282, 0.395]	-0.042	[-0.235, 0.426]	-0.015	[-0.324, 0.282]	
	NB × Population density	0.251	[-0.509, 0.821]	0.109	[-0.548, 0.604]	0.054	[-0.328, 0.453]	
	NB × House type heterogeneity	-0.112	[-0.456, 0.466]	-0.080	[-0.463, 0.551]	-0.047	[-0.326, 0.346]	
	NB × Greenery	-0.035	[-0.356, 0.460]	-0.038	[-0.377, 0.446]	-0.047	[-0.35, 0.205]	
	NB × Walkability	0.124	[-0.727, 0.922]	0.262	[-0.381, 1.003]	-0.006	[-0.341, 0.370]	
	NB × Amenities	-0.166	[-1.001, 0.771]	-0.096	[-0.815, 0.749]	-0.063	[-0.392, 0.347]	
Informal								
Support								
	Neighbourhood balance (NB)	0.195	[-0.380, 0.747]	0.146	[-0.408, 0.672]	0.255	[-0.188, 0.580]	
	Social cohesion	-0.001	[-0.544, 0.692]	0.011	[-0.509, 0.699]	0.023	[-0.455, 0.586]	
	Ethnic heterogeneity	-0.437	[-0.898, 0.289]	-0.451	[-0.974, 0.290]	-0.303	[-0.787, 0.137]	
	Residence length	-0.083	[-0.441, 0.220]	-0.079	[-0.394, 0.204]	-0.070	[-0.392, 0.209]	
	Population density	-0.019	[-0.394, 0.506]	0.053	[-0.466, 0.604]	0.203	[-0.146, 0.683]	
	House type heterogeneity	-0.296	[-0.783, 0.194]	-0.310	[-0.837, 0.152]	-0.470**	[-0.752, -0.144]	
	Greenery	-0.245	[-0.516, 0.060]	-0.226	[-0.470, 0.147]	0.129	[-0.247, 0.458]	
	Walkability	0.092	[-0.596, 0.454]	0.118	[-0.505, 0.578]	0.127	[-0.481, 0.622]	
	Amenities	0.000	[-0.283, 0.708]	0.085	[-0.248, 0.734]	0.109	[-0.362, 0.779]	
	NB × Ethnic heterogeneity	-0.415	[-1.200, 0.378]	-0.413	[-1.151, 0.248]	0.154	[-0.565, 0.712]	
	NB × Residence length	-0.200	[-0.547, 0.298]	-0.134	[-0.482, 0.245]	0.071	[-0.446, 0.519]	
	NB × Population density	0.844	[-0.025, 1.638]	0.532	[-0.427, 1.416]	-0.055	[-0.647, 0.597]	
	NB × House type heterogeneity	0.054	[-0.658, 0.728]	0.111	[-0.651, 0.747]	-0.158	[-0.782, 0.426]	
	NB × Greenery	-0.101	[-0.407, 0.533]	-0.130	[-0.531, 0.278]	-0.058	[-0.537, 0.346]	
	NB × Walkability	0.163	[-0.641, 1.431]	0.319	[-0.677, 1.449]	0.013	[-1.023, 0.639]	
	NB × Amenities	-0.606	[-1.972, 0.109]	-0.746	[-2.122, -0.029]	-0.268	[-1.157, 0.416]	

Significance levels: *p<0.001 **p<0.01

Note: No superscript indicates statistically non-significant values.

support in any model (H6b, H7b, H8b, H9b, H10b, and H11b: Not supported).

4.5. CONCLUSION

This chapter presented the results of applying the developed computational framework to the case study of the city of Rotterdam. It provided evidence for the research hypotheses and answered SQ2: *How can the indicators of the mechanisms that drive balanced neighbourhoods to act more resiliently be operationalised?*; and SQ3: *What is the empirical relationship between the indicators that drive balanced neighbourhoods to act more resiliently?*

First, we observed that the conditions that define a balanced neighbourhood for the municipality of Rotterdam subsume a set of different possible distributions. The application of the KL divergence showed a considerable variance and susceptibility to outliers in the measurement of *Neighbourhood balance*. From all the possible configurations, roughly 2.1% fit the proposed PLS-SEM model. From the fit models, we identified three different clusters, and we selected the best fitting distribution in each to obtain further

insights.

The results of the PLS-SEM assessment to the three chosen models show that they successfully estimated the constructs and yielded an acceptable explanatory power, but low predictive accuracy. The indicator % of residents who say they are willing to care for neighbours or friends who need help (IS1) makes up the total variance of the latent construct Informal support. The emergent construct Social cohesion was reduced to the % of residents who say that local residents share each other's views (Models 1, and 2; SC3) or to the % of residents who say they feel at home with local residents (Model 3; SC5).

Table 4.7 summarises the findings for the formulated hypotheses. The results of PLS-SEM showed that *Neighbourhood balance* is associated to *Social cohesion* but not to *Informal support*. This indicates that the relationship is fully mediated by *Social cohesion*. Models 1, and 2 present a concave distribution for balance and a negative relationship between *Neighbourhood balance* and *Social cohesion*. Model 3 presents a convex distribution for balance and a non-significant relationship between *Neighbourhood balance* and *Social cohesion*. Other models with convex distribution shows a significant positive relationship between *Neighbourhood balance* and *Social cohesion*, in opposition to the results of the concave distributions. All models indicate there is a positive association between *Social cohesion* and *Informal support*.

In the spatial analysis of *Social cohesion* and *Informal support*, we have seen the presence of spatial clusters. These clusters show a clear geographical division between the North, South, and centre of Rotterdam. In general, the North has a higher level of *Social cohesion* and *Informal support* than the South. Clusters of low *Neighbourhood balance* and high *Neighbourhood balance* can be found throughout the city, with prominent clusters in the South and North, respectively. The centre shows low levels of *Social cohesion* and *Informal support* while having a high level of *Neighbourhood balance*. The results of the spatial regression indicate, that the results of the PLS-SEM model are biased given the presence of autoregression from the spatial distribution. The outcome of the Durbin spatial model shows that there is presence of spillover or contagion effect in *Social cohesion* and *Informal support* between neighbourhoods.

To overcome the supposition that the relationships between neighbourhood balance, social cohesion, and informal support are exactly the same for all neighbourhoods, moderation analysis was conducted from a list of previously identified relevant demographic and built environment factors. The results show that *Social cohesion* is negatively associated with the *Ethnic heterogeneity*. No moderation effects are significant between *Neighbourhood balance* and *Social cohesion*. On the other hand, we can also see that *Informal support* is negatively associated with the *House type heterogeneity*, only in Model 3. No moderation effects are significant between *Neighbourhood balance* and *Informal support*.

Table 4.7: Status of the formulated hypotheses for every Model.

Hypothesis	Model 1, and 2 status	Model 3 status
H1: Social cohesion positively affects Informal support	Supported	Supported
H2: Neighbourhood balance positively affects Social cohesion	Not supported	Not supported
H3: Neighbourhood balance positively affects Informal support	Not supported	Not supported
H4: The relationship between Neighbourhood balance and Informal support is mediated by Social cohesion	Supported	Not supported
H5a: Spatial spillover effects are significant on the relationship between Neighbourhood balance and Social cohesion	Supported	Supported
H5b: Spatial spillover effects are significant on the relationship between Neighbourhood balance and Informal support	Supported	Not supported
H6a: Ethnic heterogeneity moderates the relationship between Neighbourhood balance and Informal support	Not supported	Not supported
H6b: Ethnic heterogeneity moderates the relationship between Neighbourhood balance and Social cohesion	Not supported	Not supported
H7a: Residence length moderates the relationship between Neighbourhood balance and Informal support	Not supported	Not supported
H7b: Residence length moderates the relationship between Neighbourhood balance and Social cohesion	Not supported	Not supported
H8a: Population density moderates the relationship between Neighbourhood balance and Informal support	Not supported	Not supported
H8b: Population density moderates the relationship between Neighbourhood balance and Social cohesion	Not supported	Not supported
H9a: House type heterogeneity moderates the relationship between Neighbourhood balance and Informal support	Not supported	Not supported
H9b: House type heterogeneity moderates the relationship between Neighbourhood balance and Social cohesion	Not supported	Not supported
H10a: The amount of greenery moderates the relationship between Neighbourhood balance and Informal support	Not supported	Not supported
H10b: The amount of greenery moderates the relationship between Neighbourhood balance and Social cohesion	Not supported	Not supported
H11a: Walkability moderates the relationship between Neighbourhood balance and Informal support	Not supported	Not supported
H11b: Walkability moderates the relationship between Neighbourhood balance and Social cohesion	Not supported	Not supported
H12a: The number of amenities moderates the relationship between Neighbourhood balance and Informal support	Not supported	Not supported
H12b: The number of amenities moderates the relationship between Neighbourhood balance and Social cohesion	Not supported	Not supported

5

DISCUSSION

This chapter discusses the results presented in Chapter 4 embedded within the political controversy of balanced neighbourhood policies. The discussion starts with a section dedicated to the implications of the definition of a balanced neighbourhood from the municipality of Rotterdam. Second, we discuss the results obtained for the constructs based on social indicators. Finally, the relationships between the constructs in the model are examined, and we reflect upon balanced neighbourhood policies.¹

5.1. THE DICHOTOMY OF BALANCE

The first part of this section is dedicated to a discussion about the definition of a balanced neighbourhood by the municipality of Rotterdam. As explained in Section 2.4.2, the municipality of Rotterdam aims to demolish cheaper housing in favour of middleand higher-income groups. To measure the progress and set a goal for the housing programme, the municipality defined the set of conditions in Figure 2.9.

The definition of balance can result in counterintuitive conceptions of balance. As we saw in the exploration of the conditions for balance, the municipality defined a broad combination of housing WOZ-value stock distributions (see Figure 4.1) that results in a dispersed measurement of *Neighbourhood balance* (see Figure 4.2). This shows the discrepancy in the multiple possible combinations that the definition of the municipality allows.

To get a better grasp of the differences, Figure 5.1, shows an example for the neighbourhood of Zuiderpark and Zuidrand. This extreme example aims to show the implications of the broadness of the definition of balance by the municipality. As we can see,

¹Two interviews were conducted to two experts for consultation. Interviewee 1 was Dr.ir. André Ouwehand, researcher at the Delft University of Technology. Interviewee 2 was Dr. Wenda Doff, independent researcher in Rotterdam. The interviews have been cited as Ouwehand (2022, personal communication) and Doff (2022, personal communication), respectively. While the first interviewee is a resident in the North of Rotterdam, the second lives in the South. The variety in views contributes to avoiding biased results.

this neighbourhood is not in balance according to the definition of the municipality of Rotterdam: it has more than 60% of the houses in the Social segment and less than 40% of Middle, Higher, and Top combined. What is the house stock distribution that would achieve balance with the least change? That corresponds to Balance:60-9-31 as can be seen in Figure 5.1. This is the distribution from the solution space that yields the lowest *Neighbourhood balance*, i.e. that is closer to the actual housing stock. From a pragmatic policy perspective, this facilitates the implementation of a building program that aims towards this to fit the political discourse of a city in balance. The municipality would "only" have to transform around 20% of the Social housing in the neighbourhood to 5% Middle and 15% Higher or Top to check a box. What if, on the other hand, we took a look at the distribution that is the furthest away from the housing stock in the neighbourhood? Balance:1-98-1 would fit that description. This means that the municipality could get rid of all Social housing and the box would still be checked.

This is, of course, an extreme example, but it shows how the lack of specificity of the definition of balance of the municipality can result in counterintuitive conceptions of balance. The amount of Social housing could be reduced to the minimum, and we would have balance in the neighbourhood, as both the distributions of Balance:60-9-31 and Balance:1-98-1 are both within the conditions of the municipality of Rotterdam.

The dichotomy of balance limits quantification and can be misleading. Another concern with the definition of a balanced neighbourhood by the municipality is its dichotomous specification. If the neighbourhood complies with the conditions in Figure 2.9 it is in balance, otherwise it is not. The consequence in the approach of this study was that the quantitative analysis between balance and other concepts was made more difficult. Of course, a binary variable could be used to determine balance (0 = No balance, 1 =Balance). Nevertheless, this has two fundamental limitations. First, the use of ordinal variables in statistical modelling techniques (like the ones used in this study) is limited. An ordinal variable is in the end a qualitative measure that has been numerically coded; to be treated as quantitative by models, specific assumptions and techniques need to be taken (Schuberth et al., 2018). The second limitation is the interpretation of a binary ordinal variable. A neighbourhood is in balance, so what? How much in balance is it? On what side of the balance? Ordinal variables have the disadvantage that the information they provide is often so narrow that they can easily mislead conclusions. On the other hand, ordinal variables present two advantages which make them attractive, especially in the political discourse. First, they are easy to measure and categorise. By setting a number of straightforward conditions, the municipality can easily identify which neighbourhoods require more attention. The second advantage is the ease of communication. As variables get too complex and obscure, it is harder to explain them to laypeople. Within the social policy context, it is important to keep things as simple as possible (de Bruijn, 2019).

The definition of balance is under-specified in the expensive housing segments. Another point of attention in the definition of balance from the municipality is the aggregation of Higher and Top segments into the same segment. The goal of the municipality of Rotterdam is to build the following composition: 20% Social segment, 30% Middle segment, 30% Higher segment and 20% Top segment, in the long run. Nevertheless, this cannot be directly measured by their definition of balance. The Higher and Top segments get confounded and leave room for a hidden increase in polarisation, as the measurement for balance would capture an increase in 50% of Top segment in the same way. In the official documentation, they are open about the exact building programme for each segment, so transparency is guaranteed to a certain extent. However, an extension of this definition to other less monitored contexts could incite malpractices.



Figure 5.1: Example of the implication of the definition of balanced neighbourhood of the municipality of Rotterdam. Bars: Distribution of the housing stock of Zuiderpark en Zuidrand in 2019. Blue line: Possible distribution of 60% Social, 9% Middle, and 31% Higher+Top which would yield the best *Neighbourhood balance* for Zuiderpark en Zuidrand in 2019 with NB = -0.08 nats. Orange line: Possible distribution of 1% Social, 98% Middle, and 1% Higher+Top which would yield the worst *Neighbourhood balance* for Zuiderpark en Zuidrand in 2019 with NB = -3.76 nats. For the municipality of Rotterdam, both cases are considered as balanced. Remember that nat is the information loss from one distribution to the other that the KL divergence calculates. To make sense of these results, the measure of Balance:60-9-31 is 1.13 standard deviations above the overall mean. This shows the lack of specificity of the definition provided by the municipality.

Only a few distributions within the definition of balance can explain the alleged effects of the municipality. Structural equation modelling provides us with the ability to test theories between complex social constructs as sets of linear equations (Hair et al., 2016). To determine how likely it is that that our conceptual model matches the real world (or better put, the data collected from the real world), it is important to assess how well the relationships coincide with empirical data. In other words, we need to test the model's goodness-of-fit. Our results showed that from all the possible integer configurations that satisfy the conditions of balance, roughly 2.1% fit the model. Why did only a few fit? To understand this in a more sensible way, let's suppose you have to make a drawing of a hat. To know if your drawing is good, you can compare the distances between pairs of points on the drawing with the corresponding measured distances on a

real hat. The closer the match, the more certain you are that your drawing is faithful to reality. The same can be applied in SEM by comparing the strength of relationships as implied by the model and the empirically observed strength of the relationships. This was quantified by the SRMR, a measure that indicates how strongly the empirical correlation matrix differs from the model-implied correlation matrix (Henseler, 2020). In that sense, the integer configurations that fit our model were the ones which could explain the relationships between Neighbourhood balance, Social cohesion, and Informal support better. However, we have to pay attention when drawing the hat, as we may really be drawing a boa constrictor digesting an elephant! (see Figure 5.2). Our model fit not only depends on the inner workings of PLS-SEM, but also on the different distributions to obtain *Neighbourhood balance*. Therefore, in every model, we kept changing the target balance distribution the same way we kept changing what the boa is digesting. This can contribute to explain why a low number of the different integer configurations that satisfy the conditions of balance fit. While PLS-SEM works with linear equations, the KL divergence to calculate Neighbourhood balance from the distributions set by the municipality is a logarithmic transformation.

With the understanding of the fit of the different models, we can give a meaning to the integer configurations that satisfy the conditions of balance that did fit the model. The balance distributions that we have found in this study are the ones which can explain the relationships in the model; all the others, did not manage to capture the relationship. A first interpretation of this result can mean that the model isn't complex enough to match the available data. As we have seen in the analysis of exogenous factors, *Ethnic heterogeneity* and *House type heterogeneity* play a significant role in the inner model. It could possibly be that by incorporating those (and other) factors in the model, it would capture reality with more detail and more estimations would fit. In addition, variables may also be affected by systematic measurement error or assumptions like uncorrelated measurement errors may not hold.

This interpretation opens a reflection of possible malpractices in using statistics for policymaking. A model's goodness-of-fit can be increased by increasing its complexity, i.e. if we add more parameters, the model will be able to explain the data with more accuracy. However, if we overfit the model, it will **only** be able to explain the data to which it is fitted and will obscure the theoretical foundations. Decision-makers run the risk of accepting the model with the best goodness-of-fit, but with ineligible relationships. This reflection supports the logic of using a deductive approach rather than a data-driven approach (Schwab & Held, 2020; Shih & Chai, 2016). In the deductive approach, the theory is first developed and then checked against available data, so the theory is central and clear. Instead, in the data-driven approach, the theory is reasoned from the data, so complexity (and, therefore, goodness-of-fit) can be increased until the theory fits the modeller's interests.

Another interpretation of the few fitting distributions is that the study has uncovered which are the balance distributions for the city that can actually show the alleged effects of balance. This would mean that the distributions of the models that did not fit do not provide empirical evidence of the policy discourse of the municipality of Rotterdam. In this sense, if the municipality wants to keep developing the housing programme to achieve balance, the found distributions would be more specific objectives which actu-



Figure 5.2: Analogy of model fit. In PLS-SEM we estimate the model fit by comparing the strength of relationships as implied by the model (hat drawing) and the empirically observed strength of the relationships (real hat). Nevertheless, in our model we also varied the construct of *Neighbourhood balance*, showing that we were not just drawing a hat, but a boa digesting different things. Adapted from: de Saint-Exupéry (1943).

ally show the intended outcomes.

Each neighbourhood should aim at a balance according to its characteristics, but the definition of balance is the same for all. While in this study the same target WOZ-value distribution was used for every neighbourhood in each model run, the Woonvisie programme aims to intervene on a city-wide and neighbourhood level where necessary and in a targeted manner (Municipality of Rotterdam, 2019c). This means that balance should not be measured as a unique city-wide distribution, but that each neighbourhood could have a different goal that achieves balance in relation to the population in the neighbourhood. Nonetheless, the definition of balance from the municipality of Rotterdam inherently posits the same ideal for all neighbourhoods. The indicator can identify which neighbourhood requires attention, but it fails to capture the distinctive features of every community. Therefore, the results of this study can be of use to the urban planners of the municipality. The computational framework could be adapted to calculate all the combinations of every balance distribution for each neighbourhood to select which balance explains better the relationships in each neighbourhood. This would pro-

vide specific insights for the building programme for each neighbourhood, as well as a more concrete measure for each. This suggestion was not performed in this research and should be optimised due to the time constraints and computational power limitations. If we were to calculate all possible combinations, that would result in $\binom{k}{n} = 4 \times 10^{144}$ estimations with k = 3, 164 integer configurations that satisfy the conditions of balance and n = 70 neighbourhoods. At an average time of 13 seconds per estimation,² we can clearly see the limitation. And this is just considering integer combinations...

Balance as physical indicator for social resilience carries normative and political load.

The assumption of balance is grounded in the presumed superiority of the newcomers that will transform the neighbourhood. As such, "Balance" is associated with a resilient and inclusive city, and makes neighbourhoods not in balance undesirable. The concept in itself is politically powerful: If you are against balance, you favour vulnerability and segregation. The policy has an indirect political implication for the citizens dwelling in those designated areas.

Instead of being a victim of the political discourse, the detractors of the Woonvisie in the movement Recht op de Stad have reframed the paradigm (Right to the City, 2022). If you are in favour of balance you support the cold technocratic city, if you are against it, you champion a city for the people. In their frame, a resilient and cohesive city requires to go beyond an indicator and embrace the social characteristics of a neighbourhood.

The measurement of a physical indicator like the WOZ-value fails to recognise the importance of social factors, while it points towards social resilience capabilities of a community. This is partially due to how challenging it is to assess traits like the ability for community members to develop resilience via their agency (Copeland et al., 2020). Social resilience indicators like informal support can factor in that agency, but as we will see in the next section, they also carry their own limitations.

5.2. Social indicators for resilience

Before jumping into the discussion of the relationship between the constructs, this section first elaborates on the obtained results for the measurement of *Social cohesion* and *Informal support* and then critically assesses their implications.

Measurement theory could not identify the common factors of social cohesion. To begin with, it is relevant to mention how approaching the conceptualisation of *Social cohesion* using measurement theory failed to provide positive results. Appendix B.2 elaborates on the procedure and the possible reasons, but we concluded that no clear factors could be extracted to treat *Social cohesion* as a latent variable. Thereupon, the construct was conceptualised from synthesis theory using PLS-SEM.

The (lack of) reliability of social cohesion as a single-indicator construct. In PLS-SEM, Models 1, and 2 reduced *Social cohesion* to *SC3* (% of residents who say that local residents share each other's views) when the objective balance distribution was concave. On the other hand, Model 3 estimated that it can be reduced to *SC5* (% of residents who

²Estimations were conducted on a quad-core Intel Core i7-6700HQ @ 2.60GHz with 16GB RAM.

say they feel at home with local residents) when the objective balance distribution was convex. The other indicators were either non-significant or highly collinear with the final measures. The reduction to single-indicator measurements for Social cohesion could stem from the fact that aggregate data was used. The use of aggregate data has been widely acknowledged to result in ecological fallacy, where relationships that are concluded as true at a level of aggregation, must be true at another level (G. King, 2013). In this case, the fact that aggregate data of the Wijkprofiel survey suggest that Social cohesion can be reduced to a single-indicator, does not mean that at an individual level response Social cohesion would be characterised in the same way. Assuming that our measure of *Social cohesion* is not a victim of the ecological fallacy, the results may cast doubts on the utility of the scale employed by the Dutch government, or on the data collection process from the Wijkprofiel. Previous research emphasises the multidimensional nature of social cohesion. Tolsma et al. (2009) even states that 'it is imperative that social cohesion is not reduced to one single indicator, let alone that different dimensions are simply aggregated. The overarching concept of social cohesion is not easily reduced to one or two indicators.' (p. 303). Taking this into account, and knowing that other research from Benitez-Avila et al. (2022) also showed that the scale from the Dutch government could be reduced to two indicators using microdata, put into question the validity of the indicators.

Social cohesion in urban settings is mostly based on forms of personal relations governed by relatedness. *Informal support* was defined as the % *of residents who say they are willing to care for neighbours or friends who need help (IS1). Informal support* is therefore characterised by strong relationship ties with friends and weak relationship ties with neighbours, without invisible or non-existent ties. Appendix C explored other conceptualisations of *Informal support* to obtain different results.

First, we assessed whether balanced neighbourhoods foster *Informal support* conceptualised as a latent construct from the % of residents who say they are willing to care for: (i) relatives, (ii) neighbours or friends, and (iii) others in the area. Results showed that a common factor could only be found between the two first indicators, thus reducing *Informal support* to the strong and weak relationship ties. This is aligned with the research from CBS on help and support, in which they state that the order of family, friends, and finally acquaintances has always been maintained because of the assumption that the relationship with family is stronger than the relationship with friends and the relationship with friends is then stronger than that with acquaintances. The resources are expected to be most accessible through family members, followed by those from friends. Because of the less close ties with acquaintances, resources are the least accessible via this group (CBS, 2012).

To further explore the relationship of balanced neighbourhoods with resilient actions to others, we then conceptualised *Informal support* as a latent construct from the % of residents who say they are willing to care for: (i) neighbours or friends, and (ii) others in the area who need help. Again, a common factor could not be identified, due to the low collinearity between indicators. Following the CBS study on help and support, the tight relationship of Dutch people with strong and weak relationship ties inhibits the relationship with others and thus results in a lower willingness to help them (CBS, 2012).

As a final exploration, *Informal support* was conceptualised as only consisting of % *of residents who say they are willing to care for others in the area who need help.* We found that none of the estimations yielded an acceptable goodness-of-fit. This suggests that invisible and non-existent ties are not a guarantee of resilient action, and that social cohesion in urban settings is mostly based on forms of personal relations governed by relatedness (Felder, 2020).

Strong and weak ties constitute informal support, but stated choice might not represent revealed preference. The expert interviews concurred in a caveat of the measurement of informal support. The data provided by the Wijkprofiel measured the willingness to help, i.e. the stated choice of the respondent. Nevertheless, during the COVID-19 pandemic, the revealed support was actually higher than the stated. In the vocabulary of resilience, the measurement of potential action underestimated the measurement of the effective action (Doff, 2017). Veld Academie (2021) shows how in the neighbourhoods of the BoTU programme (Bospolder/Tussendijken), low-income ethnically mixed areas exhibited more helping and caring than the previously stated. On one hand, this shows that it is after the stressor (COVID-19 in this case) when the actual level of informal support can be measured so if only potential action is considered, obstacles to actual action are not taken into account.

Additionally, both interviewees also pointed out that from their experience, qualitative research does not show the same as quantitative, especially in the South. Interviewing residents shows that the level of informal support is higher than the recorded because people are not interested in filling out questionnaires. This is also observed in the research from the CBS on help and support, which also indicated that non-Western immigrants have access to resources somewhat less often through their social network than native Dutch and Western immigrants. Our results are aligned as they show how in the South of Rotterdam, where there is a higher percentage of non-Dutch residents, the willingness to help is generally lower. This outcome is aligned with the idea in resilience literature that that social resilience must see social positions (Copeland et al., 2020).

Social indicators towards a technocratic city. The results of the PLS-SEM models showed that the indicators used to measure *Social cohesion* and *Informal support* have a low explanatory power. Even though this is not a problem for our confirmatory deductive approach, it limits the possibility to obtain consistent estimates of relationships between constructs (Henseler, 2020). Consequently, quantitative research that aims to employ these indicators is limited in essence.

Taking a closer look at *Social cohesion*, significant indicators portrayed the feeling of belonging and the mutual tolerance between groups. Neighbourhoods in the South of Rotterdam showed low levels of *Social cohesion*, while they are acknowledged to have more social contacts (Doff, 2022, personal communication). We can also see that measuring *Social cohesion* as the % of residents who say that local residents share each other's views yields less negative outcomes in the South than measuring it as the % of residents who say they feel at home with local residents. This shows a difference in the consequences of how to measure social cohesion. In the South, people might not feel at home in the neighbourhood, but they feel more connected to the people than to the place.

Second, *Informal support* to friends and family was used as a proxy for resilient actions based on the literature search. Other formulations of social resilience could have been formulated to check the consistency of results. For example, using indicators that reflect effective resilient action instead of potential action. On the other hand, the willingness to help is liable to conflict of multiple interpretations. People have different beliefs, values, and understanding of what is helping someone. As such, policies that rely on the indicator are subjectively loaded given its normative quality.

Indicators for social cohesion and social resilience hardly capture the actions and perceptions within the community. Furthermore, the selection, measurement, and use of indicators are accompanied by a normative leverage (Copeland et al., 2020). In this study, we tried to overcome this issue by contextualising the social indicators within the spatial location and other exogenous factors. Even so, social indicators are subjective to interpretations of individual values, their change over time, and the comparison across places. Therefore, the establishment of predefined indicators, together with the increasing norm of quantitative-based political decision-making due to the vast amount of data available in recent years, show an approach towards the technocratic city.

5.3. BALANCED NEIGHBOURHOODS AS AN INSTRUMENT TO ACHIEVE SOCIAL RESILIENCE BY MEANS OF SOCIAL COHESION

The effects of balance on the social cohesion and the resilience of a neighbourhood are widely disputed. While the policy discourse and ample qualitative literature indicate that balanced neighbourhoods improve neighbourhood effects, most quantitative studies show that these claims are unfounded. This research aimed to throw light on the debate by providing additional empirical evidence.

Balanced neighbourhoods are not in line with the discourse from the municipality. Results show that they are associated with a lower social cohesion and satisfy the homophily principle. Models 1, and 2 suggested a negative relationship between the level of balance and social cohesion, while Model 3 suggested the opposite.³ Models 1, and 2 are directly aligned with the homophily principle of Putnam (2000) by which people share the same views when surrounded by those who resemble them, thus social cohesion in mixed neighbourhoods is affected by one's aversion to the alien (Méreiné-Berki et al., 2021). Instead, Model 3 shows a positive relationship between the level of balance and the feeling of being at home with local residents, **apparently** in correspondence with contact hypothesis, by which ties between heterogeneous groups lead to overcoming social differences.

Both interviewees stated that from their field experience, the homophily principle is predominant in Rotterdam. Ouwehand (2022, personal communication) indicated that restructuring neighbourhoods does not contribute to better and integrated relationships between different groups in the neighbourhood. From the beginning of the housing restructuring policy in the Netherlands, urban planners have known that if part

 $^{^{3}}$ Note that the relationship in Model 3 was non-significant, however, Figure 4.5 shows that other results from the same cluster did have a significant relationship. Therefore, we discuss the implication of such relationship.

of a neighbourhood is demolished and expensive units are built, the high-income newcomers do not mix much with the previous residents. The newcomers visit different amenities, sporting facilities, and schools, thus not contributing to the social cohesion in the neighbourhood. This perspective was corroborated by Doff (2022, personal communication), who stated that demolishing housing destroys social networks and social cohesion. Our cross-sectional study cannot capture these effects that can only be observed with a longitudinal analysis that gathers data on the evolution of the neighbourhood after the incorporation of newcomers.

What about Model 3 in which *Neighbourhood balance* and *Social cohesion* were positively correlated? As mentioned earlier, this result **apparently** contradicts the other models and suggests that a balanced neighbourhood leads to higher social cohesion. This can be explained from the shape of the balance distribution of Model 3. While the other models had a concave shape, Model 3 presented a convex distribution. This means that Models 1, and 2 aimed towards a higher percentage of Low or Middle than Top, while balance in Model 3 minimises the number of Middle housing. This distribution for a balanced neighbourhood — which is considered balanced according to the municipality —, actually aims for a stronger segregation by polarising the housing value. The outcome of the model can then be interpreted as neighbourhoods which are more polarised show a stronger relationship with *Social cohesion* (see Figure 5.3). We can therefore see how Model 3, even though it would seem that it is in line with the contact hypothesis, it actually follows the same logic as the other models. These results are of extreme relevance, as they show how the ambiguity of the definition of balance by the municipality could actually mislead the interpretation of results.

This study is innovative as it measures balance from the house market value level, and thus cannot be directly compared to results from previous literature. Nevertheless, balance in terms of house value is an intermediate for income, as acknowledged in the Woonvisie (Municipality of Rotterdam, 2016b). Therefore, this perspective can also be identified in previous empirical research that relates income mix to social cohesion. For example, Coffé and Geys (2006) did not find a correlation between income mix and social capital in Flemish neighbourhoods and even indicates that ethnic balance had a negative effect on social capital. Also, Wang and Kemeny (2022) found that mixed income neighbourhoods in China were associated with lower social cohesion.

Fostering housing balance strengthens the average real estate value and socioeconomic status of a neighbourhood, but it does not necessarily improve the situation of the worst off that live or lived in that neighbourhood. As mentioned by Ouwehand (2022, personal communication), the indicators of a neighbourhood improve when attracting higher income families. The amount of residents that are associated with high levels of vulnerability like migrants, the young, the old, and the poor are reduced (D. King & Mac-Gregor, 2000), but they do not obtain the benefits of living with newcomer high income residents. These conclusions can be seen in studies like Tolsma et al. (2009) who suggest social cohesion in Dutch neighbourhoods can be improved by increasing the average income of the neighbourhood. This does not necessarily imply that the social cohesion between income groups improves, but that the measured social cohesion is higher because of the introduction of a new group in a neighbourhood. We can then see how quantitative indicators of social phenomena cannot be fully tested with quantitative re-



Figure 5.3: Concave and convex distributions of balance follow the homophily principle. Concave WOZ-value distributions reduce the polarisation between income groups and have a negative path coefficient between *Neighbourhood balance* and *Social cohesion*. This shows how aiming for a decrease in polarisation decreases social cohesion, thus follows the homophily principle. Convex WOZ-value distributions increase the polarisation between income groups and have positive path coefficients between *Neighbourhood balance* and *Social cohesion*. This shows how aiming for an increase in polarisation increases social cohesion, thus follows the homophily principle.

search designs.

Social cohesion fully mediates the relationship between balance and informal sup-Moving on to the interpretation of the relationship between Social cohesion and port. Informal support, we can see that our results are in line with the expected outcomes. In more cohesive neighbourhoods, residents are more willing to help friends, and neighbours in moments of hardship. Both interviewees also agreed with this outcome. Taking this into account, the relationship between Neighbourhood balance and Informal support is fully mediated by Social cohesion. The inclusion of Social cohesion as a mediator variable moves the focus away from whether a physical indicator like the WOZ-value affects the resilient actions and instead toward how the variables are related. This result is interesting as it shows that changing the level of balance of the neighbourhood will not have any direct effects in the willingness to help friends and neighbours. Potential resilience can be increased by other means that increase social cohesion other than the WOZ-value distribution. For example, Doff (2022, personal communication) suggested that within the BoTu programme, positive results in social cohesion and community support have been observed by the development of neighbourhood organisations which provide a space for bonding and bridging (Veld Academie, 2021).

Overall, the results from mediation analysis indicate that balanced neighbourhoods are associated with lower levels of *Social cohesion*, and therefore of *Informal support*. A physical mix in the housing stock value discourages resilient action.⁴ These results are in line with previous research in Rotterdam using the Wijkprofiel survey results. Custers (2021) found that mixed neighbourhoods do not foment bridging between social groups, and that even if there exist mixed social networks in a neighbourhood, these are not strong enough to understand the needs of others and share necessary resources in the event of need.

Balanced neighbourhoods have been criticised for reducing the amount of affordable housing in cities and consequently displacing low-income families (UN-OHCHR, 2021). The negative effects are allegedly countered by the benefits that balanced neighbourhoods bring on the social cohesion and the resilience of Rotterdam. Our results show, however, that balanced neighbourhoods are less cohesive and are less willing to help their friends and neighbours. Therefore, the legitimacy of the discourse of the municipality of Rotterdam is contested by our results. Rotterdam wants to create a city for everyone, however, together with the claims of the UN and the movement Right to the City, our results indicate that the housing policies could serve as a gentrification instrument that works towards the polarisation in the city by which the social class is gradually replaced by high-income newcomers to increase the average real state value of neighbourhoods (Custers, 2021; Versluis, 2017).

The use of Wijkprofiel survey data can lead to biased results if spatial effects are not taken into consideration. The spatial analysis helped unravel other relevant implications of the study. First, in the analysis each neighbourhood contributes equally to understanding the relationship between the concepts, however, this can result in biased results because this is not proportional to the amount of people that live in the city. For example, Noord Kethel and Schiebroek are industrial neighbourhoods with few residents compared to other locations. Nevertheless, these neighbourhoods are included in the Wijkprofiel and used in the evaluation of urban policies. We saw how these neighbourhoods obtained high scores for *Social cohesion* and *Informal support*. Similarly, the outlying characteristics of the centre of the city also call for caution. The city presents a high level of balance due to the recent building of middle, higher, and top level units (Custers, 2021). In line with the results, *Social cohesion* levels are low in the centre. This is not necessarily attributed to the level of balance, but is specially influenced by the characteristics of the groups that live in the city centre, so results are biased by Type I error. While people in residential areas are usually more attached to the neighbourhood,

⁴The theory of the model is based on the grounds that geographically connected people become affected by the social mix in the area to promote solidarity. As such, it uses the willingness of friends and neighbours to characterise resilient action. Appendix C shows similar results when conceptualising *Informal support* as the common factor of the % of residents who say they are willing to care for relatives who need help (IS3), and the % of residents who say they are for neighbours or friends who need help (IS1). When incorporating relatives, the geographical boundaries are broken and arguably the change at a physical level might not have an effect in the resilient action. We see, however, that the path coefficients have higher absolute values when incorporating strong relationship ties from relatives. The results show an intrinsic capacity of people to be willing to help relatives if the balance is low. This preliminary results motivate future research on the effect of changes at a physical level that could have an impact on the behaviour of the private deterritorialized domain.

residents in the centre are more cosmopolitan and footloose. Therefore, it is relevant to acknowledge the influence of other neighbourhood characteristics in the results and how can that affect the development of policies.

Second, spatial analysis shows the presence of spillover or contagion effects. We observed that the relationship between *Neighbourhood balance* and *Social cohesion* is significantly dependent on the average value of *Social cohesion* of surrounding neighbourhoods. Similarly, the dependence of *Informal support* on *Neighbourhood balance* and *Social cohesion* is also affected by the average *Informal support* value of surrounding neighbourhoods. This confirms that *Social cohesion* and *Informal support* are not administratively bounded, so the interaction between residents of adjacent neighbourhoods can influence the perception and actions of others.⁵ As a result, the outcomes of the PLS-SEM model are again biased by Type I error. These results are very context dependent, however, too little research on social cohesion has taken this into consideration (Uitermark et al., 2017). The spatial capabilities in SEM are still very limited, but this research shows the relevance of incorporating spatial dependence to avoid biased results.

Finally, spatial autocorrelation is relevant in this case study due to the geography of Rotterdam. Provided that the city is divided by the Nieuwe Maas river, the difference in interaction between and within river beds is expected to considerably impact social cohesion. Furthermore, the larger concentration of reported problems and supply of lowcost housing is greatest in Rotterdam South (Municipality of Rotterdam, 2016b, 2019b). For that reason, the NPRZ and Woonvisie demolition plans are primarily concentrated on the South side, despite the fact that this is not expressly stated (Versluis, 2017). The main findings reported spatial effects without taking into consideration the river effects, i.e. by assuming that neighbourhoods face to face across the river were direct neighbours. Appendix E shows the same results by considering that neighbourhoods on both sides of the river are not neighbouring and there is no spillover between them. The LISA results show the clear presence of polarisation between both sides of the river in Rotterdam. These results are relevant because if reality corresponds to a strong separation caused by the river (i) improving areas in the centre or North of the city will not have a beneficial spillover effect on the South, and (ii) negative neighbourhood effects in the South do not impact other areas, reinforcing the marginalisation of the South. Future research should deepen into the structure and strength of the interactions across the river.

Ethnic heterogeneity, and house type heterogeneity are relevant urban factors that require additional examination. The moderation analysis showed the absence of moderation effects from the identified exogenous factors. However, we found that *Ethnic heterogeneity* and *House type heterogeneity* are the two factors which directly affect the constructs in our model.

The effects of *Ethnic mix* are very disputed in the literature, with proponents of both the contact hypothesis and the homophily principle. In our results, we can see that a

⁵Additionally, we observed no spillover effects of *Informal support* when studying the common factor that included the willingness to help relatives. We argue that this is because the construct is forged by strong ties, therefore informal support is a characteristic that stays within the closest network and does not spill over it.

high *Ethnic heterogeneity* is associated with lower *Social cohesion*, thus agreeing with the homophily principle. Although *Ethnic heterogeneity* was treated as an exogenous factor as it is not the same as balance in terms of income, they could be confounded as wealthiest neighbourhoods correspond to those of Dutch majority. *House type heterogeneity* was also significant in Model 3, showing a negative direct relationship with *Informal support*. This variable can be interpreted in a very similar way to ethnic or income mix. In neighbourhoods with a high mix of housing types, spatial segregation occurs and therefore follows the homophily principle. Consequently, we argue that future research should elaborate on the theory on which the model is grounded and create coherence to the empirical relationships identified with *Ethnic heterogeneity* and *House type heterogeneity*.

Both direct relationships and moderation effects were found as non-significant for the other built environment factors. For instance, regarding Greenery, previous research using the Wijkprofiel data showed how a green living environment offers opportunities for recreation and relaxation and contributes to a sustainable neighbourhood, indirectly contributing to a more resilient neighbourhood (Veld Academie, 2021). The fact that our results were non-significant in contrast to the study from Veld Academie (2021) could be explained by the difference in the indicators. Veld Academie (2021) used personal perceptions like 'Sufficient use of green space' to understand its relationship to social cohesion, while we used a physical variable that measures the % of green areas in the neighbourhood. The use of subjective indicators carries the same normative weight previously explained. Survey respondents answer from their perception of what is "sufficient" for a green space. In addition, this relates to the individual characteristics such as intentions and preferences that regulate the perception and experience of urban elements (Kim & Kaplan, 2004; Rogers & Sukolratanametee, 2009; Wood et al., 2010) (see Figure 2.5). As such, not only the % of greenery or % of walkable area are important, but also their urban design and how residents experience it. Appendix F qualitatively explores urban design elements of the North and South of Rotterdam which could have an influence in the social cohesion of the residents. It does not intend to be a rigorous qualitative analysis, but an opening door to neighbourhood specific research to uncover the underlying consequences of these results.

Finally, the interviewees provided two points of attention to the results of the exogenous factors. Ouwehand (2022, personal communication) expected the *Length of residence* to be a significant factor. Previous studies on the liveability in Rotterdam pointed out that liveability is strongly affected by this variable. It is possible that the relationship with liveability is not extended to social cohesion. Studies like Rogers and Sukolratanametee (2009) did not find significant effects between sense of community and length of residence. Further research should study these relationships more in depth. Next, Doff (2022, personal communication) also pointed out that *Amenities* had been observed as significant in previous research in Rotterdam by Veld Academie (2021). After examination of their study, we saw that they define amenities as cultural facilities and religious centres, differing from our conceptualisation of cafés, restaurants, community centres, bars, and pubs. This semantic confusion calls attention to additional issues that arise in the utilisation of resilience indicators. Using the same label for different indicators can be a result of lack of semantic accuracy, i.e. the conformity to its real value, or a lack of mapping consistency, i.e. the uniformity in the keys of values (Y. Huang, 2013). The former puts into question the essence of what are considered amenities, again showing the normative quality of indicator practises. On the other hand, the second would suggest an error in the key used to define the values considered as "amenities". In any case, this shows potential misleading interpretations of research studies.

6

CONCLUSION

In this final chapter, we present the conclusions that can be drawn from this study. To this end, we first discuss the findings of this study that give an answer to each research sub-question. Next, we translated the insights gained from the study into policy recommendations for municipal governments. Finally, we explicitly state the limitations of the study coupled with future work that could address these limitations. This chapter thereby answers the main research question *Does neighbourhood balance increase resilient action of neighbourhood residents*?

6.1. RESEARCH QUESTIONS

Taking into account the core concepts and gaps detected in the literature, the research was formulated to answer the following sub-questions:

SQ1: What are the underlying mechanisms that drive balanced neighbourhoods to act more resiliently?

Literature suggests that social factors govern the resilience of urban areas. Communities in which the residents work together and have common goals have a stronger willingness to cooperate. Scholars generally agree that resilience can only be present if the members in a community are capable of collective action to collect enough resources and have the capacity to mobilise them in the event of a disaster. Given that resilience is not directly measurable, we argue that informal support is a suitable proxy to measure the potential resilience of a neighbourhood, as the end mechanism through which social resilience is achieved is mutual aid.

Mutual aid depends on the level of social cohesion of the community. In cohesive societies, local communities protect residents against threats, care of others during hardships, and ultimately promote community resilience. This is because social cohesion is an involuntary product of the form and quality of shared values, trust, and the relationships among the individuals of a society. The municipality of Rotterdam also maintains this claim, and regards social cohesion as a challenge to address to reach the goal of a resilient city. Balanced neighbourhoods are considered a policy instrument to achieve a more resilient city by means of social mix. Balanced neighbourhoods mix the population composition in a neighbourhood to allegedly improve the social cohesion of its residents. Large concentrations of segregated groups in specific areas are believed to reinforce and perpetuate poverty and exclusion, and consequently reduce social cohesion. Based on the contact hypothesis, in socially mixed areas, different groups meet and interact with each other, and eventually this reduces hostility toward out-groups. As a result, residents of balanced areas are more cohesive and thus more willing to take resilient actions for subsistence in the event of a catastrophe or to tackle long-lasting problems.

Putting all elements together, the resilience mechanism that we studied is that of social cohesion mediating the relationship between neighbourhood balance and informal support. Social cohesion acts as the mechanism for resilient actions triggered by the balance in a neighbourhood, and mediation analysis explains how the change in balance and resident support occurs by means of social cohesion.

In addition, social cohesion and informal support have been associated to characteristics of the built environment and the demographics. This is based on the idea that urban elements can promote or deter social interactions, and thus are aspects of what can be considered a resilient neighbourhood. As such, the moderating effects of ethnic heterogeneity, residence length, population density, house type heterogeneity, amount of greenery, walkability, and the number of amenities were determined to overcome the supposition that the relationships between neighbourhood balance, social cohesion, and informal support are exactly the same for all neighbourhoods and to identify instances in which the moderating variables change the strength or direction of the relationship between the concepts.

SQ2: How can the indicators of the mechanisms that drive balanced neighbourhoods to act more resiliently be operationalised?

From the underlying mechanism, we distinguish three main indicators: the level of balance, the level of social cohesion, and the resilient action.

First, we have seen that, in the case of Rotterdam, balanced neighbourhoods are defined from a set of conditions for the amount of houses in different house price segments. Only specific combinations satisfy the definition of balance. A first analysis pointed out the under-specification of the definition of balance from the municipality of Rotterdam. The possible combinations of housing price segments is broad, and can result in combinations which are counterintuitive to the idea of balance, such as having all middle-class housing or none at all. As a result, the indicator can be misleading. The question arises as to how balanced is a neighbourhood with respect to the possible distributions. The negative of the Kullback-Leibler divergence has been found to be an adequate tool to measure the level of balance of a neighbourhood. Given a target house price distribution, it can determine how far off a neighbourhood is to achieve it. It is a versatile technique which is transferable between contexts, as the target distribution can be adapted accordingly. This provides a two-fold advantage. First, the level of balance can be
adjusted to the evolution of housing prices in the same area. Second, it can be adjusted to different aggregation levels or even different areas, as it does not depend on intrinsic characteristics of the measured distributions.

Second, the abstraction and the cross-disciplinary use of social cohesion between social sciences and policymaking have made the measurement of the concept challenging. However, the interest in monitoring the social cohesion of communities has driven scholars to the development of several approaches. In this study, we have discussed and measured social cohesion from the grounds of measurement theory and from the grounds of synthesis theory. Taking the measurement theory approach, we conceptualised social cohesion as a latent construct that causes a number of observed variables and their relationships. Social cohesion as a latent construct emanates from and reveals the structure of empirical data. The results of such approach suggested that no clear factors could be extracted, thus social cohesion could not be treated as latent from the available data. Next, taking the synthesis approach, social cohesion was treated as en emergent variable which acts as a whole instead of being an assembly of parts. As emergent constructs rely as far as possible on the revealed principles of measurement theory, social cohesion was conceptualised using an existing set of five indicators used by Dutch Office of Social and Cultural Planning. The results reduced the set of indicators to a single indicator being the % of residents who say they feel at home with local residents or % of residents who say that local residents share each other's view. Literature suggests that the overarching concept of social cohesion is not easily reduced to one or two indicators, which raised doubts on the utility and validity of the measurement scale.

Finally, informal support is based on the grounds that geographically connected people become affected by the social mix in the area to promote solidary actions. As such, it was conceptualised as a latent construct of the % *of residents who say they are willing to care for neighbours or friends who need help.* The indicator represented informal support from strong ties with friends and weak ties with neighbours. This indicator can measure potential resilience action, however, actual action is not taken into account and the revealed support can differ from what residents state.

SQ3: What is the empirical relationship between the indicators that drive balanced neighbourhoods to act more resiliently?

To begin with, only 2.1% of the possible balance distributions yielded an acceptable goodness-of-fit of our model. This could be indicative that the model needs to be reevaluated, but, in contrast, it could otherwise mean that that social cohesion and informal support cannot be explained by the balance in a neighbourhood and that the policy should be reevaluated. Under this second interpretation, the study has uncovered which are the balance distributions for the city that can actually show the alleged effects of balance.

The possible distributions of a balanced neighbourhood can, on one hand, aim for a polarised neighbourhood with houses on both ends of the low- and highpriced segments, or, on the other hand, gap the bridge and increase the amount of middle-priced housing. The results of the study suggest that balanced neighbourhoods which foment a reduction in polarisation are negatively associated with social cohesion, and that balanced neighbourhoods which foment polarisation are positively associated with social cohesion. This indicates that our results are in line with Putnam's homophily principle, i.e. 'birds of a feather flock together'. This outcome is opposite to the policy discourse of governments in favour of balanced neighbourhoods, including the municipality of Rotterdam, that mixed neighbourhoods foster social cohesion.

Next, the results of the study suggest that the association of social cohesion to informal support is positive. This outcome agrees with the existing literature which points out that socially cohesive communities are more resilient given their willingness to protect residents against threats and care of others during hardships. In this sense, this result is in line with Resilience challenge 1 of the municipality 'Social cohesion and education', by which the resilience of the city can be positively influenced by the social cohesion. Taking this into account, the results of the study suggest that balanced neighbourhoods are not directly associated with informal support, but indirectly related, fully mediated by social cohesion. Our results therefore indicate that mixed neighbourhoods are less cohesive and less willing to help each other. This outcome is in opposition with the policy discourse of the municipality of Rotterdam, that state that a city in balance is a more resilient city.

To increase the explanatory power, the model outcomes were followed by an exploratory spatial analysis. The inspection identified clusters of social cohesion and informal support within the city, visibly geographically divided between North, South, and city centre. This pointed out the relevance of tailored neighbourhood analyses to come up with strong inference. The history, development, and composition of each neighbourhood should be considered. In addition, another potential source of the clusters is the presence of spillover or contagion effects, as identified by spatial regression. This confirms that social cohesion and informal support are not administratively bounded, so the interaction between residents of adjacent neighbourhoods can influence the perception and actions of others.

Finally, the moderating effect of exogenous factors related to the demographics and the built environment of the neighbourhoods was assessed. None of the factors showed moderating effects. However, ethnic heterogeneity and house type heterogeneity were negatively associated to social cohesion and informal support, respectively. Future research should elaborate on the theory on which the model is grounded and create coherence to the empirical relationships identified if actionable policy wants to be developed and implemented.

6.2. POLICY ADVICE

The confirmatory analysis conducted in this thesis represents a first step in understanding the mechanism that drives resilient action in balanced neighbourhoods. Hence, the policy advice provided in this section does not feature specific, quantified metrics, but rather point towards a general policy direction given the conclusions drawn. We found that balanced neighbourhoods are not associated with higher levels of social cohesion, and therefore of resilient action. In fact, the higher the balance, the fewer residents are willing to help their friends and neighbours. Consequently, if the objective is to increase social cohesion and resilient actions, we would discourage the municipality of Rotterdam to approach this by building balanced neighbourhoods. Even so, we have seen the relevance of tailored neighbourhood analyses to come up with strong inference. The history, development, and composition of each neighbourhood should be considered. This has an impact in the characteristics of the population as well as the built environment. As a result, although the research outcomes show that balanced neighbourhoods are not positively associated to their alleged effects, it is hard to attribute the results to the housing policies.

Moving residents around in space to achieve balance may not address many of the underlying reasons for the pathology of deprived areas. Balancing a neighbourhood is a very indirect and lengthy way of getting the resources to residents in the need of help, therefore, the municipality should focus on alternatives to foster resilient actions between residents. In this study, we have seen that building social cohesion is a way to build social resilience, but the question arises whether the excluded (vulnerable) residents really want to be mixed. The movement Recht op de Stad shows that in Rotterdam this mix is imposed, rather than requested (Right to the City, 2022). The movement highlights that the end goal is to have a city for the residents that live there, and not a technocratic city governed by indicators of balance. In light of this, the municipality could favour a bottom-up participatory approach, in which they ask residents what is wrong with the neighbourhood and provide resources directly. In this sense, the BoTu programme is a step forward from the Woonvisie in this regard with the development of neighbourhood organisations which not only provide a space for social cohesion but also to collect and share resources (Municipality of Rotterdam, 2022a). Instead of balancing the neighbourhoods and letting the free market supply the resources to families, the programme provides resources directly.

On a separate note, we also encourage the municipality of Rotterdam to develop a more specific and transparent definition of neighbourhood balance if they were to continue with the policy. The ranges of the conditions could be neighbourhood specific to openly show what is the target that the municipality wants to achieve in each neighbourhood. This would allow for a more detailed analysis on the effects of the policy on the resilience of the city as well as prevent possible misinterpretations.

6.3. LIMITATIONS AND FUTURE RESEARCH DIRECTIONS

The research approach showed the strong capabilities of the computational framework to perform this study. It specified multidimensional constructs; direct, mediating, and moderating effects between variables; and considered spatial phenomena. Even so, we acknowledge a series of limitations and show how we mitigated them or how future research could approach them.

 Validity of results: Testing cause and effect relationships requires an assessment of the validity of the outcomes to conclude we have evidence of causality (Barth et al., 2011). Validity can be divided into the following categories (Hox, 2017): internal validity, external validity, and construct validity.

- (a) Internal validity: refers to the 'degree to which a study or experiment is free from flaws in its internal structure and its results can therefore be taken to represent the true nature of the phenomenon' (Guha, 2017). Internal validity was guaranteed at a theory level and at a statistical level (Marczyk et al., 2005). First, a literature review which combined academic and policy literature laid the grounds for the theory to test to provide a theoretical validity support. The internal validity of the results was also validated in the discussion of principal findings by comparing the results to alternative methods in the field and to previous studies (Barth et al., 2011). In addition, two interviews were conducted to two experts for additional consultation (Barth et al., 2011). Second, statistical validity is threatened by the PLS-SEM goodness-of-fit indicators. Internal validity is threatened by the omitted variable bias or the spatial autocorrelation bias, explained below in the 'PLS-SEM' item.
- (b) External validity: refers to 'the extent to which the results of research or testing can be generalized beyond the sample that generated the results to other individuals or situations' (Guha, 2017). This is concerned with how generalizable the findings are. For instance, do the findings apply to other cities? Replication would be the best way of guaranteeing external validity, however, due to time constraints, the framework was not applied to other contexts (Marczyk et al., 2005). According to Doff (2022, personal communication), the concepts studied here are very general sociological patterns, so they should not be context dependent on the city itself and can be generalised. Nevertheless, future research may address this limitation by replicating the computational framework to other areas.
- (c) Construct validity: refers to 'the degree to which a test or instrument is capable of measuring a theoretical construct, trait, or ability' (Guha, 2017). In our research, *Social cohesion* and *Informal support* were measured using previously validated psychometric scales used by the Dutch government (Schnabel et al., 2008). The PLS-SEM statistical tests also guaranteed the convergence validity, explanatory power, and predictive accuracy of the constructs. Even so, we recommend that future research re-validates the psychometric scales in the Wijkprofiel, as our results cast doubts on their validity.
- 2. **Data**: This research heavily relied on the availability of public data. On this account, the following assumptions and limitations were identified:
 - (a) Willingness to help friends is confounded with willingness to help neighbours: The theory of the model is based on the grounds that geographically connected people become affected by the social mix in the area to promote solidarity. Nevertheless, to measure informal support, it uses the indicator % of residents who say they are willing to care for neighbours or friends who need help (IS1). This indicator combines neighbours, who are geographically connected to the survey respondent, and friends, who are not necessarily geographically connected. Therefore, the assumption that social mix is the

mechanism that triggers neighbours to cooperate does not need to be the same as the one that triggers friends to help each other, as friendly relationships are driven by mutual affection (Felder, 2020). We recommend the Wijkprofiel researchers to separate willingness to help friends and neighbours into two questions in future surveys.

- (b) Neighbourhood aggregation: The Wijkprofiel data for the neighbourhood of Groot IJsselmonde had to be aggregated from Groot IJsselmonde North and Groot IJsselmonde South. For this, the weighted average using the number of respondents was used. Nevertheless, some respondents may have not answered, biasing the estimation. Future research should correct for the actual number of respondents if microdata is available.
- (c) WOZ-value aggregation: The Social, Middle, Higher, and Top WOZ-value brackets had to be adapted to the available data by OBI. Given the narrow width of the Middle segment (just €50,000), this could bias the estimation results. Future research should procure more precise WOZ-value segments.
- (d) Ecological fallacy: Relationships that are regarded as true at a level of aggregation, may not be true at another level (G. King, 2013). The fact that aggregate data of the Wijkprofiel survey yields the present results does not mean that at an individual level the relationships would follow the same way. Thus, the aggregate relationship could go in the opposite direction to the individual relationship. Future research should consider multilevel regression to combine different levels of aggregation to increase the validity of the results. So far, the multilevel capabilities of structural equation modelling are limited, so developments on the methodological level are limiting (Rabe-Hesketh et al., 2007). Furthermore, we also suggest that the relationships are not just studied at administrative levels of aggregation. Social cohesion and social resilience are phenomena that cross administrative boundaries, so future research could incorporate network analysis as a different aggregation mechanism.
- (e) Selection bias: This study relies on the validity of the Wijkprofiel survey. Nevertheless, Doff (2022, personal communication) suggested that residents from the South of Rotterdam are less keen than residents in the North to fill in surveys. Consequently, the possibility that failure to accomplish correct randomization that is representative of the population arises. We rely on the expertise of the researchers in OBI that conducted the Wijkprofiel survey to guarantee the mitigation of this bias.
- 3. **PLS-SEM**: The principal relationships for this study were obtained by means of PLS-SEM. Even though it proved to be a complete methodology to satisfy the many requirements of the analysis, some limitations were present.
 - (a) Spatial autocorrelation bias: PLS-SEM does not consider the spatial characteristics of the data. From the results of the spatial analysis, we have seen that there is presence of spatial spillover, so the outcomes of the structural model are biased. This involves the calculation of the relationships between

constructs as well as the measurement of the constructs themselves. Effort has already been put into incorporating spatial autoregression within SEM, however, it is at its early stages and not available in popular packages.

- (b) Omitted variable bias: Our model showed a good fit, which guarantees internal validity and the existence of the studied relationships. Nevertheless, the results from moderation analysis showed that *Ethnic heterogeneity*, and *House type heterogeneity* have significant direct effects on *Social cohesion* and *Informal support*, respectively. Consequently, future research should elaborate on the theory on which the model is grounded and create coherence to the empirical relationships identified with *Ethnic heterogeneity* and *House type heterogeneity*.
- (c) Single-indicator latent variable: Given that only one indicator was used for the measurement of *Informal support*, it is not possible to determine the amount of random measurement error in the indicator based on the available data (Henseler, 2020). Literature recommends that, in these cases, an additional error is imposed from the modeller (Brown, 2015), however, the package cSEM does not have that feature yet. We recommend that the study is replicated when the capability is incorporated.
- 4. **Spatial analysis:** The main purpose of the spatial analysis in this research was to search for the presence of local or spillover effects that could increase the explanatory power of the outcomes. First, the outcomes of the analysis are limited given the proposed weight matrix (see Equation (3.8)). Future research could change the structure of the weight matrix to incorporate a decay effect as distance increases. Furthermore, the effects of the river that divides Rotterdam were considered superficially. Future research should deepen into the structure and strength of the social interactions between both sides of the river by means of space syntax or network theory (van Nes & Yamu, 2021). Social processes arise from social interaction, and space syntax and network theory would allow quantifying and describing the places of interaction. Second, spatial spillover or contagion effects can be further tested with more complex spatial models that can capture non-stationarity. Future research could, for instance, use geographically weighted regression to increase the accuracy of the results (Ewing & Park, 2020).
- 5. **Computational power:** In this study, the models were estimated using the same objective WOZ-value distribution for each neighbourhood due to the computational power limitation. The computational framework could be adapted to calculate all the combinations of every balance distribution for each neighbourhood to select which balance explains better the relationships in each neighbourhood. This would provide specific insights for the building programme for each neighbourhood, as well as a more concrete measure for each.
- 6. **High abstraction:** The research has analysed the relationships of moderating factors at a high level of abstraction, i.e. low level of detail. As we have seen in the literature review, urban factors like walkability and greenery have been found to be significant in previous research. However, our study has only considered these

indicators at a high level of abstraction to capture the effect of these variables on the social cohesion and the willingness to help (e.g. greenery as percentage of green area in the neighbourhood and walkability as percentage of walkable area). Future research should take into account the design elements of the urban environment that could play a role in the actions and perceptions of residents (Rogers & Sukolratanametee, 2009; Wood et al., 2010). A first step has been shown in Appendix F which hopefully opens the door to future research. In addition, we have seen that each neighbourhood could have a different goal that achieves balance in relation to the population and history in the neighbourhood. Consequently, future research should tailor the analysis to the characteristics of every neighbourhood. The BoTu programme is a good start from the hand of the municipality of Rotterdam.

7. Deductive approach for causal inference: Following the deductive research approach, we first elaborated a theory based on the literature, and then we tested whether the theory holds according to the data. The model represents the causal assumptions stated by the researcher, and the credibility of these assumptions depends on their theoretical foundations. The robustness of the causal inference is based on the validity of the theory. Finally, what the results show is whether the data rejects the null hypothesis. To increase the validity, this thesis developed a literature-grounded conceptualisation, validated the results with previous research, and with expert interviews. To increase the reliability of the causal inference, future research should perform a longitudinal study of the modelled relationships (Spector, 2019). Longitudinal studies are preferred to determine the direction of causal relationships due to their ability to detect changes in the population. Furthermore, future research could conduct an impact evaluation to infer causality to a larger extent. Parallel trends is an example of impact evaluation analysis that could be useful (Angrist & Pischke, 2014). It can compare neighbourhoods in which the balance has not been changed, to assess if treated neighbourhoods have seen an improvement in resilience in comparison.

In addition, cross-sectional studies limit causal inferences because of the possibility of residual confounding and the inability to take into account the neighbourhood self-selection (Cobb et al., 2015). This brings to the next well known causal inference problem in neighbourhood studies: **Self-selection bias** (not to confuse with the previously mentioned selection bias). This bias suggests that the statistical model outcomes cannot determine whether the neighbourhood characteristics are the cause or the effect, i.e. if the characteristics are caused by a certain effect or if a certain effect is caused by the characteristics of the neighbourhood (Custers, 2021; Manley et al., 2011; van Ham & Manley, 2012). In the case of resilience, the problem is whether the neighbourhood favours resilient responses to problems, or people who have resilient capabilities are often able to concentrate in some areas of the city (usually brought by their income). Put simply, people selfselect themselves into neighbourhood sand the selection mechanism explains the associations on the neighbourhood level (Manley et al., 2011). The same recommendations of longitudinal studies or impact evaluation can reduce this bias. In conclusion, using Rotterdam as a case study, we have seen that balanced neighbourhoods do not foster resilient action between friends and neighbours. Consequently, we discourage the municipality to employ balanced neighbourhoods as policies to increase the social resilience of the city. However, we have also listed a series of limitations of the study which future research should overcome to provide finer conclusions and recommendations.

BIBLIOGRAPHY

- Abada, T., Hou, F., & Ram, B. (2007). Racially mixed neighborhoods, perceived neighborhood social cohesion, and adolescent health in Canada. *Social Science & Medicine*, 65(10), 2004–2017. https://doi.org/10.1016/j.socscimed.2007.06.030
- ABN-AMRO. (2022). Dutch National Mortgage Guarantee (NHG). Retrieved May 5, 2022, from https://www.abnamro.nl/en/personal/mortgages/buying-a-house/ dutch-national-mortgage-guarantee/index.html
- Adger, W. N., Kelly, P. M., Winkels, A., Huy, L. Q., & Locke, C. (2002). Migration, remittances, livelihood trajectories, and social resilience. *Ambio*, 31(4), 358–366. https: //doi.org/10.1579/0044-7447-31.4.358
- Aldrich, D. P. (2017). The Importance of Social Capital in Building Community Resilience. In W. Yan & W. Galloway (Eds.), *Rethinking Resilience, Adaptation and Transformation in a Time of Change* (pp. 357–364). Springer International Publishing. https://doi.org/10.1007/978-3-319-50171-0_23
- Allport, G. W., Clark, K., & Pettigrew, T. (1954). The nature of prejudice. Retrieved June 25, 2022, from https://scholar.google.com/scholar_lookup?title=The%20Nature% 20of%20Prejudice&author=G.%20Allport&publication_year=1954
- Angrist, J. D., & Pischke, J.-S. (2014). *Mastering'metrics: The path from cause to effect.* Princeton university press.
- Anselin, L. (1988). The Maximum Likelihood Approach to Spatial Process Models. In L. Anselin (Ed.), *Spatial Econometrics: Methods and Models* (pp. 57–80). Springer Netherlands. https://doi.org/10.1007/978-94-015-7799-1_6
- Anselin, L. (1995). Local Indicators of Spatial Association—LISA. *Geographical Analysis*, 27(2), 93–115. https://doi.org/10.1111/j.1538-4632.1995.tb00338.x
- Arribas-Bel, D. (2015). Spatial autocorrelation and Exploratory Spatial Data Analysis -Lab 6. http://darribas.org/gds15/content/labs/lab_06.html
- Arup. (2015). City Resilience Framework (tech. rep.). The Rockefeller Foundation. London. Retrieved March 7, 2022, from https://www.rockefellerfoundation.org/ wp-content/uploads/City-Resilience-Framework-2015.pdf
- Australian Institute of Health and Welfare. (2021). *Australia's welfare 2021: In brief* (tech. rep.). Canberra. Retrieved March 21, 2022, from https://www.aihw.gov.au/getmedia/b5c49e10-c8c7-4629-b183-83354894204b/aihw-aus-237.pdf.aspx?inline=true
- Baetens, J. M., & De Baets, B. (2012). Cellular automata on irregular tessellations [Publisher: Taylor & Francis _eprint: https://doi.org/10.1080/14689367.2012.711300].
 Dynamical Systems, 27(4), 411–430. https://doi.org/10.1080/14689367.2012.
 711300
- Baldwin, C., & King, R. (2017). What about the people? The socially sustainable, resilient community and urban development (tech. rep.). Retrieved May 15, 2022, from

https://www.anthro.ox.ac.uk/sites/default/files/anthro/documents/media/ what_about_the_people_report_baldwin_king_2017-brookes.pdf

- Barth, A., Caillaud, E., & Rose, B. (2011). HOW TO VALIDATE RESEARCH IN ENGINEER-ING DESIGN? DS 68-2: Proceedings of the 18th International Conference on Engineering Design (ICED 11), Impacting Society through Engineering Design, Vol. 2: Design Theory and Research Methodology, Lyngby/Copenhagen, Denmark, 15.-19.08.2011, 41–50. Retrieved August 9, 2022, from https://www.designsociety. org/publication/30456/HOW+TO+VALIDATE+RESEARCH+IN+ENGINEERING+ DESIGN%3F
- Beauvais, C., & Jenson, J. (2002). Social cohesion: Updating the state of the research (Vol. 62). CPRN Ottawa.
- Benitez-Avila, C. A., Schuberth, F., & Copeland, S. (2022). *Governing day-to-day life with the lenses of social resilience? Do not overlook people's positions within social orders. [Unpublished manuscript]* (tech. rep.). Delft University of Technology.
- Berkman, L. F., Glass, T., Brissette, I., & Seeman, T. E. (2000). From social integration to health: Durkheim in the new millennium. *Social Science & Medicine (1982)*, 51(6), 843–857. https://doi.org/10.1016/s0277-9536(00)00065-4
- Bernard, P. (1999). Social cohesion: A critique. Canadian Policy Research Networks.
- Bolt, G., Phillips, D., & Van Kempen, R. (2010). Housing Policy, (De)segregation and Social Mixing: An International Perspective. *Housing Studies*, 25(2), 129–135. https:// doi.org/10.1080/02673030903564838
- Bourdieu, P. (1986). The forms of capital. *Cultural theory: An anthology, 1,* 81–93.
- Bricocoli, M., & Cucca, R. (2016). Social mix and housing policy: Local effects of a misleading rhetoric. The case of Milan [Publisher: SAGE Publications Ltd]. *Urban Studies*, 53(1), 77–91. https://doi.org/10.1177/0042098014560499
- Brown, T. A. (2015). *Confirmatory factor analysis for applied research, 2nd ed* [Pages: xvii, 462]. The Guilford Press.
- Cabrera, J. F. (2013). New Urbanism and Selection Bias in the Formation of Social Capital [Publisher: Routledge _eprint: https://doi.org/10.1080/10511482.2013.766626]. *Housing Policy Debate*, 23(2), 376–394. https://doi.org/10.1080/10511482.2013. 766626
- CBS. (2012). Hulp en steun van familie, vrienden en kennissen. Retrieved June 1, 2022, from https://www.cbs.nl/nl-nl/achtergrond/2012/29/hulp-en-steun-vanfamilie-vrienden-en-kennissen
- CBS. (2015). Sociale samenhang in het raamwerk van participatie, vertrouwen en integratie. In H. Schmeets (Ed.), *Sociale samenhang: Wat on bindt en verdeelt* (pp. 9– 14). CBS.
- Center for American Progress. (n.d.). Social and Cultural Cohesion. Retrieved June 1, 2022, from https://livewellaz.org/tag/social-and-cultural-cohesion/
- Chakraborty, A., & McMillan, A. (2018). Is Housing Diversity Good for Community Stability?: Evidence from the Housing Crisis. *Journal of Planning Education and Research*, 0739456X18810787. https://doi.org/10.1177/0739456X18810787
- Chan, J., To, H.-P., & Chan, E. (2006). Reconsidering Social Cohesion: Developing a Definition and Analytical Framework for Empirical Research. *Social Indicators Research*, 75(2), 273–302. https://doi.org/10.1007/s11205-005-2118-1

- Chaskin, R. J., & Joseph, M. L. (2011). Social Interaction in Mixed-Income Developments: Relational Expectations and Emerging Reality. *Journal of Urban Affairs*, 33(2), 209–237. https://doi.org/10.1111/j.1467-9906.2010.00537.x
- Cheshire, L. (2015). 'Know your neighbours': Disaster resilience and the normative practices of neighbouring in an urban context. *Environment and Planning A: Economy and Space*, 47(5), 1081–1099. https://doi.org/10.1177/0308518X15592310
- Cheshire, P. (2009). Policies for Mixed Communities: Faith-Based Displacement Activity? International Regional Science Review, 32(3), 343–375. https://doi.org/10.1177/ 0160017609336080
- Chin, W. W. (1998). The partial least squares approach for structural equation modeling. *Modern methods for business research* (pp. 295–336). Lawrence Erlbaum Associates Publishers.
- Claridge, T. (2018). Functions of social capital–bonding, bridging, linking. *Social capital research*, *20*, 1–7.
- Cobb, L. K., Appel, L. J., Franco, M., Jones-Smith, J. C., Nur, A., & Anderson, C. A. M. (2015). The relationship of the local food environment with obesity: A systematic review of methods, study quality, and results. *Obesity (Silver Spring, Md.)*, 23(7), 1331–1344. https://doi.org/10.1002/oby.21118
- Coffé, H., & Geys, B. (2006). Community Heterogeneity: A Burden for the Creation of Social Capital? Social Science Quarterly, 87(5), 1053–1072. Retrieved March 7, 2022, from https://www.jstor.org/stable/42956594
- Cohen, J. (1988). *Statistical Power Analysis for the Behavioral Sciences* (2nd ed.). Routledge. https://doi.org/10.4324/9780203771587
- Colomb, C. (2011). Urban regeneration and policies of "social mixing" in British cities: A critical assessment. *ACE: Architecture, City and Environment,* 6(17), 223–244. https://doi.org/10.5821/ace.v6i17.2535
- Copeland, S., Comes, T., Bach, S., Nagenborg, M., Schulte, Y., & Doorn, N. (2020). Measuring social resilience: Trade-offs, challenges and opportunities for indicator models in transforming societies. *International Journal of Disaster Risk Reduction*, 51, 101799. https://doi.org/10.1016/j.ijdrr.2020.101799
- Council of Europe. (2010). *New Strategy and Council of Europe Action Plan for Social Cohesion* (tech. rep.). Retrieved March 21, 2022, from https://www.coe.int/t/dg3/socialpolicies/socialcohesiondev/source/2010Strategy_ActionPlan_SocialCohesion.pdf
- Curley, A. M. (2009). Draining or gaining? The social networks of public housing movers in Boston. *Journal of Social and Personal Relationships*, *26*(2-3), 227–247. https: //doi.org/10.1177/0265407509106716
- Custers, G. (2021). *The new divided city: Class transformation, civic participation and neighbourhood context.* Retrieved February 25, 2022, from https://repub.eur. nl/pub/135427
- Dall'erba, S. (2009). Exploratory Spatial Data Analysis. In R. Kitchin & N. Thrift (Eds.), International Encyclopedia of Human Geography (pp. 683–690). Elsevier. https: //doi.org/10.1016/B978-008044910-4.00433-8
- de Bruijn, H. (2019). *The art of political framing: How politicians convince us that they are right*. Amsterdam University Press. https://doi.org/10.5117/9789463721127

- de Saint-Exupéry, A. (1943). Le petit prince [The little prince]. Verenigde State van Amerika: Reynal & Hitchkock (US), Gallimard (FR).
- Diamantopoulos, A., & Winklhofer, H. M. (2001). Index Construction with Formative Indicators: An Alternative to Scale Development. *Journal of Marketing Research*, 38(2), 269–277. https://doi.org/10.1509/jmkr.38.2.269.18845
- Dijkstra, T. K., & Henseler, J. (2011). Linear indices in nonlinear structural equation models: Best fitting proper indices and other composites. *Quality & Quantity*, 45(6), 1505. https://doi.org/10.1007/s11135-010-9359-z
- Dijkstra, T. K., & Henseler, J. (2015). Consistent Partial Least Squares Path Modeling. *MIS Quarterly*, 39(2), 297–316. Retrieved May 7, 2022, from https://www.jstor.org/stable/26628355
- Doff, W. (2017). VEERKRACHT VAN LOKALE GEMEENSCHAPPEN: DE LITERATUUR OP EEN RIJ, 20.
- Doff, W. (2022). Personal communication.
- Dragolov, G., Ignácz, Z., Lorenz, J., Delhey, J., & Boehnke, K. (2013). Social Cohesion Radar Measuring Common Ground: An international Comparison of Social Cohesion Methods Report. Retrieved June 15, 2022, from http://aei.pitt.edu/ 74134/
- Dudovskiy, J. (2016). The ultimate guide to writing a dissertation in business studies: A step-by-step assistance. *Pittsburgh, USA, 51.*
- Duncan, O. D., & Duncan, B. (1955). A Methodological Analysis of Segregation Indexes. *American Sociological Review*, 20(2), 210–217. https://doi.org/10.2307/2088328
 Durkheim É. (1907). Étudo de sociological Alegu. Paris.
- Durkheim, É. (1897). Étude de sociologie. Alcan, Paris.
- Elliott, J. R., Haney, T. J., & Sams-Abiodun, P. (2010). Limits to social capital: Comparing network assistance in two New Orleans neighborhoods devastated by Hurricane Katrina. *The Sociological Quarterly*, 51(4), 624–648. https://doi.org/10. 1111/j.1533-8525.2010.01186.x
- European Commission. (2001). *The social situation in the European Union 2001* (tech. rep.). Office for Official Publications of the European Communities. Luxembourg. Retrieved March 21, 2022, from https://ec.europa.eu/employment_social/social_situation/docs/SSR2001_en.pdf
- European Commission. (2010). Europe 2020. A strategy for smart, sustainable and inclusive growth, COM (2010) 2020 final.
- Ewing, R., & Park, K. (Eds.). (2020). Advanced Quantitative Research Methods for Urban Planners. Routledge. https://doi.org/10.4324/9780429325038
- Fay-Ramirez, S., Antrobus, E., & Piquero, A. R. (2015). Assessing the effect of the Queensland "Summer of Disasters" on perceptions of collective efficacy. *Social Science Research*, 54, 21–35. https://doi.org/10.1016/j.ssresearch.2015.06.017
- Felder, M. (2020). Strong, Weak and Invisible Ties: A Relational Perspective on Urban Coexistence. *Sociology*, 54(4), 675–692. https://doi.org/10.1177/0038038519895938
- Field, A. (2013). Discovering Statistics Using IBM SPSS Statistics. SAGE.
- Fonseca, X., Lukosch, S., & Brazier, F. (2019). Social cohesion revisited: A new definition and how to characterize it. *Innovation: The European Journal of Social Science Research*, 32(2), 231–253. https://doi.org/10.1080/13511610.2018.1497480

- Forrest, R., & Kearns, A. (2001). Social Cohesion, Social Capital and the Neighbourhood. Urban Studies, 38(12), 2125–2143. https://doi.org/10.1080/00420980120087081
- Galster, G., Andersson, R., & Musterd, S. (2015). Are Males' Incomes Influenced by the Income Mix of Their Male Neighbors? Explorations into Nonlinear and Threshold Effects in Stockholm. *Housing Studies*, 30(2), 315–343. https://doi.org/10.1080/ 02673037.2014.931357
- Garrido-Velarde, J., Montero-Parejo, M. J., Hernández-Blanco, J., & García-Moruno, L. (2018). Visual Analysis of the Height Ratio between Building and Background Vegetation. Two Rural Cases of Study: Spain and Sweden. *Sustainability*, *10*(8), 2593. https://doi.org/10.3390/su10082593
- Gorsuch, R. L. (1990). Common Factor Analysis versus Component Analysis: Some Well and Little Known Facts. *Multivariate Behavioral Research*, 25(1), 33–39. https: //doi.org/10.1207/s15327906mbr2501_3
- Graham, E., Manley, D., Hiscock, R., Boyle, P., & Doherty, J. (2009). Mixing Housing Tenures: Is it Good for Social Well-being? *Urban Studies*, 46(1), 139–165. https://doi.org/ 10.1177/0042098008098640
- Guha, M. (2017). APA College Dictionary of Psychology (second edition) [Publisher: Emerald Publishing Limited]. *Reference Reviews*, 31(2), 11–11. https://doi.org/10. 1108/RR-10-2016-0240
- Hair, J. (2009). Multivariate Data Analysis. *Faculty Publications*. https://digitalcommons. kennesaw.edu/facpubs/2925
- Hair, J., Hult, T., Ringle, C., & Sarstedt, M. (2016). *A Primer on Partial Least Squares Structural Equation Modeling (PLS-SEM)*. Retrieved May 10, 2022, from https://uk. sagepub.com/en-gb/eur/a-primer-on-partial-least-squares-structuralequation-modeling-pls-sem/book244583
- Hair, J., Risher, J. J., Sarstedt, M., & Ringle, C. M. (2019). When to use and how to report the results of PLS-SEM. *European Business Review*, 31(1), 2–24. https://doi.org/ https://doi.org/10.1108/EBR-11-2018-0203
- Hananel, R., Fishman, R., & Malovicki-Yaffe, N. (2022). Urban diversity and epidemic resilience: The case of the COVID-19. *Cities*, *122*, 103526. https://doi.org/10. 1016/j.cities.2021.103526
- Harell, A., & Stolle, D. (2010). Reconciling Diversity and Community? Defining Social Cohesion in Diverse Democracies. *Social cohesion : Contemporary theoretical perspectives on the study of social cohesion and social capital.*
- Hartig, T., Mitchell, R., de Vries, S., & Frumkin, H. (2014). Nature and health. *Annual Review of Public Health*, 35, 207–228. https://doi.org/10.1146/annurev-publhealth-032013-182443
- Haughton, J., & Khandker, S. R. (2009). *Handbook on Poverty and Inequality*. World Bank. https://openknowledge.worldbank.org/handle/10986/11985
- Heinrich Mora, E., Heine, C., Jackson, J. J., West, G. B., Yang, V. C., & Kempes, C. P. (2021). Scaling of urban income inequality in the USA. *Journal of The Royal Society Interface*, 18(181). https://doi.org/10.1098/rsif.2021.0223
- Henseler, J. (2020). Composite-Based Structural Equation Modeling: Analyzing Latent and Emergent Variables. Guilford Press. Retrieved May 4, 2022, from https://research.

utwente.nl/en/publications/composite-based-structural-equation-modelinganalyzing-latent-and

- Hochstenbach, C., Uitermark, J., & van Gent, W. (2015). Evaluatie effecten Wet bijzondere maatregelen grootstedelijke problematiek ("Rotterdamwet") in Rotterdam. Retrieved March 9, 2022, from https://dare.uva.nl/search?identifier=2b95a138-23f6-4de5-9d7f-2266259cbb6d
- Hox, J. J. (2017). Computational social science methodology, anyone? *Methodology: European Journal of Research Methods for the Behavioral and Social Sciences*, 13, 3–12. Retrieved August 9, 2022, from https://doi.org/10.1027/1614-2241/a000127
- Hu, L.-t., & Bentler, P. M. (1999). Cutoff criteria for fit indexes in covariance structure analysis: Conventional criteria versus new alternatives. *Structural Equation Modeling: A Multidisciplinary Journal*, 6(1), 1–55. https://doi.org/10.1080/10705519909540118
- Huang, G. (2020). *Flood Impact Mitigation and Resilience Enhancement*. BoD Books on Demand.
- Huang, Y. (2013). *Automated simulation model generation* (Dissertation (TU Delft)) [ISBN: 9789462034617]. Delft. https://doi.org/10.4233/uuid:dab2b000-eba3-42ee-8eab-b4840f711e37
- Hulse, K., & Stone, W. (2007). Social Cohesion, Social Capital and Social Exclusion. *Policy Studies*, *28*(2), 109–128. https://doi.org/10.1080/01442870701309049
- IBM Corp. (2020). *IBM SPSS Statistics for Windows, Version 27.0* (tech. rep.). Armonk, NY: IBM Corp.
- ILO. (2020). Asia-Pacific Employment and Social Outlook 2020: Navigating the crisis towards a human-centred future of work (Report) [ISBN: 9789220338452]. Retrieved March 1, 2022, from http://www.ilo.org/asia/publications/WCMS_764084/ lang--en/index.htm
- International and Ibero-American Foundation for Administration and Public Policies. (2011). *Strategies for integrating social cohesion in public policies* (tech. rep.). Retrieved March 21, 2022, from https://issuu.com/fiiapp/docs/fiiapp_paris_ ocde_2_
- Islam, M. K., Gerdtham, U.-G., Gullberg, B., Lindström, M., & Merlo, J. (2008). Social capital externalities and mortality in Sweden. *Economics & Human Biology*, 6(1), 19–42. https://doi.org/10.1016/j.ehb.2007.09.004
- Jordan, R. (2018). *The Dutch Urban Areas Act : A black liberation perspective on the policy of exclusion* (Thesis). Massachusetts Institute of Technology. Retrieved March 2, 2022, from https://dspace.mit.edu/handle/1721.1/118207
- Kawachi, I., Kennedy, B. P., Lochner, K., & Prothrow-Stith, D. (1997). Social capital, income inequality, and mortality. *American Journal of Public Health*, 87(9), 1491– 1498. https://doi.org/10.2105/ajph.87.9.1491
- Kearns, A., & Forrest, R. (2000). Social cohesion and multilevel urban governance. *Urban studies*, *37*(5-6), 995–1017.
- Kearns, A., & Mason, P. (2007). Mixed Tenure Communities and Neighbourhood Quality. Housing Studies, 22(5), 661–691. https://doi.org/10.1080/02673030701474628
- Kim, J., & Kaplan, R. (2004). Physical and Psychological Factors in Sense of Community: New Urbanist Kentlands and Nearby Orchard Village. *Environment and Behavior*, 36(3), 313–340. https://doi.org/10.1177/0013916503260236

- King, D., & MacGregor, C. (2000). Using social indicators to measure community vulnerability to natural hazards. *Australian Journal of Emergency Management*, 15(3), 52–57.
- King, G. (2013). A Solution to the Ecological Inference Problem: Reconstructing Individual Behavior from Aggregate Data. Princeton University Press. Retrieved July 22, 2022, from https://www.degruyter.com/document/doi/10.1515/9781400849208/ html
- Klein, C. (2013). Social Capital or Social Cohesion: What Matters For Subjective Well-Being? Social Indicators Research, 110(3), 891–911. Retrieved March 21, 2022, from https://www.jstor.org/stable/24719083
- Klinenberg, E. (2015). *Heat Wave: A Social Autopsy of Disaster in Chicago* (W. a. N. Preface, Ed.). University of Chicago Press. Retrieved June 1, 2022, from https://press.uchicago.edu/ucp/books/book/chicago/H/bo20809880.html
- Koohsari, M. J., Nakaya, T., McCormack, G. R., Shibata, A., Ishii, K., Yasunaga, A., Hanibuchi, T., & Oka, K. (2021). Traditional and novel walkable built environment metrics and social capital. *Landscape and Urban Planning*, 214, 104184. https://doi. org/10.1016/j.landurbplan.2021.104184
- Kullback, S. (1959). Information Theory and Statistics. *John Wiley & Sons. Republished by* Dover Publications in 1968; reprinted in 1978.
- Kullback, S. (1987). Letters to the Editor: The Kullback–Leibler distance. *The American Statistician*, 41(4), 338–341. Retrieved April 29, 2022, from https://www.jstor.org/stable/2684769
- Langer, A., Stewart, F., Smedts, K., & Demarest, L. (2017). Conceptualising and Measuring Social Cohesion in Africa: Towards a Perceptions-Based Index. *Social Indicators Research*, 131(1), 321–343. https://doi.org/10.1007/s11205-016-1250-4
- Larimian, T., Sadeghi, A., Palaiologou, G., & Schmidt III, R. (2020). Neighbourhood Social Resilience (NSR): Definition, Conceptualisation, and Measurement Scale Development. *Sustainability*, 12(16), 6363. https://doi.org/10.3390/su12166363
- Lauro, N. C., Grassia, M. G., & Cataldo, R. (2018). Model Based Composite Indicators: New Developments in Partial Least Squares-Path Modeling for the Building of Different Types of Composite Indicators. *Social Indicators Research*, 135(2), 421– 455. https://doi.org/10.1007/s11205-016-1516-x
- Leung, Y., Mei, C.-L., & Zhang, W.-X. (2003). Statistical Test for Local Patterns of Spatial Association. *Environment and Planning A: Economy and Space*, 35(4), 725–744. https://doi.org/10.1068/a3550
- Lubberink, A. S., van der Post, W., & Veuger, J. (2018). Valuation of real estate market values as an indicator. *Real estate finance*, *34*(4), 159–168.
- Manikandan, S. (2011). Measures of dispersion. *Journal of Pharmacology & Pharmacotherapeutics*, 2(4), 315–316. https://doi.org/10.4103/0976-500X.85931
- Manley, D., van Ham, M., & Doherty, J. (2011). Social Mixing as a Cure for Negative Neighbourhood Effects: Evidence Based Policy or Urban Myth? https://doi.org/10. 1332/policypress/9781847424938.003.0011
- Marczyk, G. R., DeMatteo, D., & Festinger, D. (2005). *Essentials of Research Design and Methodology* | *Wiley*. John Wiley & Sons. Retrieved August 9, 2022, from https:

//www.wiley.com/en-us/Essentials+of+Research+Design+and+Methodologyp-9780471470533

- Maxwell, J. (1996). *Social dimensions of economic growth*. Department of Economics, University of Alberta.
- Mazumdar, S., Learnihan, V., Cochrane, T., & Davey, R. (2018). The Built Environment and Social Capital: A Systematic Review. *Environment and Behavior*, *50*(2), 119–158. https://doi.org/10.1177/0013916516687343
- Meerow, S., Newell, J. P., & Stults, M. (2016). Defining urban resilience: A review. *Landscape and Urban Planning*, *147*, 38–49. https://doi.org/10.1016/j.landurbplan. 2015.11.011
- Méreiné-Berki, B., Málovics, G., & Creţan, R. (2021). "You become one with the place": Social mixing, social capital, and the lived experience of urban desegregation in the Roma community. *Cities*, *117*, 103302. https://doi.org/10.1016/j.cities.2021. 103302
- Miao, J., Wu, X., & Sun, X. (2019). Neighborhood, social cohesion, and the Elderly's depression in Shanghai. Social Science & Medicine, 229, 134–143. https://doi.org/ 10.1016/j.socscimed.2018.08.022
- Ministry for Housing and the Civil Service of the Netherlands. (2017). Urban Areas Special Measures Act. Retrieved March 2, 2022, from https://wetten.overheid.nl/ BWBR0019388/2017-07-01
- Ministry of General Affairs. (2016). How do municipalities value properties? Valuation of immovable property - Government.nl. Retrieved May 2, 2022, from https:// www.government.nl/topics/valuation-of-immovable-property/how-domunicipalities-value-properties
- Molenaar, A. (2016). Rotterdam Resilient Delta City. https://pt.slideshare.net/GRFDavos/ rotterdam-resilient-delta-city-arnaud-molenaar
- Molin, E. (2018). Introduction to experimental designs Lecture 1, SEN1221.
- Mouratidis, K., & Poortinga, W. (2020). Built environment, urban vitality and social cohesion: Do vibrant neighborhoods foster strong communities? *Landscape and Urban Planning*, 204, 103951. https://doi.org/10.1016/j.landurbplan.2020.103951
- Municipality of Rotterdam. (2016a). Rotterdam Resilience Strategy. Retrieved February 25, 2022, from https://www.resilientrotterdam.nl/en/download
- Municipality of Rotterdam. (2016b). Woonvisie Rotterdam koers naar 2030 agenda tot 2020. https://archief12.archiefweb.eu/archives/archiefweb/20200809090118/ http://www.rotterdam.nl/wonen-leven/woonvisie/DEFINITIEF-Woonvisie-Rotterdam-2030-dd-raad-15-december-2016.pdf
- Municipality of Rotterdam. (2019a). Nationaal Programma Rotterdam Zuid. https:// www.rotterdam.nl/nieuws/het-gaat-beter-op-zuid/NPRZ-uitvoeringsprogramma-2019-2022
- Municipality of Rotterdam. (2019b). RESILIENT BOTU 2028. https://bospoldertussendijken. nl/wp-content/uploads/2019/06/Brochure_Bospolder_3.0_English.pdf
- Municipality of Rotterdam. (2019c). Woonvisie Rotterdam koers naar 2030 agenda tot 2020 - ADDENDUM. https://rotterdam.raadsinformatie.nl/document/7594499/ 1/19bb12971

- Municipality of Rotterdam. (2020). *GEBIEDSATLAS 2.0* (tech. rep.). Municipality of Rotterdam. Retrieved June 18, 2022, from https://rotterdam.raadsinformatie.nl/ document/9112505/1/
- Municipality of Rotterdam. (2022a). Resilient Rotterdam Strategie 2022-2027. https://rotterdam.raadsinformatie.nl/document/11230526/2
- Municipality of Rotterdam. (2022b). Wijkprofiel Rotterdam. Retrieved March 10, 2022, from https://wijkprofiel.rotterdam.nl/nl/2022/rotterdam
- Musterd, S., & Andersson, R. (2005). Housing Mix, Social Mix, and Social Opportunities. *Urban Affairs Review*, 40(6), 761–790. https://doi.org/10.1177/1078087405276006
- Naganuma, Y., Kihara, S., Fujita, Y., Yamaoka, K., & Takahashi, K. (2018). The relation between social cohesion and the care burden of family healthcare providers. *Bioscience Trends*, *12*(5), 520–522. https://doi.org/10.5582/bst.2018.01175
- National Academies of Sciences, Engineering, and Medicine. (2020). Social Isolation and Loneliness in Older Adults: Opportunities for the Health Care System. The National Academies Press. https://doi.org/10.17226/25663
- Netherlands Enterprise Agency. (2022). Addresses and Buildings key register (BAG). Retrieved May 5, 2022, from https://business.gov.nl/regulation/addresses-andbuildings-key-geo-register/
- Noguchi, Y. (1991). Senior's social support—the concept and measurement. *Soc Gerontol*, *34*, 37–48.
- OECD. (2011). Perspectives on Global Development 2012: Social Cohesion in a Shifting World. Organisation for Economic Co-operation; Development. Retrieved March 21, 2022, from https://www.oecd-ilibrary.org/development/perspectives-onglobal-development-2012_persp_glob_dev-2012-en
- OECD. (2021). *Designing Policies for Efficient, Inclusive and Sustainable Housing* (tech. rep.). OECD. Paris. https://doi.org/10.1787/8ec0f507-en
- OpenStreetMap contributors. (2017). Planet dump retrieved from https://planet.osm.org [https://www.openstreetmap.org].
- Ouwehand, A. (2022). Personal communication.
- Overman, H. G. (2009). Regional Science. In R. Kitchin & N. Thrift (Eds.), *International Encyclopedia of Human Geography* (pp. 279–285). Elsevier. https://doi.org/10. 1016/B978-008044910-4.00737-9
- Park, S., Deakin, E., & Lee, J. S. (2014). Perception-Based Walkability Index to Test Impact of Microlevel Walkability on Sustainable Mode Choice Decisions [Publisher: SAGE Publications Inc]. *Transportation Research Record*, 2464(1), 126–134. https:// doi.org/10.3141/2464-16
- Pedregosa, F., Varoquaux, G., Gramfort, A., Michel, V., Thirion, B., Grisel, O., Blondel, M., Prettenhofer, P., Weiss, R., & Dubourg, V. (2011). Scikit-learn: Machine learning in Python. *the Journal of machine Learning research*, *12*, 2825–2830.
- Pelling, M. (2003). *The Vulnerability of Cities: Natural Disasters and Social Resilience*. Routledge. https://doi.org/10.4324/9781849773379
- Perrin, L., & Grant, J. L. (2014). Perspectives on mixing housing types in the suburbs. *The Town Planning Review*, 85(3), 363–385. Retrieved May 5, 2022, from https: //www.jstor.org/stable/24580651

- Peterson, M. (2017). Living with difference in hyper-diverse areas: How important are encounters in semi-public spaces? *Social & Cultural Geography*, 18(8), 1067– 1085. https://doi.org/10.1080/14649365.2016.1210667
- Photographie, L. d. l. (2016). Photographic Deconstruction. Retrieved August 11, 2022, from https://loeildelaphotographie.com/en/photographic-deconstruction/
- Problovski, H. Z. G., Sherman, M., & Melnikova, V. (2022). A link between social isolation during the coronavirus outbreak and social alignment in balcony parties [Publisher: Public Library of Science]. *PLOS ONE*, *17*(4), e0264109. https://doi.org/ 10.1371/journal.pone.0264109
- Putnam, R. D. (2000). *Bowling alone: The collapse and revival of American community.* Simon; schuster.
- Quigley, M., Blair, N., & Davison, K. (2018). Articulating a social-ecological resilience agenda for urban design. *Journal of Urban Design*, 23(4), 581–602. https://doi.org/10.1080/13574809.2018.1440176
- Rabe-Hesketh, S., Skrondal, A., & Zheng, X. (2007). 10 Multilevel Structural Equation Modeling. In S.-Y. Lee (Ed.), *Handbook of Latent Variable and Related Models* (pp. 209–227). North-Holland. https://doi.org/10.1016/B978-044452044 -9/50013-6
- Rademaker, M., & Schuberth, F. (2020). *cSEM: Composite-Based Structural Equation Modeling* (tech. rep.). https://m-e-rademaker.github.io/cSEM/
- Research and Business Intelligence. (2022). Research010 New. Retrieved May 5, 2022, from https://onderzoek010.nl/jive
- Rey, S., & Anselin, L. (2007). PySAL: A Python Library of Spatial Analytical Methods. *The Review of Regional Studies*, 37(1), 5–27. Retrieved May 13, 2022, from https://pysal.org/esda/
- Rey, S., Arribas-Bel, D., & Wolf, L. (2020). *Local Spatial Autocorrelation Geographic Data Science with Python*. Retrieved May 13, 2022, from https://geographicdata.science/book/notebooks/07_local_autocorrelation.html
- Rigdon, E. (2012). Rethinking Partial Least Squares Path Modeling: In Praise of Simple Methods. *Long Range Planning*, 45, 341–358. https://doi.org/10.1016/j.lrp. 2012.09.010
- Right to the City. (2021a). Een alternatief verhaal over wijken in balans | Doordenkavond #2 verslag. Retrieved March 9, 2022, from https://rechtopdestad.nl/een-alternatief-verhaal-over-wijken-in-balans-doordenkavond-2-verslag/
- Right to the City. (2021b). VN-Speciale Rapporteurs: Woonbeleid Rotterdam schendt mensenrechten. Retrieved March 9, 2022, from https://rechtopdestad.nl/vnspeciale-rapporteurs-woonbeleid-rotterdam-schendt-mensenrechten/
- Right to the City. (2022). Recht op de stad. Retrieved June 26, 2022, from https://rechtopdestad. nl/
- Roberto, E. (2015). *The Divergence Index: A Decomposable Measure of Segregation and Inequality* (tech. rep.). Retrieved April 29, 2022, from https://ui.adsabs.harvard.edu/abs/2015arXiv150801167R
- Rogers, G. O., & Sukolratanametee, S. (2009). Neighborhood design and sense of community: Comparing suburban neighborhoods in Houston Texas. *Landscape and*

Urban Planning, 92(3), 325–334. https://doi.org/10.1016/j.landurbplan.2009. 05.019

- Ruan, D. (2005). *Intelligent data mining : Techniques and applications*. Berlin ; New York : Springer-Verlag. Retrieved August 9, 2022, from http://archive.org/details/ intelligentdatam00ruan_742
- Ruiz-Tagle, J. (2019). Social Mix. *The Wiley Blackwell Encyclopedia of Urban and Regional Studies*, 1–5.
- Saja, A. M. A., Goonetilleke, A., Teo, M., & Ziyath, A. M. (2019). A critical review of social resilience assessment frameworks in disaster management. *International Journal of Disaster Risk Reduction*, 35, 101096. https://doi.org/10.1016/j.ijdrr.2019. 101096
- Sampson, R. J., Raudenbush, S. W., & Earls, F. (1997). Neighborhoods and violent crime: A multilevel study of collective efficacy. *Science (New York, N.Y.)*, 277(5328), 918– 924. https://doi.org/10.1126/science.277.5328.918
- Schiefer, D., & van der Noll, J. (2017). The Essentials of Social Cohesion: A Literature Review. Social Indicators Research, 132(2), 579–603. https://doi.org/10.1007/ s11205-016-1314-5
- Schiefer, D., van der Noll, J., Delhey, J., & Boehnke, K. (2012). *Cohesion Radar: Measuring Cohesiveness* (tech. rep.). Jacobs University Bremen. https://www.bertelsmann-stiftung.de/fileadmin/files/Projekte/Gesellschaftlicher_Zusammenhalt/englische_site/further-downloads/social-cohesion/Social_Cohesion_2012.pdf
- Schinkel, W., & van den Berg, M. (2011). City of Exception: The Dutch Revanchist City and the Urban Homo Sacer. *Antipode*, *43*(5), 1911–1938. https://doi.org/10. 1111/j.1467-8330.2010.00831.x
- Schmeets, H. (2012). Social Cohesion: An integrated empirical approach. *Contemporary Theoretical Perspectives on the Study of Social Cohesion and Social Capital*, 127.
- Schnabel, P., Bijl, R., & de Hart, J. (2008). *Betrekkelijke betrokkenheid: Studies in sociale cohesie*. Retrieved May 15, 2022, from https://catalogus.boekman.nl/pub/P09-153.pdf
- Schuberth, F., Henseler, J., & Dijkstra, T. K. (2018). Partial least squares path modeling using ordinal categorical indicators. *Quality & Quantity*, 52(1), 9–35. https:// doi.org/10.1007/s11135-016-0401-7
- Schwab, S., & Held, L. (2020). Different worlds Confirmatory versus exploratory research. *Significance*, 17(2), 8–9. https://doi.org/10.1111/1740-9713.01369
- Shih, W., & Chai, S. (2016). Data-Driven vs. Hypothesis-Driven Research: Making sense of big data. Academy of Management Proceedings, 2016(1), 14843. https://doi. org/10.5465/ambpp.2016.14843abstract
- Simpson, E. H. (1949). Measurement of Diversity. *Nature*, *163*(4148), 688–688. https://doi.org/10.1038/163688a0
- Skrondal, A., & Rabe-Hesketh, S. (2004). *Generalized latent variable modeling: Multilevel, longitudinal, and structural equation models.* Chapman; Hall/CRC.
- Sluiter, Tolsma, J., Scheepers, P., Sociology, & Social Change and Conflict (SCC). (2015). At which geographic scale does ethnic diversity affect intra-neighborhood social capital? *Social Science Research*, 54(November 2015), 80–95. https://doi.org/10. 1016/j.ssresearch.2015.06.015

- Small, M. L. (2007). Racial Differences in Networks: Do Neighborhood Conditions Matter?*. Social Science Quarterly, 88(2), 320–343. https://doi.org/10.1111/j.1540-6237.2007.00460.x
- Snow, D. A., Leahy, P. J., & Schwab, W. A. (1981). Social Interaction in a Heterogeneous Apartment: An Investigation of the Effects of Environment upon Behavior. *Sociological Focus*, 14(4), 309–319. https://doi.org/10.1080/00380237.1981. 10570404
- Spaans, M., & Waterhout, B. (2017). Building up resilience in cities worldwide Rotterdam as participant in the 100 Resilient Cities Programme. *Cities*, *61*, 109–116. https://doi.org/10.1016/j.cities.2016.05.011
- Spector, P. E. (2019). Do Not Cross Me: Optimizing the Use of Cross-Sectional Designs. Journal of Business and Psychology, 34(2), 125–137. https://doi.org/10.1007/ s10869-018-09613-8
- Staalduine, J. v. (2019). Rotterdam jaagt armen de stad uit. Retrieved May 2, 2022, from https://www.trouw.nl/gs-baa69256
- Stuij, C. (2020). RESILIENCE IN TIMES OF POPULATION DECLINE (Master's thesis). Radboud University. https://theses.ubn.ru.nl/bitstream/handle/123456789/9168/ Stuij%2C_Clara_1.pdf?sequence=1
- Takagi, D., & Shimada, T. (2019). A Spatial Regression Analysis on the Effect of Neighborhood-Level Trust on Cooperative Behaviors: Comparison With a Multilevel Regression Analysis. *Frontiers in Psychology, 10*. Retrieved March 22, 2022, from https: //www.frontiersin.org/article/10.3389/fpsyg.2019.02799
- Tolsma, J., van der Meer, T., & Gesthuizen, M. (2009). The impact of neighbourhood and municipality characteristics on social cohesion in the Netherlands. *Acta Politica*, 44(3), 286–313. https://doi.org/10.1057/ap.2009.6
- Tong, P. (2021). Characteristics, dimensions and methods of current assessment for urban resilience to climate-related disasters: A systematic review of the literature. *International Journal of Disaster Risk Reduction*, 60, 102276. https://doi.org/10. 1016/j.ijdrr.2021.102276
- Townshend, I., Awosoga, O., Kulig, J., & Fan, H. (2015). Social cohesion and resilience across communities that have experienced a disaster. *Natural Hazards*, 76(2), 913–938. https://doi.org/10.1007/s11069-014-1526-4
- Tunstall, R., & Lupton, R. (2010). *Mixed communities: Evidence review*. Department for Communities; Local Government.
- Uitermark, J., Hochstenbach, C., & van Gent, W. (2017). The statistical politics of exceptional territories. *Political Geography*, 57, 60–70. https://doi.org/10.1016/j. polgeo.2016.11.011
- UN-Habitat. (2016). New Urban Agenda | UN-Habitat. Retrieved February 25, 2022, from https://unhabitat.org/about-us/new-urban-agenda
- UN-Habitat. (2020). World Cities Report 2020: The Value of Sustainable Urbanization | UN-Habitat. Retrieved March 1, 2022, from https://unhabitat.org/World% 20Cities%20Report%202020
- UNISDR. (2015). Sendai Framework for Disaster Risk Reduction 2015-2030.
- United Nations, Department of Economic and Social Affairs, & Population Division. (2018). *World urbanization prospects: The 2018 revision* [OCLC: 1120698127].

- UN-OHCHR. (2021). Mandates of the Special Rapporteur on adequate housing as a component of the right to an adequate standard of living, and on the right to nondiscrimination in this context; the Special Rapporteur on the right to development; the Special Rapporteur on the human rights of migrants; the Special Rapporteur on minority issues and the Special Rapporteur on extreme poverty and human rights (tech. rep.). Report by Balakrishnan Rajagopal. Retrieved March 2, 2022, from https://spcommreports.ohchr.org/TMResultsBase/DownLoadPublicCommunication gId=26359
- van Beuningen, J., & Schmeets, H. (2013). Developing a Social Capital Index for the Netherlands. *Social Indicators Research*, 113(3), 859–886. https://doi.org/10.1007/ s11205-012-0129-2
- van Eijk, G. (2010). Exclusionary Policies are Not Just about the 'Neoliberal City': A Critique of Theories of Urban Revanchism and the Case of Rotterdam. *International Journal of Urban and Regional Research*, 34(4), 820–834. https://doi.org/ 10.1111/j.1468-2427.2010.00944.x
- van Gent, W., Hochstenbach, C., & Uitermark, J. (2018). Exclusion as urban policy: The Dutch 'Act on Extraordinary Measures for Urban Problems'. *Urban Studies*, 55(11), 2337–2353. https://doi.org/10.1177/0042098017717214
- van Ham, M., & Manley, D. (2010). The effect of neighbourhood housing tenure mix on labour market outcomes: A longitudinal investigation of neighbourhood effects. *Journal of Economic Geography*, *10*(2), 257–282. https://doi.org/10.1093/jeg/ lbp017
- van Ham, M., & Manley, D. (2012). Neighbourhood Effects Research at a Crossroads. Ten Challenges for Future Research Introduction. *Environment and Planning A: Economy and Space*, 44(12), 2787–2793. https://doi.org/10.1068/a45439
- van Nes, A., & Yamu, C. (2021). Established Urban Research Traditions and the Platform for Space Syntax. In A. van Nes & C. Yamu (Eds.), *Introduction to Space Syntax in Urban Studies* (pp. 1–34). Springer International Publishing. https://doi.org/ 10.1007/978-3-030-59140-3_1
- Veld Academie. (2021). VEERKRACHT IN BOSPOLDER-TUSSENDIJKEN. https://verhalenvanbotu. nl/wp-content/uploads/2021/04/20210412_Rapportage-BoTu-2_DEF-klein. pdf
- Versluis, R. (2017). Urban Renewal in Rotterdam (Master's thesis). Wageningen University. Wageningen. https://edepot.wur.nl/425058
- Virtanen, P., Gommers, R., Oliphant, T. E., Haberland, M., Reddy, T., Cournapeau, D., Burovski, E., Peterson, P., Weckesser, W., Bright, J., van der Walt, S. J., Brett, M., Wilson, J., Millman, K. J., Mayorov, N., Nelson, A. R. J., Jones, E., Kern, R., Larson, E., ... van Mulbregt, P. (2020). SciPy 1.0: Fundamental algorithms for scientific computing in Python. *Nature Methods*, *17*(3), 261–272. https://doi.org/10.1038/ s41592-019-0686-2
- Walker-Arnott, E. (2020). This is the coolest neighbourhood in the world right now. Retrieved March 1, 2022, from https://www.timeout.com/news/this-is-thecoolest-neighbourhood-in-the-world-right-now-100720

- Wang, Y., & Kemeny, T. (2022). Are mixed neighborhoods more socially cohesive? Evidence from Nanjing, China. Urban Geography, 0(0), 1–21. https://doi.org/10. 1080/02723638.2021.2021714
- Watkins, M. W. (2021). A Step-by-Step Guide to Exploratory Factor Analysis with SPSS. Routledge. https://doi.org/10.4324/9781003149347
- Wolf, T., Chuang, W.-C., & McGregor, G. (2015). On the Science-Policy Bridge: Do Spatial Heat Vulnerability Assessment Studies Influence Policy? *International Journal* of Environmental Research and Public Health, 12(10), 13321–13349. https://doi. org/10.3390/ijerph121013321
- Wood, L., Frank, L. D., & Giles-Corti, B. (2010). Sense of community and its relationship with walking and neighborhood design. *Social Science & Medicine*, 70(9), 1381– 1390. https://doi.org/10.1016/j.socscimed.2010.01.021
- Woolcock, M. (2001). Microenterprise and social capital:: A framework for theory, research, and policy. *The Journal of Socio-Economics*, *30*(2), 193–198.

A

APPENDIX A

A.1. DEFINITIONS OF SOCIAL COHESION

Table A.1: Definitions of social cohesion in the academic literature.

Source	Definition of social cohesion			
Chan et al. (2006)	Social cohesion is a state of affairs concerning both the vertical and the horizontal interactions among members of society as characterized by a set of attitudes and norms that includes trust, a sense of belonging and the willingness to participate and help, as well as their behavioural manifestations.			
Maxwell (1996)	Social cohesion involves building shared values and communities of interpretation, reducing disparities in wealth and income, and generally enabling people to have a sense that they are engaged in a common enterprise, facing shared challenges, and that they are members of the same community			
Kearns and Forrest (2000)	A society in which the members share common values which enable them to identify common aims and objectives, and share a common set of moral principles and codes of behaviour through which to conduct their relations with one another			

continues on next page

Source	Definition of social cohesion		
Beauvais and Jenson (2002) (as cited in Fonseca et al., 2019)	On-going process, with known group structures, levels of solidarity and shared values between individuals, and with mechanisms to solve conflict, while arguing that it is comprised of five different dimensions (belonging, inclusion, participation, recognition, and legitimacy).		
Fonseca et al. (2019)	The ongoing process of developing well-being, sense of belonging, and voluntary social participation of the members of society, while developing communities that tolerate and promote a multiplicity of values and cultures, and granting at the same time equal rights and opportunities in society.		
Schiefer and van der Noll (2017)	A descriptive, multifaceted and gradual phenomenon attributed to a collective, indicating the quality of collective togetherness.		
Schmeets (2012)	Cooperative relations among individuals and groups of individuals that are based on mutual recognition, equality and norms of reciprocity.		

continued from previous page

 Table A.2: Definitions of social cohesion in the policy discourse.

Source	Definition of social cohesion
OECD (2011)	A cohesive society works towards the well-being of all its members, fights exclusion and marginalization, creates a sense of belonging, promotes trust, and offers its members the opportunity of upward mobility.
Council of Europe (2010)	The capacity of a society to ensure the well-being of all its members – minimising disparities and avoiding marginalisation – to manage differences and divisions and ensure the means of achieving welfare for all members.

continues on next page

A

Source	Definition of social cohesion
European Commission (2001)	The degree to which individuals and groups within a particular society are bound by common feelings of consensus, share common values and goals and relate to one another on a co-operative basis.
Australian Institute of Health and Welfare (2021)	People's perception of whether most people can be trusted, are fair and most of the time try to be helpful.
Statistics Netherlands (CBS, 2015)	The extent to which people: establish ties or networks with each other and provide each other with help and support; participate in social organizations, such as being a member of associations and organizations and work as a volunteer; and participate in political activities.
Dutch Office of Social and Cultural Planning (Schnabel et al., 2008)	The extent to which people express in behaviour and perception their involvement in social connections in their personal lives, as citizens in society and as members of society.
International and Ibero-American Foundation for Administration and Public Policies (2011)	An attribute of societies which implies equality of opportunity so that people can exercise their fundamental rights and ensure their welfare, without discrimination of any kind and in response to diversity. From an individual perspective, it assumes the existence of people who feel part of a community, participate in decision-making areas and can exercise active citizenship. It also involves the development of public policies and mechanisms of solidarity between individuals, groups, territories and generations.

continued from previous page

B

APPENDIX B

B.1. WIJKPROFIEL DATA

The municipality of Rotterdam makes publicly available the aggregated responses of the Wijkprofiel survey. In this appendix, we will explain the methodology of the Wijkprofiel to aggregate the individual responses into the indicators used in this study (described in Table 3.1).

First, respondents are asked to respond to 5-point Likert scales in the Wijkprofiel survey. Table B.1 shows the questions for the corresponding indicators used in this study. Then, the Wijkprofiel researchers use the percentage that answered the statement positively for the aggregation. Let's take one indicator from *Social cohesion* and another from *Informal support* as examples. Starting with *People in this neighbourhood hardly know each other* (*SC1*), the percentage that answered positively are indexed as follows:

- 1. if range(1, 6) | $SC1 = 0 \rightarrow$ all respondents that answered get 0.
- 2. if range(4, 5) | $SC1 = 1 \rightarrow$ respondents who (totally) disagree get 1.

If the question is formulated in the opposite direction like it is the case of *I live in a nice neighbourhood, where people socialise a lot (SC2)*, then the indexing is done in the opposite direction:

- 1. if range(1, 6) | $SC2 = 0 \rightarrow$ all respondents that answered get 0.
- 2. if range(1, 2) | $SC2 = 1 \rightarrow$ respondents who (totally) agree get 1.

In the case of the indicators for *Informal support*, like *To what extent are you willing to take care of neighbours or friends who need help?* (*IS1*), the indexing is done as follows:

- 1. if range(1, 6) | $IS1 = 0 \rightarrow$ all respondents that answered get 0.
- 2. if range(1, 2) | $IS1 = 1 \rightarrow$ respondents who (already) do get 1.

The (weighted) percentage of respondents with value 1 is then the value presented on the aggregated data set.

Table B.1: Questions in the Wijkprofiel survey associated to the indicators used in the study for each construct.

ļ	Socia	l cohesion (SC)
	Each	item ranked as (1.) Totally disagree, (2.) Disagree, (3.) Nor agree or disagree, (4.) Agree, (5.) Totally agree, and (6.) Don't know / no opinion.
	SC1	People in this neighbourhood hardly know each other.
	SC2	I live in a nice neighbourhood, where people socialise a lot.
	SC3	The residents in this neighbourhood have the same opinions about what can and cannot be done in the neighbourhood.
	SC4	People in this neighbourhood help each other when needed.
	SC5	I feel at home with the people who live in this neighbourhood.
	Infor	mal support ((IS))
	Each	item ranked as (1.) I already do, (2.) Definitely do, (3.) Maybe, (4.) Maybe not, (5.) Not applicable, and (6.) Don't know / no opinion.
	IS1	To what extent are you willing to take care of neighbours or friends who need help?
	IS2	To what extent are you willing to take care of others in your environment you do not know very well and who need help?
	153	To what extent are you willing to take care of relatives in need of help who do not live with you?

B.2. EXPLORATORY FACTOR ANALYSIS

The results of the literature review pointed out that social cohesion is a multi-dimensional concept that can be formed either by means of measurement theory or by synthesis theory. In the study, we conceptualise *Social cohesion* as an emergent variable grounded in synthesis theory. This is because we take a pragmatic strategy of conceiving of these unobservable variables in terms of what is already familiar and well understood. The indicators employed are forged concepts that rely on the revealed principles of measurement theory. Nevertheless, we also explored the possibility of conceptualising social cohesion from measurement theory.

In the measurement theory approach, the theoretical concept comes first, and then it is connected to empirical data. The theoretical concept causes a number of observed variables and their relationships, and the aim is revealing the structure of empirical data. This means that we need to use specific methodologies that can capture the common factor of the underlying dimensions. Exploratory Factor Analysis (EFA), is a technique based on the concept that unobserved constructs underlie the **variation** of observed variables. This technique has been previously used to measure social cohesion as it is the case in Schiefer et al. (2012) and Dragolov et al. (2013), or social resilience as in Larimian et al. (2020).

EFA assumes that the correlation between observed variables is due to latent variables. This correlation is explained by the covariance between the observed variables. In this sense, EFA is a multivariate statistical method that captures the covariance between variables to account for the correlations among observed measurements. For further details on the mechanism of EFA we refer the readers to the extensive literature on the topic (see, for example, Watkins (2021)).

In our intent to use EFA to measure social cohesion, we based the indicators on CBS (2015), the Dutch framework that uses measurement theory using the indicators in Figure 2.4. We combined this with the insights from Schiefer and van der Noll (2017), especially because they differentiate between factors of social cohesion and consequences of it. Table B.2 shows the selected indicators from the Wijkprofiel next to their corresponding dimension, which represents the associated latent construct.

In the following paragraphs, we will explain the steps taken to guarantee the validity of the EFA results, as explained in Watkins (2021). First, EFA assumes that a **linear relationship** exists between the variables. This relationship was explored by means of scatter plots. Variables which did not follow a linear relationship were excluded. Next,

Table B.2: Indicators and their corresponding dimensions for EFA.

Dimension	Indicator
	% of residents who often take the initiative to contact people
	% of residents who say they do not feel abandoned
Social	% of residents who say that local residents know each other ^{a}
potworke	% of residents with weekly family contacts
networks	% of residents with weekly friends contacts
	% of residents with weekly neighbor contacts
	% of residents with weekly contacts with other neighbours
	% of residents who say they are satisfied with their own participation
	% of residents who visit ideological or religious gatherings every month
Socio cultural	% of residents who visit cultural facilities every month
participation	% of residents who visit a hobby club or association every month
participation	% of households that do not use social facilities (WMO, Participation Act, Youth Act)
	% of residents who are actively involved in the neighbourhood
	% of residents who have been involved in making plans for a neighborhood or city
	% of residents who say they trust government organizations
	% of residents who say they have confidence in authorities and care providers
Truct	% of residents who say they have confidence in area government
iiust	% of residents who say they expect progress from the city
	% of residents who say they have confidence in municipal authorities
	% of residents who say they have confidence in the future of the city
	% of residents who say that local residents share each other's views
Mutual	% of residents who say that young and old get along well in the neighbourhood
tolerance	% of residents who say they do not experience discrimination in and outside their own neighbourhood
	$\%$ of residents who say that interaction between ethnic groups in the neighborhood is good b
	% of residents who say they feel at home with local residents ^a
Attachmont	% of residents who say they are proud of the neighbourhood ^{a}
Attachiment	% of residents that say they like the neighborhood ^{a}
	% of residents who say they feel connected to the city ^c
	% of residents who say they feel responsible for their neighbourhood
Responsibility for	% of residents who are active as volunteers
common good	% of residents who have been involved in making plans for a neighbourhood or city
	% of residents that are active in a residents' initiatives
Compliance	Score for perceived victim probability in own neighbourhood
of order	Score for perceived chance of victimization in the neighborhood of someone else in the household ^{a}
of ofder	Score for avoidance behavior ^a
<i>A</i> x x x x	

^a Item dropped due to correlation larger than 0.8
 ^b Item dropped due to non-linearity
 ^c Item dropped due to deviation from normality

117

multicollinearity should be avoided between indicators. Variables with a correlation above 0.8 not acceptable and were excluded. The removal of one item from the highly correlated pairs is based on a qualitative analysis of the items. The likelihood of a multicollinearity problem can be checked by calculating the determinant of the correlation matrix. Multicollinearity is not a problem if the determinant is greater than 0.00001. In our case, the determinant yielded a value of 1.72E-13, which could potentially result in a non-invertible correlation matrix. However, the high correlation between indicators might be due to a strong general factor between all the variables (Gorsuch, 1990), which would be social cohesion in our case. The KMO measure is another way of determining if the correlation matrix is invertible. The KMO value of our data was above the recommended 0.7 threshold, so we increase our confidence that multicollinearity might not be that much of an issue. A second appraisal of the suitability of the correlation matrix is Bartlett's test of sphericity, which 'tests the null hypothesis that the original correlation matrix.' (Field, 2013). The test was significant, which strengthens the idea that there are no strong correlations between variables.

The next step in EFA is to analyse **outliers** and deviation from **normality**. EFA assumes that there is an underlying normal distribution within and between variables. Moderately skewed distributions are acceptable, and as a rule of thumb, skewness above 2 and kurtosis above 7 indicate deviation from normality. Variables which had skewness and kurtosis above the stated values were excluded. In the case of normality between variables, i.e. multivariate normality, the same rule of thumb applies. By means of Mardia's multivariate normality test, we found out that our data had a kurtosis of 786 with p<0.001 and a skewness of 382 with the same p-value. Data in the social sciences are rarely multivariate normally distributed, but this suggests that in our data there is a strong deviation which can affect the results of EFA. The effects of normality deviation strongly depend on the estimation technique used. For example, Maximum Likelihood is strongly affected. Literature suggests that Principal Axis Factoring is the least affected estimation method, so in light of the results of Mardia's test, we will use it in the estimation of our EFA model.

The next step in EFA, is to determine the **number of underlying factors** in our data. We used three different techniques to validate the appropriate number. First, parallel analysis revealed 4 factors as appropriate. Second, the minimum average partial method suggested 5 or 4 factors. The third method is the subjective analysis of the scree plot, which suggested that 4 was the right number of factors.

Before extracting the common factors, the **rotation method** has to be selected. The rotation determines how the variance across factors is distributed. This is analogous to taking a picture from different angles; the object does not change, but the quality of the picture does. On one hand, Ministry for Housing and the Civil Service of the Netherlands (2017) indicates that oblique rotation recognises the high correlation between variables in the social sciences. On the other hand, Schmeets (2012) indicates that orthogonal rotation that has the restriction that the factors cannot be correlated captures better social science variables. For example, people who have frequent contact with family members do not necessarily interact a lot with friends. Consequently, we explored the varimax rotation within the orthogonal rotations and the promax rotation within the oblique rotations, as recommended by Watkins (2021).

Table B.3: EFA results using varimax rotation.

Factor	Indicator	Loading	Corresponding dimension
1	% of residents who say they do not feel abandoned	0.809	Social networks
	% of households that do not use social facilities (WMO, Participation Act, Youth Act)	0.809	Socio-cultural participation
	Score for perceived victim probability in own neighbourhood	0.770	Compliance of order
	% of residents who visit ideological or religious gatherings every month	0.763	Socio-cultural participation
	% residents who say they do not experience discrimination in and outside their own neighbourhood	0.740	Mutual tolerance
	% of residents who say they have confidence in municipal authorities	0.754	Trust
2	% of residents who say they are satisfied with their own participation	0.704	Socio-cultural participation
4	% of residents who say they trust government organizations	0.657	Trust
	% of residents who say they have confidence in area government	0.639	Trust
	% of residents who often take the initiative to contact people	0.831	Social networks
2	% of residents with weekly neighbor contacts	0.814	Social networks
5	% of residents who say they expect progress from the city	0.731	Trust
	% of residents with weekly contacts with other neighbours	0.703	Social networks
	% of residents who are actively involved in the neighbourhood	0.786	Socio-cultural participation
4	% of residents that are active in a residents' initiative	0.776	Socio-cultural participation
	% of residents who are active as volunteers	0.731	Responsibility for common good

Table B.4: EFA results using promax rotation.

Factor	Indicator	Loading	Corresponding dimension
	% of households that do not use social facilities (WMO, Participation Act, Youth Act)	0.928	Socio-cultural participation
	% of residents who visit ideological or religious gatherings every month	0.916	Socio-cultural participation
1	% of residents who say they do not feel abandoned	0.833	Social networks
1	% of residents who visit cultural facilities every month	0.775	Socio-cultural participation
	% residents who say they do not experience discrimination in and outside their own neighbourhood	0.769	Mutual tolerance
	Score for perceived victim probability in own neighbourhood	0.762	Compliance of order
-	% of residents who say they are satisfied with their own participation	0.871	Socio-cultural participation
2	% of residents who say they have confidence in municipal authorities	0.823	Trust
2	% of residents who say they have confidence in area government	0.725	Trust
	% of residents who say they trust government organizations	0.646	Trust
	% of residents who often take the initiative to contact people	0.902	Social networks
2	% of residents with weekly neighbor contacts	0.833	Social networks
3	% of residents who say they expect progress from the city	0.709	Trust
	% of residents with weekly contacts with other neighbours	0.644	Social networks
	% of residents that are active in a residents' initiative	0.851	Socio-cultural participation
4	% of residents who are actively involved in the neighbourhood	0.824	Socio-cultural participation
4	% of residents who have been involved in making plans for a neighborhood or city	0.730	Responsibility for common good
	% of residents who are active as volunteers	0.726	Responsibility for common good

Finally, the **factor loading threshold** should be chosen before extraction. After exploring our data, we have seen that multivariate normality is not met, and that there is still presence of multicollinearity between variables. Therefore, we decided to pick a more strict threshold that satisfies significance to 99% CI, such that

Threshold =
$$\frac{5.152}{\sqrt{N-2}}$$

with 70 neighbourhoods in the Wijkprofiel, this yields a threshold for the factor loadings of 0.62.

With all this, the extraction results for varimax rotation can be seen in Table B.3 and results for promax rotation in Table B.4. The extraction was done using the software SPSS by IBM (IBM Corp., 2020). The end goal in EFA is to find the common factors between a set of variables. We can see that the **factors are entangled and are not clearly identi-fiable**. This means that EFA was not successful in extracting clear common factors for any of the rotations. This can be the result of different reasons. First, the presence of high multicollinearity even after the screening of variables can confound the meaning of the extracted factors. Most trust indicators were highly collinear with the other indicators. This suggests the possibility that trust is also a consequence of the other factors. Second, the little amount of data also plays a large role in the final outcome. We have 70

data points, and the minimum recommended is of 10 points per factor indicator. Third, after the screening, some dimensions had missing indicators, which can confound the meaning of the factors. Finally, defining social cohesion from measurement theory is tricky as it is a catch-all concept hard to translate.

Overall, the results from EFA were not satisfactory as no clearly identifiable factors were extracted. Consequently, we argue that for this study, synthesis theory is a better way to conceptualise social cohesion.

C

APPENDIX C

The aim of the study was to assess the capabilities of balanced neighbourhoods to promote resilient actions between neighbourhood residents. As such, *Informal support* was defined as the willingness to help of friends and neighbours. In addition to this indicator, Table C.1 shows two other indicators the Wijkprofiel provides under the section for care giving. To explore the effects in the results of employing the other indicators, this chapter shows the results of applying the same model to different conceptualisations of informal support.

C.1. INFORMAL SUPPORT AS THE COMMON FACTOR BETWEEN WILLINGNESS TO HELP FRIENDS AND NEIGHBOURS, AND TO HELP OTHERS

Informal support was conceptualised and estimated as a latent construct from the % of residents who say they are willing to care for neighbours or friends who need help (IS1) and the % of residents who say they are willing to care for others in the area who need help (IS2). This conceptualisation is of interest as it expands the definition of the main results to include support to invisible or non-existent ties. As such, it can assess whether balanced neighbourhoods foster informal support in the different social spheres. The PLS-SEM algorithm could not find a common factor between the two indicators, as they are not correlated ($r^2 = 2.1 \times 10^{-5}$).

Table C.1: Characteristics of the indicators for care giving in the Wijkprofiel in 2019. N indicates the number of neighbourhoods included in the Wijkprofiel. Source: OBI.

Cons	structs and indicators	N	Min/Max	Mean	s.d.
Info	rmal support				
IS1	% of residents who say they are willing to care for neighbors or friends who need help	70	45/71	57	6
IS2	% of residents who say they are willing to care for others in the area who need help	70	9/36	22	4
IS3	% of residents who say they are willing to care for relatives who need help	70	55/82	69	5

C.2. INFORMAL SUPPORT AS WILLINGNESS TO HELP OTHERS

Informal support was conceptualised and estimated only as the % of residents who say they are willing to care for others in the area who need help (IS2). This conceptualisation differs from the one in the main results as it does not take into account forms of support to strong or weak ties, but it is still of relevance to find out whether balanced neighbourhoods foster support only between invisible or non-existent ties. However, the PLS-SEM algorithm did not yield any model with an acceptable fit (SRMR<0.08) for any distribution of balance.

C.3. INFORMAL SUPPORT AS THE COMMON FACTOR BETWEEN WILLINGNESS TO HELP RELATIVES, FRIENDS AND NEIGHBOURS, AND OTHERS

The theory of this study argues that a change towards balanced neighbourhoods can trigger resilient actions between the residents of that neighbourhood. This theory is based on the grounds that geographically connected people become affected by the social mix in the area to promote solidarity. As such, the mechanism that drives help to relatives is other than the balance of a neighbourhood. Nevertheless, an existing effect could indicate that geographically bounded effects like neighbourhood balance can promote geographically unbounded actions like willingness to help relatives.

With this premise, *Informal support* was conceptualised and estimated as the common factor of the % of residents who say they are willing to care for neighbours or friends who need help (IS1), the % of residents who say they are willing to care for others in the area who need help (IS2), and the % of residents who say they are willing to care for relatives who need help (IS3). This conceptualisation differs from the one in the main results as it takes into account all forms of support that are available in the Wijkprofiel, strong, weak ties, and invisible or non-existent ties.

Out of 3,162 models — the total number of possible integer configurations for the distribution of balance—, 127 ($\approx 4\%$) met the SRMR<0.08 criterion (see Figure C.1). In this case, the elbow method recommended 4 different clusters (Figure C.2).

C.3.1. PLS-SEM MODEL RESULTS

The same procedure as in the main results was applied, so outcomes are simplified in the appendix.

MODEL GOODNESS-OF-FIT

All the models meet the 99% confidence interval (CI) quantile criteria for all the goodnessof-fit measures (see Table C.2).

REFLECTIVE MEASUREMENT MODEL

Examining the internal consistency reliability is the first step in evaluating a reflective measurement model. For that, Dijkstra-Henseler's rho, ρ_A , should lie between 0.70 and 0.95 to guarantee a good level of internal consistency that avoids indicator redundancy



Figure C.1: House stock distributions within the balance definition of Rotterdam that have an acceptable goodness-of-fit ($\approx 4\%$).



Figure C.2: Clusters of the house stock distributions that meet the definition of balance from the municipality and that have an acceptable goodness-of-fit. Clusters 1, 2, and 4 exhibit concave shapes, while Cluster 3 has a convex shape.

Table C.2:	Goodness-of-fit	of the estimated	models.
------------	-----------------	------------------	---------

	Model 1 Value 99% CI		Model 2 Value 99% CI		Model 3		Model 4	
					Value 99% CI		Value	99% CI
dG	0.010	0.029	0.010	0.032	0.002	0.034	0.010	0.028
SRMR	0.037	0.061	0.037	0.061	0.016	0.065	0.038	0.057
dL	0.014	0.037	0.014	0.037	0.003	0.042	0.015	0.033
dML	0.050	0.145	0.051	0.155	0.008	0.166	0.055	0.140



Figure C.3: Housing stock distributions of the models with best goodness-of-fit for each identified cluster.
	aggIS1	aggIS2	aggIS3
aggIS1	1.000	0.005	0.572
aggIS2	0.005	1.000	-0.283
aggIS3	0.572	-0.283	1.000

Table C.3: Correlation matrix of Informal support indicators.

Table C.4: Reflective construct reliability assessment.

	Model 1		Мос	lel 2	Мос	lel 3	Model 4		
	ρ_A	AVE	$ ho_A$	AVE	$ ho_A$	AVE	$ ho_A$	AVE	
Informal Support	0.75	0.61	0.75	0.61	0.74	0.59	0.75	0.61	

(Dijkstra & Henseler, 2015). The construction of *Informal support* with the three indicators yielded a ρ_A under this threshold for all models, therefore, to increase internal consistency reliability, *IS2* was dropped because it had the lowest loading in all instances. As a result, ρ_A increased to above 0.70 and lies within the accepted threshold for all models (Table C.4). The % of residents who say they are willing to care for relatives who need help (*IS3*) is correlated with the % of residents who say they are willing to care for neighbours or friends who need help (*IS1*), however, they are both lowly correlated to the % of residents who say they are willing to care for others in the area who need help (*IS2*) (see Table C.3).

The AVE of *Informal support* is higher than the accepted value of 0.50, which indicates that the construct explains at least 50% of the variance of its items and a dominant factor could be extracted in all models (Henseler, 2020). Finally, Table C.5 shows that *IS3* had a loading above 0.7 in all models, while *IS1* did not reach that threshold but was above 0.6 in all instances. This means that *IS3* is more closely related and explains more of the variance of the extracted latent variable *Informal support* than *IS1*.

FORMATIVE MEASUREMENT MODEL

The model goodness-of-fit with an SRMR below 0.08 within the 99% CI (see Table C.2) indicates the confirmatory power of our emergent construct and is suggestive of its reliability and validity (Henseler, 2020).

When applying Mode BNNLS to Model 3, the results suggested that *Social cohesion* can be fully captured by *SC5*. On the other hand, when applied to Models 1, 2, and 4, the outcomes showed that *Social cohesion* was a composite of *SC3* and *SC5*. Nevertheless, the weight of *SC5* was not statistically significant, thus in those Models *Social cohesion* can be fully captured by *SC3*. These results are shown in Table C.6.

		Model 1		М	odel 2	М	odel 3	Model 4		
	Item	Loading	95% CI							
Informal	IS3	0.911	[0.591, 0.993]	0.911	[0.569, 0.991]	0.840	[0.632, 0.981]	0.906	[0.600, 0.987]	
Gummont	IS1	0.628	[0.522, 0.940]	0.628	[0.480, 0.936]	0.681	[0.496, 0.874]	0.632	[0.495, 0.954]	
support	IS2 ^a	NA	NA	NA	NA	NA	NA	NA	NA	

a: Item dropped in order to increase internal consistency reliability

Table C.6: Formative construct assessment.

		Mode	ls 1, 2, and 4	Ν	Iodel 3
	Item	Weight	95% CI	Weight	95% CI
	SC1	NA ^a	NA ^a	NA ^a	NA ^a
Social	SC2	NA^{b}	NA^b	NA^{b}	NA^b
	SC3	1.000	[1.000, 1.000]	NA ^a	NA^{a}
Collesion	SC4	NA^{b}	NA^b	NA^b	NA^b
	SC5	NA^b	NA^b	1.000	[1.000, 1.000]
Neighbourhood	NR	1 000	[1,000, 1,000]	1 000	[1 000 1 000]
Balance	ND	1.000	[1.000, 1.000]	1.000	[1.000, 1.000]

a: Item dropped due to high multicollinearity

b: Item dropped due to insignificant weight.

Table C.7: Assessment of the structural model reliability.

	_		Model	1		Model	2		Model	3	Model 4		
Dependent variable	Independent variable	Adj. R ²	VIF	f ²	Adj. R ²	VIF	f ²	Adj. R ²	VIF	f ²	Adj. R ²	VIF	f ²
Informal		0.16			0.16			0.24			0.16		
Support		0.10			0.10			0.24			0.10		
	Social Cohesion		1.19	0.184		1.20	0.185		1.11	0.323		1.21	0.195
	Neighbourhood Balance		1.19	0.000		1.20	0.000		1.11	0.000		1.21	0.000
Social Cohesion		0.15			0.16			0.09			0.16		
	Neighbourhood Balance			0.194			0.203			0.115			0.215

STRUCTURAL MODEL

After guaranteeing that the measurement model is satisfactory, the structural model results were assessed. First, Table C.7 shows that the VIF between the constructs is below 2 in all instances, so we can discard any multicollinearity issues.

Second, the adjusted coefficient of determination (Adjusted R² or Adj. R²) was considered. All four models showed a weak explanatory power for the construct of *Social cohesion*, with Model 3 providing the least. On the other hand, Model 3 showed a stronger power when explaining *Informal support* compared to the other models, although it still ranged within the weak explanatory power range (Table C.7).

A third consideration to evaluate the PLS-SEM model was to calculate Cohen's effect size (f²). Table C.7 shows that the effect size of *Social cohesion* on *Informal support* is medium for Models 1, 2, and 4, and large for Model 3. Second, *Neighbourhood balance* had a very small effect size in all Models. Third, the effect size of *Neighbourhood balance* on *Social cohesion* is medium to large in Models 1, 2, and 4, and medium to small in Model 3.

The next step to evaluate the predictive power of the structural model was to calculate the predictive relevance (Q^2) of the construct items (Hair et al., 2019). Table C.8 shows how *SC3* has a small to medium predictive power for *Social cohesion*, whereas *SC5* lacks accuracy. Regarding *Informal support*, Models 1 and 3 lack predictive power, whereas Models 2 and 4 have a small predictive power from the hand of *IS3* and lack of it from *IS1*. In addition, Table C.8 shows how almost all indicators have a higher pre-

		Model 1				Model 2			Model 3			Model 4		
	Item	Δ MAE	Δ RMSE	Q^2	Δ MAE	Δ RMSE	\mathbf{Q}^2	Δ MAE	Δ RMSE	Q^2	Δ MAE	Δ RMSE	Q^2	
	SC1	NA ^a												
Social	SC2	NA^{b}	NA ^b	NA^{b}	NA^{b}									
Cohosion (SC)	SC3	-0.012	-0.020	0.121	0.002	0.001	0.147	NA ^a	NA ^a	NA ^a	0.002	0.001	0.147	
Collesion (SC)	SC4	NA^{b}												
	SC5	NA^{b}	NA^{b}	NA^{b}	NA^{b}	NA^{b}	NA^{b}	-0.113	-0.123	-0.020	NA^{b}	NA^{b}	NA^{b}	
Informal	IS3	-0.119	-0.182	-0.001	-0.047	-0.069	0.008	-0.154	-0.182	-0.059	-0.047	-0.069	0.008	
finormant (IC)	IS1	-0.097	-0.147	-0.026	-0.043	-0.057	-0.020	-0.117	-0.144	-0.017	-0.043	-0.057	-0.020	
Support (IS)	IS2 ^c	NA												

Table C.8: Assessment of the predictive power of the structural model.

a: Item dropped due to high multicollinearity

b: Item dropped due to insignificant weight.

c: Item dropped in order to increase internal consistency reliability

	Table (C.9: Structural	model total,	direct,	and indirect	path coefficients.
--	---------	-----------------	--------------	---------	--------------	--------------------

	_												
			Model 1			Model 2			Model 3	3		Model 4	
Dependent	Independent	β	β	β	β	β	β	β	β	β	β	β	β
variable	variable	Total	Direct	Indirect	Total	Direct	Indirect	Total	Direct	Indirect	Total	Direct	Indirect
Informal													
Support													
	Social	0 40 4*	0 40 4*		0.400*	0.400*		0.510*	0 510*		0.420*	0.420*	
	Cohesion	0.424*	0.424*		0.426*	0.426*		0.516*	0.516		0.439*	0.439*	
	Neighbourhood												
	Balance	-0.183	-0.012	-0.171	-0.181	-0.006	-0.175***	0.151	-0.014	0.166	-0.162	0.022	-0.185
Social													
Cohesion													
	Neighbourhood	0 1000											
	Balance	-0.403*	-0.403*		-0.411*	-0.411*		0.321	0.321		-0.421*	-0.421*	
Cimificanas	lougles \$p +0.001 **p	0.01 **** 0.0	05										

Note: No superscript indicates statistically non-significant values.

dicted error compared to the benchmark, so this is suggestive that the model has a low predictive power. Only Social cohesion in Models 2 and 4 has a smaller predicted error.

Finally, after verifying the PLS-SEM model's explanatory and predictive power, the final step is to assess the statistical significance and relevance of the path coefficients (see Table C.9). The results of the effects can be summarised as follows. First, in Models 1, 2, and 4, Neighbourhood balance has a significant negative association with Social cohesion. Second, the direct relationship between Neighbourhood balance and Informal support is non-significant. There is, however, the existence of a slight indirect effect brought by the mediation effect of *Social cohesion* in Model 2. Finally, we see a positive association between Social cohesion and Informal support.

We can see different results for Model 3. Here, *Neighbourhood balance* is positively associated with Social cohesion, but to the 90% CI. As in the previous models, the relationship between Neighbourhood balance and Informal support is non-significant, but here there is not even a slight indirect effect brought by Social cohesion. The association between Social cohesion and Informal support is again positive.

Regarding the mediation effects, we can see that the direct effect between Neighbourhood balance and Informal support is non-significant in Models 1 and 3, so we can talk about full mediation. The effects of balance on willingness to help are all due to social cohesion. In Models 2, the relationship between Neighbourhood balance and Informal support is significant to the 95% CI, so we talk about complementary partial mediation (Henseler, 2020).

D

APPENDIX D

This appendix includes additional information to the report results. First, it shows the assessment of the explanatory and predictive power of the PLS-SEM model. Then it shows the value of the outliers not included in Figure 4.8, and finally the results from spatial exploration by means of the LISA.

D.1. PLS-SEM STRUCTURAL MODEL

First, to guarantee that the results of the regression equations are not biased, multicollinearity between the constructs was considered. Using the reflective construct *Informal support* as dependent variable, we calculated the VIF. Table D.1 shows that the VIF between the constructs is below 2 in all instances, so we can discard any multicollinearity issues.

Second, the adjusted coefficient of determination (Adjusted R^2 or Adj. R^2) was considered. The Adjusted R^2 is the most commonly used measure to evaluate the measurement model and indicates the model's predictive accuracy and explanatory power. The values are context dependent, however, Hair et al. (2019) suggests that values of 0.25, 0.50 and 0.75 can be considered as weak, moderate, and substantial. All four models showed a weak explanatory power for the construct of *Social cohesion*, with Model 3 providing the least. On the other hand, Model 3 showed a stronger power when explaining *Informal support* compared to the other models, although it still ranged within the weak explanatory power range (Table D.1). In this interpretation, it is important to consider that Hair et al. (2019) highlights how Adjusted R^2 values that predict human attitudes like those in our study are usually lower than for physical processes.

A third consideration to evaluate the PLS-SEM model was to calculate Cohen's effect size (f^2). f^2 indicates the change in the R^2 value when a specified exogenous construct is removed from the model. Given that R^2 is context dependent, Cohen (1988) suggested that values higher than 0.02, 0.15 and 0.35 depict small, medium, and large f^2 effect size. First, Table D.1 shows that the effect size of *Social cohesion* on *Informal support* is medium for all Models. Second, *Neighbourhood balance* had a very small effect

		Model 1				Model	2		Model 3		
Dependent variable	Independent variable	Adj. R ²	VIF	\mathbf{f}^2	Adj. R ²	VIF	\mathbf{f}^2	Adj. R ²	VIF	\mathbf{f}^2	
Informal		0.07			0.08			0.09			
Support		0.07			0.00			0.05			
	Social Cohesion		1.28	0.108		1.20	0.114		1.11	0.107	
	Neighbourhood Balance		1.28	0.008		1.20	0.011		1.11	0.001	
Social		0.21			0.16			0.00			
Cohesion		0.21			0.10			0.09			
	Neighbourhood Balance			0.279			0.204			0.115	

Table D.1: Assessment of the structural model reliability.

size in all Models. Third, the effect size of *Neighbourhood balance* on *Social cohesion* is medium to large in Models 1, and 2, and medium to small in Model 3.

The main takeaway of these results is that the % of residents who say that local residents share each other's views (SC3) is better at explaining Social cohesion, but that SC5 captures a stronger dependency between Social cohesion and Informal support within our model, especially because it has a weaker association with Neighbourhood balance. Finally, Model 1 is considered superior to Models 2 and 3 as it presented the best goodnessof-fit.

The next step to evaluate the predictive power of the structural model was to calculate the predictive relevance (Q^2) of the construct items (Hair et al., 2019). In a nutshell, the value for Q^2 results from the difference between the true data points and the predicted ones. As a rule of thumb, Q^2 values higher than 0, 0.25 and 0.50 depict small, medium, and large predictive relevance. Values smaller than zero for all indicators of a construct suggest a lack of predictive accuracy for that construct. Table D.2 shows how Models 1, and 2 have a small to medium predictive power for *Social cohesion*, whereas Model 3 which specifies *Social cohesion* as the % of residents who say they feel at home with local residents (SC5) has lower accuracy. Regarding *Informal support*, all Models lack predictive power.

The assessment of the predictive accuracy of the structural model is finalised with the calculation of the prediction error. If Q^2 indicated the presence of predictive power (only for *Social cohesion* in our Models), then the difference between mean absolute error (Δ MAE) and the root mean squared error (Δ RMSE) of the predictions and a benchmark quantify the predictive power. The recommended benchmark is the most naïve and corresponds to a linear regression model (Hair et al., 2019). If the PLS-SEM prediction yields higher RMSE or MAE for all indicators compared to the naïve benchmark, this means that the model lacks predictive accuracy (positive Δ RMSE and Δ MAE). Table D.2 shows how Model 1 has a higher predicted error compared to the benchmark for *Social cohesion*, so this is suggestive that the model has a low predictive power. In Models 2, and the values are negative, which suggest the presence of higher predictive power.

	_		Model 1			Model 2			Model 3	
	Item	Δ MAE	Δ RMSE	\mathbf{Q}^2	Δ MAE	Δ RMSE	\mathbf{Q}^2	Δ MAE	Δ RMSE	\mathbf{Q}^2
	SC1	NA ^a	NA^{a}	NA ^a	NA ^a	NA^{a}	NA ^a	NA ^a	NA ^a	NA ^a
Social	SC2	NA^b	NA^b	NA^{b}	NA^b	NA^{b}	NA^b	NA^b	NA^{b}	NA^{b}
Cohesion	SC3	0.001	0.001	0.184	-0.013	-0.022	0.139	NA^b	NA^{b}	NA^{b}
Collesion	SC4	NA^{b}	NA^{b}	NA^{b}	NA^{b}	NA^{b}	NA^{b}	NA^{b}	NA^{b}	NA^{b}
	SC5	NA ^b	NA^b	NA^b	NA ^b	NA^b	NA ^b	-0.114	-0.126	0.008
Informal Support	IS1	-0.046	-0.069	-0.032	-0.099	-0.152	-0.043	-0.117	-0.145	-0.020

Table D.2: Assessment of the predictive power of the structural model.

a: Item dropped due to high multicollinearity

b: Item dropped due to insignificant weight.

 Table D.3:
 Standardised Neighbourhood balance of the outliers not included in Figure 4.8 for visibility purposes.

	Model 1	Model 2	Model 3
Noord Kethel	-3.93	-3.51	1.27
Strand en Duin	-3.71	-4.15	-3.71
Nesselande	-3.42	-3.46	-1.21

D.2. OUTLIERS

Table D.3 shows the value of the standardised *Neighbourhood balance* of outliers that were not included in Figure 4.8.

D.3. LISA RESULTS

D.3.1. LISA OF Social cohesion

To further explore the spatial distribution of *Social cohesion*, Figure D.1 shows its LISA classification. We can see a cluster of neighbourhoods with high *Social cohesion* surrounded by other neighbourhoods with high *Social cohesion* in the North of Rotterdam, and the opposite in the South. Model 3 created larger clusters and more polarised. The other Models showed a large HH cluster in the North, but several small LL clusters. In addition, they identified an HL cluster in the centre of the city, which could be brought by the fact that the river was not taken into account in the spatial model. The presence of HH and LL clusters is suggestive evidence about possible spillover or contagion effects. *Social cohesion* is not administratively bounded, so the interaction between residents of adjacent neighbourhoods can influence the level of cohesion of others.

D.3.2. LISA OF Informal support

In order to obtain a more nuanced representation of the relevance of the geographical distribution, Figure D.2 shows the LISA clusters for *Informal support*. In the first figure, we can see that all models identify two clear clusters divided by their geographical location. In Southern neighbourhoods, citizens are less willing to help each other and are surrounded by neighbourhoods where the willingness is also low, while the opposite happens in the North-East of the city.

In addition, Figure D.3 shows the LISA choropleth of Informal support and its associ-



Figure D.1: LISA choropleth of *Social cohesion*. HH indicates neighbourhoods with high *Social cohesion* surrounded by neighbourhoods with high *Social cohesion*. HL indicates neighbourhoods with high *Social cohesion* surrounded by neighbourhoods with low *Social cohesion*. LH indicates neighbourhoods with low *Social cohesion* surrounded by neighbourhoods with high *Social cohesion*. LL indicates neighbourhoods with low *Social cohesion* surrounded by neighbourhoods with high *Social cohesion*. LL indicates neighbourhoods with low *Social cohesion* surrounded by neighbourhoods with low *Social cohesion*. Is are neighbourhoods which do not belong to any of the LISA clusters, as results were non-significant. LISA clusters significant to 95% confidence interval.

ation with the average *Social cohesion* around a neighbourhood. We can see that Models 1, and 2 identified smaller clusters than Model 3. Again, this is due to a stronger polarisation in *Social cohesion* brought by the % *of residents who say they feel at home with local residents* (*SC5*). Of special attention are the neighbourhoods in the centre of Rotterdam, which show a mix of LL and HL LISA clusters and bring into picture the uniqueness of the centre in comparison to the rest of neighbourhoods.



Figure D.2: LISA choropleth of *Informal support*. HH indicates neighbourhoods with high *Informal support* surrounded by neighbourhoods with high *Informal support*. HL indicates neighbourhoods with high *Informal support* surrounded by neighbourhoods with low *Informal support*. LH indicates neighbourhoods with low *Informal support*. LL indicates neighbourhoods with low *Informal support*. Is are neighbourhoods with low *Informal support*. It is not support to any of the LISA clusters, as results were non-significant. LISA clusters significant to 95% confidence interval.



Figure D.3: LISA choropleth of *Informal support* and *Social cohesion*. HH indicates neighbourhoods with high *Informal support* surrounded by neighbourhoods with high *Social cohesion*. HL indicates neighbourhoods with high *Informal support* surrounded by neighbourhoods with low *Social cohesion*. LH indicates neighbourhoods with high *Informal support* surrounded by neighbourhoods with low *Social cohesion*. LH indicates neighbourhoods with low *Informal support* surrounded by neighbourhoods with high *Social cohesion*. LL indicates neighbourhoods with low *Informal support* surrounded by neighbourhoods with high *Social cohesion*. LL indicates neighbourhoods with low *Informal support* surrounded by neighbourhoods with low *Social cohesion*. LL indicates neighbourhoods with low *Informal support* surrounded by neighbourhoods with low *Social cohesion*. LL indicates neighbourhoods with low *Informal support* surrounded by neighbourhoods with low *Social cohesion*. LL indicates neighbourhoods with low *Informal support* surrounded by neighbourhoods with low *Social cohesion*. LL indicates neighbourhoods with low *Informal support* surrounded by neighbourhoods with low *Social cohesion*. It indicates neighbourhoods which do not belong to any of the LISA clusters, as results were non-significant. LISA clusters significant to 95% confidence interval.

E

APPENDIX E

The main results of the research assume that the river that divides Rotterdam has no consequences on the spatial analysis, i.e. neighbouring neighbourhoods across the river have a spatial weight (Equation (3.8)) equivalent to 1. This appendix shows the same results by considering that neighbourhoods on both sides of the river are not neighbouring, with a spatial weight equivalent to a 0 for the neighbouring neighbourhoods across the river. This is equivalent to assuming that characteristics of neighbourhoods do not have an effect across the river.

E.1. SPATIAL EFFECTS IN SOCIAL COHESION TAKING INTO ACCOUNT RIVER EFFECTS

The LISA choropleth of *Social cohesion* considering that the river does not allow the spillover of neighbourhood effects (see Figure E.1 shows a stronger polarisation in comparison to the LISA choropleth of the main results. We see the new presence of HH clusters towards the North of the city and of LL clusters towards the South in all models. In addition, the city centre also shows a new LL cluster which highlights the low *Social cohesion* characteristic of busy downtowns.

The identification of these clusters can be understood with the results from the Durbin model in Table E.1. We can see that the spillover or contagion effect is smaller when taking into account the river given the spatial segregation which does not allow spill over effects between South and North. The effects are still significant.

E.2. Spatial effects in Informal support taking into account river effects

The LISA choropleth of *Informal support* considering that the river does not allow the spillover of neighbourhood effects (see Figure E.2 shows a stronger polarisation in comparison to the LISA choropleth of the main results. We see the new presence of large HH



Figure E.1: LISA choropleth of *Social cohesion* taking into account river effects. HH indicates neighbourhoods with high *Social cohesion* surrounded by neighbourhoods with high *Social cohesion*. HL indicates neighbourhoods with high *Social cohesion* surrounded by neighbourhoods with low *Social cohesion*. LH indicates neighbourhoods with low *Social cohesion* surrounded by neighbourhoods with high *Social cohesion*. LL indicates neighbourhoods with low *Social cohesion* surrounded by neighbourhoods with high *Social cohesion*. LL indicates neighbourhoods with low *Social cohesion* surrounded by neighbourhoods with high *Social cohesion*. LL indicates neighbourhoods with low *Social cohesion* surrounded by neighbourhoods with low *Social cohesion*. Is are neighbourhoods which do not belong to any of the LISA clusters, as results were non-significant. LISA clusters significant to 95% confidence interval.

clusters towards the North of the city and of large LL clusters towards the South in all models.

In addition, Figure E.3 shows the LISA choropleth of *Informal support* and its association with the average *Social cohesion* around a neighbourhood considering the presence of the river. Once again, we can see the emergence of HH clusters in the North ans LL clusters in the South. Notably, the city centre shows a larger HL cluster than previously considered, showing the presence of a gradient in *Social cohesion* from South to North, but the presence of a higher *Informal support* in the city centre.

As in the case of *Social cohesion*, we can see from the Durbin model in Table E.2 that the spillover or contagion effect is smaller when taking into account the river given the spatial segregation which does not allow spill over effects between South and North. The effects are still significant.

Variable	Model 1	Model 2	Model 3	
Intercept	-0.026	-0.028	-0.061	
	(0.098)	(0.101)	(0.103)	
Neighbourhood Balance	-0.458*	-0.404*	0.234***	
	(0.101)	(0.103)	(0.104)	
Lagged Neighbourhood Balance	-0.242	-0.178	0.208	
	(0.171)	(0.166)	(0.164)	
Lagged Social Cohesion	0.341**	0.382**	0.390**	
	(0.132)	(0.128)	(0.127)	
Log-likelihood	-81.466	-83.727	-84.765	
Akaike information criterion	170.933	175.454	177.530	
R ² (pseudo)	0.375	0.340	0.2902	
Significance levels: *p<0.001 **p<0.01 ***p<0.05				

Table E.1: Results of the Durbin spatial model with *Social cohesion* as dependent variable taking into account river effects. Lagged variables are the weighted averages of neighbouring values. Significant values of the lagged variables indicate spillover or contagion effects. In brackets, the standard deviations.

Note: No superscript indicates statistically non-significant values.

Table E.2: Results of the Durbin spatial model with *Informal support* as dependent variable taking into account river effects. Lagged variables are the weighted averages of neighbouring values. Significant values of the lagged variables indicate spillover or contagion effects. In brackets, the standard deviations.

Variable	Model 1	Model 2	Model 3
Intercept	-0.010	-0.011	0.000
	(0.110)	(0.109)	(0.107)
Neighbourhood Balance	0.083	0.086	-0.028
	(0.126)	(0.121)	(0.113)
Lagged Neighbourhood Balance	0.136	0.144	0.084
	(0.208)	(0.189)	(0.163)
Social Cohesion	0.360**	0.359**	0.384**
	(0.135)	(0.131)	(0.128)
Lagged Social Cohesion	-0.120	-0.121	-0.147
	(0.225)	(0.216)	(0.184)
Lagged Informal Support	0.324***	0.317***	0.301***
	(0.136)	(0.137)	(0.139)
Log-likelihood	-87.525	-87.434	-86.308
Akaike information criterion	187.051	186.867	184.616
R ² (pseudo)	0.204	0.205	0.227

Significance levels: *p<0.001 **p<0.01 ***p<0.05

Note: No superscript indicates statistically non-significant values.



Figure E.2: LISA choropleth of *Informal support* taking into account river effects. HH indicates neighbourhoods with high *Informal support* surrounded by neighbourhoods with high *Informal support*. HL indicates neighbourhoods with high *Informal support* surrounded by neighbourhoods with low *Informal support*. LH indicates neighbourhoods with low *Informal support* surrounded by neighbourhoods with high *Informal support*. LH indicates neighbourhoods with low *Informal support* surrounded by neighbourhoods with high *Informal support*. LH indicates neighbourhoods with low *Informal support* surrounded by neighbourhoods with high *Informal support*. LL indicates neighbourhoods with low *Informal support* surrounded by neighbourhoods with low *Informal support*. Is are neighbourhoods which do not belong to any of the LISA clusters, as results were non-significant. LISA clusters significant to 95% confidence interval.



Figure E.3: LISA choropleth of *Informal support* and *Social cohesion* taking into account river effects. HH indicates neighbourhoods with high *Informal support* surrounded by neighbourhoods with high *Social cohesion*. HL indicates neighbourhoods with high *Informal support* surrounded by neighbourhoods with low *Social cohesion*. LH indicates neighbourhoods with low *Informal support* surrounded by neighbourhoods with high *Social cohesion*. LL indicates neighbourhoods with low *Informal support* surrounded by neighbourhoods with high *Social cohesion*. LL indicates neighbourhoods with low *Informal support* surrounded by neighbourhoods with high *Social cohesion*. LL indicates neighbourhoods with low *Informal support* surrounded by neighbourhoods with low *Social cohesion*. LL indicates neighbourhoods with low *Informal support* surrounded by neighbourhoods with low *Social cohesion*. LL indicates neighbourhoods with low *Informal support* surrounded by neighbourhoods with neighbourhoods with low *Social cohesion*. LL indicates neighbourhoods with low *Informal support* surrounded by neighbourhoods with neighbourhoods wi

F

APPENDIX F

This study takes a confirmatory approach, so elements hard to quantify are disregarded. Moderation analysis results indicate that none of the built environment or land use elements moderate the relationships between *Neighbourhood balance* and *Social cohesion* or *Neighbourhood balance* and *Informal support*. In addition, none of them are directly associated with *Social cohesion* and *Informal support*. However, we have seen in Section 2.2.4 that urban elements have been previously related to social cohesion. We therefore argue that our quantitative approach might not be capable to capture the effects that urban design elements have on the experience and perceptions of residents (Kim & Kaplan, 2004; Rogers & Sukolratanametee, 2009; Wood et al., 2010).

In this appendix, urban design elements that can have an effect on the sense of community are described by means of image deconstruction. To deconstruct means 'to unpack through analysis, but it also means to show the different connections that allow us to understand how something might take shape in another way' (Photographie, 2016). All images were taken on June 12^{th} , 2022, between 3pm and 7pm. This appendix does not intend to be a rigorous qualitative analysis, but an opening door to future research on the effects of urban design on social cohesion.

F.1. IMAGE DECONSTRUCTION OF URBAN DESIGN ELEMENTS

Figure F1 compares special features of public spaces from selected neighbourhoods in the North and South of Rotterdam. We can see that Figure F1a is inclusive for a larger age range: it has a kindergarten for the smallest and a larger sitting possibility for the eldest. Furthermore, it also provides a space for shopping which increases the possibilities of interaction and concentration of people (Mouratidis & Poortinga, 2020). The building elements on Figure F1a promote the interaction between people while on Figure F1b they encourage a sense of privacy. In that regard, the space in Figure F1b could be labelled as semi-public (Peterson, 2017). We can see the presence of determined borders from the semi-public inside the park and the public street space. Having a group occupying the space in Figure F1b would discourage other people from utilizing it. The space

on Figure E1a provides a more accessible transition from biking to walking into the public space while the other space has a green barrier. This could increase the interaction in Figure E1b by forcing people to share the same entrance to the space (Wood et al., 2010). Finally, the space on Figure E1a is provided with more lighting than the other. This allows people to spend a longer time in the area while in Figure E1b people are dispersed when the sun goes down (Wood et al., 2010).

Figure E2 compares special features of the public-private interplay from the same selected neighbourhoods in the North and South of Rotterdam. The comparison between these two images shows the possible conflict from private to public space. In Figure E2a we can see that the recreational area is in front of small buildings. This creates a closer connection from the private to the public sphere (Garrido-Velarde et al., 2018). On the other hand, Figure E2b shows that the buildings surrounding the park are taller, thus increasing the separation from the park. Next, the park on Figure E2a is surrounded by a fence: on one hand, this creates a boundary between the street and the park, on the other, it also constraints the entrance to certain points. This single points of circulation can promote interaction between people (Rogers & Sukolratanametee, 2009; Wood et al., 2010). The presence of a fence can also emit a sense of danger and discourage people from using or else provide a feeling of safety and encourage the use (Park et al., 2014). Finally, on Figure E2b, we can see that there is wider space, which could suggest a reduction in interaction between the park users (Mouratidis & Poortinga, 2020).

Figure E3 compares special features of the buildings' facades and layout from the same selected neighbourhoods in the North and South of Rotterdam. Figure E3a shows flats while the other terraced houses. Flats have a shared entrance and staircase for all residents which provide a space of encountering and interaction with the other people (Snow et al., 1981). In this case, the staircase is shared at both sides of the building. The buildings in Figure F.3b do not provide this, but increase the change of interaction with the neighbour by designing the entrances next to each other (Snow et al., 1981). Next, the flats have an open balcony, whereas the houses have an integrated balcony. An open balcony can connect the residents to the street as well as to neighbouring balconies, as widely seen during the COVID-19 pandemic (Problovski et al., 2022). The hanged clothes and the decorations of the balconies promote a sensation of inhabitation and appropriation by the people. Integrated balconies promote the sense of privacy. In addition, we can see an awning which strengthens the anonymity of the residents. Finally, we can see that on the houses in Figure E3b, the facade is separated from the street by a set of bushes. These create a space between the public and the private spheres and increase the sensation of privacy (Rogers & Sukolratanametee, 2009). Instead, in Figure F.3a we can see that there is a basement directly pointed towards the street.





141

(a) Coordinates: 51.89320, 4.50150. South of Rotterdam.

(b) Coordinates: 51.93564, 4.46737. North of Rotterdam.

Figure F.1: Special features of public-private interplay in selected neighbourhoods.



(a) Coordinates: 51.89015, 4.50147. South of Rotterdam.



(b) Coordinates: 51.93379, 4.45993. North of Rotterdam.

Figure F.2: Special features of public spaces in selected neighbourhoods.



(a) Coordinates: 51.88842, 4.48655. South of Rotterdam.

(b) Coordinates: 51.93574, 4.46750. North of Rotterdam.

Figure F.3: Special features of the buildings' facades and layout in selected neighbourhoods.



Figure F.4: Approximate location of the images in figures A.1a, A.1b, A.2a, A.2b, A.3a, and A.3b in Rotterdam.